University of Strathclyde Department of Naval Architecture, Ocean & Marine Engineering

Kelvin Palhares Bastos Sathler

Assessment of Operational Losses as a Mean for Decision Making and Risk Assessment in Wind Energy Assets

Supervisor: Athanasios Kolios

2023

University of Strathclyde Department of Naval Architecture, Ocean & Marine Engineering

A thesis submitted in partial fulfilment of the requirements for the degree of Doctor of Philosophy

Kelvin Palhares Bastos Sathler

Assessment of Operational Losses as a Mean for Decision Making and Risk Assessment in Wind Energy Assets

Supervisor: Athanasios Kolios 2023

© All rights reserved. No part of this publication may be reproduced without the written permission of the copyright owner.

ABSTRACT

In recent years, the wind energy industry has rapidly advanced and is now considered a mature technology. Nevertheless, some researchers and critics doubt its capacity to survive without governmental incentives and subsidies, particularly in the medium- and long-term period, given the increase in costs and operational losses. This thesis aims to explore the challenges of wind energy deployments, focusing on operational losses and costs trends that threaten projects viability. To achieve this goal, first an extended review on operational losses were performed, and an adaptation of the tool Overall Equipment Effectiveness (OEE) was proposed, considering the entire process and losses elements namely availability, performance, and quality. Then, three operational loss trends, increase in failure rate, ageing, and curtailment were identified. Finally, economic analyses, which incorporated traditional metrics such as Levelized Cost of Energy (LCoE) and Net Present Value (NPV), were conducted to establish the impact of cost and losses trends on project viability over their anticipated lifespan. The results suggest that the economic failure of wind systems is mainly due to the underestimation or neglection of part of these losses. Moreover, the newly proposed metric, adapted from OEE, is shown to be an effective tool for highlighting hidden losses and revealing the impact of certain decisions on the entire system. Several case studies and economic analyses were performed to demonstrate the advantages of the proposed framework, including a Multivariate Monte Carlo Simulation (MMCS), which provided a more comprehensive understanding of loss trends rates impact and the benefits of the implementation of solutions such as Condition Monitoring System (CMS) and overplanting to reduce operational losses. Overall, this study provided practical tools that can be easily adapted and tailored to different deployments. The findings have significant implications for researchers, investors, and industry professionals, helping them make better-informed decisions over wind energy projects.

Keywords: Economic appraisal, Operational Losses, Overall Equipment Effectiveness, Availability, Performance, Quality, Ageing, Curtailment, Levelized

i

cost of energy, Net Present Value, Monte Carlo Simulation, Condition Monitoring System, Overplanting.

ACKNOWLEDGEMENTS

I would like to express my sincere gratitude to everyone who supported me, both directly and indirectly, throughout the completion of this thesis. First and foremost, I am deeply thankful for the guidance, encouragement, and unwavering support provided by my PhD supervisor, Athanasios Kolios. His expertise, insightful feedback, and commitment have been crucial in shaping the direction and scope of this project.

I would also like to acknowledge the financial and administrative support provided by Petrobras and CNPq, which enabled me to effectively carry out this research.

I would like to thank all my colleagues and employers from the University of Strathclyde, especially Susan Pawson, who assisted me in resolving administrative issues even before I moved to the UK. Additionally, I extend my appreciation to several employers and contacts from various wind energy companies who provided me with essential information and some data, and to Meteodyn for providing me with their software license and technical assistance.

I want to give a special thanks to my friends who were always there to offer support and keep me grounded amidst the stress of this project.

Most importantly, I dedicate this work to my beloved family. To my wife, Abgail, for her unconditional backing and sacrifices throughout the entire process. To my parents, sister, and niece, for their encouragement, callings, and support. I love you all.

Lastly, I offer my gratitude to God, who gave me the strength and encouragement to persevere through the challenging moments of completing this thesis. I am truly thankful for His unconditional love, mercy, and grace.

TABLE OF CONTENTS

ABSTRACT	i
ACKNOWLEDGEMENTS	iii
TABLE OF CONTENTS	v
LIST OF FIGURES	vii
LIST OF TABLES	ix
LIST OF EQUATIONS	xi
LIST OF ABBREVIATIONS	xii
1 INTRODUCTION	1
1.1 Background	1
1.2 Aims and Objectives	4
1.3 Justification	5
1.4 Thesis Structure	5
1.5 Related Publications	8
2 OEE & OPERATIONAL LOSSES IN WIND TURBINES	9
2.1 What is OEE?	9
2.1.1 Availability (A)	. 10
2.1.2 Performance (B)	. 11
2.1.3 Quality (C)	. 11
2.2 Operational Losses in Wind Turbines	. 12
2.2.1 Classification Criteria	. 12
2.2.2 Availability Losses	. 14
2.2.3 Performance Losses	. 17
2.2.4 Quality Losses	. 21
2.3 Discussion	. 23
2.4 Chapter Summary	. 27
3 QUANTITATIVE ANALYSIS OF OEE IN WIND ENERGY	. 29
3.1 Availability Analysis	. 29
3.2 Performance Analysis	. 34
3.3 Quality Analysis	. 37
3.4 OEE Analysis	. 40
3.5 Case Study	. 42
3.6 Chapter Summary	. 47
4 MONITORING and EVALUATING WIND FARMS	. 48
4.1 Predict and Monitoring Performance through Machine Learning	. 48
4.1.1 Methods	. 51
4.1.2 Data Analysis	. 54
4.1.3 Results and Discussion	. 60
4.2 OEE Analysis through SCADA	. 63
4.2.1 Availability	. 64
4.2.2 Performance	. 64

4.2.3 Quality Estimation	66
4.2.4 Partial Results and Discussion of WTG01	67
4.2.5 Results of the Wind Farm	73
4.3 Alternative of Performance Analysis	74
4.3.1 Wind Speed x Generator Speed	75
4.3.2 Power Coefficient (PC)	77
4.3.3 Discussion	
4.4 Chapter Summary	79
5 LOSSES TRENDS & COST IMPACTS	80
5.1 Losses Trends	80
5.1.1 Failure Rate	80
5.1.2 Ageing	
5.1.3 Curtailment Trend	82
5.2 Cost Impacts	
5.2.1 Methods	
5.2.2 Cost Assessment	85
5.2.3 Scenarios	
5.2.4 Economic Model Application	91
5.2.5 Results and Discussion	94
5.3 Case Study	
5.4 Chapter Summary	109
6 SOLUTIONS TO REDUCE OPERATIONAL LOSSES	111
6.1 CMS	111
6.2 Maintenance x Ageing	116
6.3 Power Integration Solutions	119
6.4 Multivariate Monte Carlo Simulation Case	121
6.5 Chapter Summary	124
7 DISCUSSION	126
8 CONCLUSION	131
8.1 Summary of Chapters	131
8.2 Thesis Contributions	133
8.3 Limitations	139
8.4 Future Works	140
REFERENCES	141
APPENDICES	177
Appendix A	177
Appendix B	179

LIST OF FIGURES

Figure 1-1 Cumulative wind energy capacity in the world	2
Figure 1-2 Costs of wind turbine lifespan	3
Figure 1-3 Comparison of two Life Cycle Cost scenarios	3
Figure 1-4 Thesis Structure	7
Figure 2-1 Six Main Losses	10
Figure 2-2 Flow Chart Losses in Wind Power according OEE tool	13
Figure 2-3 Main Operational Losses in Wind Power	24
Figure 3-1 MTTR, MTBF, and MTTF	30
Figure 3-2 Bathtub curve	
Figure 3-3 Analysis of OEE for onshore and its elements according to) literature. 41
Figure 3-4 Analysis of OEE for offshore and its elements according to	o literature. 42
Figure 3-5 Downtime Breakdown.	44
Figure 3-6 Fault Alerts and Failures Downtime Breakdown	44
Figure 3-7 Production Real Data - Outlier highlighted.	45
Figure 3-8 Grid Losses Downtime Breakdown and Fault List	46
Figure 3-9 OEE Analysis of Wind Farm Case Study.	46
Figure 4-1 Flowchart of the model proposed	50
Figure 4-2 Pre-processing steps in WTG01.	55
Figure 4-3 Box Plot of CV of Power Output Prediction with only Wind	Speed 56
Figure 4-4 Box Plot of CV of Power Output Prediction with Performance Speed (MODEL 1)	e and Wind 57
Figure 4-5 Box Plot of CV Performance Prediction through Met Prediction Model. (MODEL 2)	Mast Data 58
Figure 4-6 Boxplot of MODEL 2 after RFR tuning	59
Figure 4-7 Performance Prediction X Real Performance – WTG01 – J	anuary. 62
Figure 4-8 SHAP results of the performance	77
Figure 5-1 Diagram with cash flow analysis	85

Figure 5-2 Diagram of Breakeven and Critical year analysis – Part I
Figure 5-3. Flowchart of Economic Model and Monte Carlo Simulation – Part II
Figure 5-4 Results Scenario A95
Figure 5-5 Results of Monte Carlo Simulation of Scenario A
Figure 5-6 Results of the sensitivity analysis on LCOE
Figure 5-7 Results of the sensitivity analysis when NPV>0
Figure 5-8 Mean Wind Speed analysis of the hypothetical farm in Meteodyn. 100
Figure 5-9 Power Curve and CT Curve from turbines assessed
Figure 5-10 Wind Distribution of each class 103
Figure 5-11 Turbulence Analysis of Wind Turbine IA.
Figure 5-12 Turbulence Analysis of Wind Turbine IIA.
Figure 5-13 Turbulence Analysis of Wind Turbine IIIA.
Figure 5-14 Turbulence Analysis of Wind Turbine IIB
Figure 6-1 P-F Curve 112
Figure 6-2 Scenarios Investigated – Trade-off Maintenance x Ageing 117
Figure 6-3 Sensitivity Analysis of Maintenance Cost Reduction X Ageing rates.
Figure 6-4 Result from Multivariate MMCS 124
Figure_Apx B-1 MMCS results for returning tax = 4%
Figure_Apx B-2 MMCS results for returning tax = 8%

LIST OF TABLES

Table 2-1 Main Cause of Losses by Availability in Wind Power	16
Table 2-2 Main Cause of Losses by Performance in Wind Power	
Table 2-3 Main Causes of Losses by Quality in Wind Power	23
Table 3-1 Summary of downtime investigation for onshore projects	32
Table 3-2 Summary of Dataset	43
Table 4-1 Description of Machine Learning Methos Selected	51
Table 4-2 RFR Optimization parameters – WTG01	59
Table 4-3 Results from WTG1 with 15,141 data points	60
Table 4-4 Average Results from Simulations	63
Table 4-5 Summary of the results of WTG01	68
Table 4-6 Comparison of two different periods with different performant	nce 70
Table 4-7 Results of WTG01 grouped per Temperature Range	71
Table 4-8 Results of WTG01 grouped per month	71
Table 4-9 Results of WTG01 grouped per TI Classes	71
Table 4-10 Results of WTG01 grouped per Wind Speed Bin	72
Table 4-11 Results of WTG01 grouped per Wind Direction Bin	73
Table 4-12 Results considering all wind farm	74
Table 4-13 Summary of Performance Analysis - Wind Speed x Generation	ator Speed
Table 4-14 Summary of Performance Analysis – CP.	78
Table 5-1 Costs from the literature review	89
Table 5-2 Average Costs and Basic Assumptions to the model	93
Table 5-3 Scenarios investigated	94
Table 5-4. Projected cost throughout lifespan considering average val	ues 94
Table 5-5. Results from Monte Carlo Simulations in all scenarios	96
Table 5-6 Basic Parameters for wind turbines	101
Table 5-7 Summary of Turbines and Meteodyn Results.	106
Table 5-8 Summary of Cost Analysis of each Turbine Class	107

Table 5-9 Summary of Case Results	108
Table 6-1. Comparison offshore case with and without CMS. 1	114
Table 6-2 Results from financial return – With and without CMS	115
Table 6-3 Results from Trade-off Maintenance x Ageing Investigation	118
Table 6-4 Operational Losses Trends and Rules for MMCS model	122
Table 6-5 Overplanting and CMS Rules 1	123
Table 6-6 Summary of MMCS per solution type 1	124
Table 8-1 Summary of Objective 1 1	134
Table 8-2 Summary of Objective 2. 1	135
Table 8-3 Summary of Objective 3. 1	136
Table 8-4 Summary of Objective 4. 1	137
Table 8-5 Summary of Objective 5. 1	138
Table_Apx A-1 Results of Economic Model - return rate = 4%	177
Table_Apx A-2 Results of Economic Model - return rate = 8%	177
Table_Apx A-3 Result MCS – return rate = 4%	178
Table_Apx A-4 Result MCS – return rate = 8%	178

LIST OF EQUATIONS

(2-1)	
(2-2)	11
(2-3)	11
(2-4)	
(3-1)	
(3-2)	
(4-1)	53
(4-2)	53
(4-3)	53
(4-4)	53
(4-5)	64
(4-6)	65
(4-7)	65
(4-8)	65
(4-9)	66
(4-10)	67
(4-11)	77
(5-1)	
(5-2)	
(5-3)	102
(5-4)	
(5-5)	
(5-6)	
(5-7)	
(5-8)	
(5-9)	

LIST OF ABBREVIATIONS

A	Availability
Amm	Availability Maintenance Metrics Methods
Атв	Availability Time-based Method
В	Performance
BS EN	British Standard European Norm
С	Quality
CAPEX	Capital Expenditure
CF	Capacity Factor
CFD	Computational Fluid Dynamic
CMS	Condition Monitoring System
СТ	Coefficient of Thrust
DECEX	Decommissioning Expenditure
DTR	Decision Tree Regression
ESS	Energy Storage System
ETM	Extreme Turbulence Model
IEC	International Electrotechnical Commission
LCC	Life Cycle Cost
LCOE	Levelized Cost of Energy
LNR	Linear Regression
MAE	Mean Absolute Error
MAPE	Mean Absolute Percentage Error
MCS	Monte Carlo Simulation
MMCS	Multivariate Monte Carlo Simulation
MTBF	Mean Time Between Failure
MTTF	Mean Time to Failure
MTTR	Mean Time to Repair
NPV	Net Present Value
O&M	Operational and Maintenance
OEE	Overall Equipment Effectiveness
OPEX	Operational Expenditure
PC	Power Coefficient
R ²	R-Squared

RAM	Reliability, Availability, and Maintainability
RFR	Random Forest Regression
RMSE	Root Mean Square Error
SVR	Support Vector Regression
TI	Turbulence Intensity
TSR	Tip Speed Ratio

- β Pitch Angle
- λ Failure Rate
- μ Mean
- σ Standard Deviation

1 INTRODUCTION

Wind energy has rapidly developed in recent years and has emerged as a leading renewable alternative to reduce dependence on fossil fuels. However, some authors have questioned the long-term viability of wind systems, particularly due to increasing operational losses. In response, this study focuses on developing a metric to effectively measure operational losses, identify loss trends over the lifespan of wind systems, and investigate critical factors that could impact their viability. This chapter provides an overview of the research, starting with the background of the problem (Section 1.1), followed by the aims and objectives of this work (Section 1.2), and its justification (Section 1.3). In Section 1.4, the thesis structure is outlined, and Section 1.5 highlights the papers produced and submitted during the PhD.

1.1 Background

With significant political and financial incentives in recent years, wind power has experienced a sustainable increase in its contribution to the national energy mix. In fact, it covered 15% of European electricity demand in 2019 [1], and there is a target to achieve approximately 30% by 2030 [2]. This growth is reflected in the cumulative wind energy capacity worldwide, surpassing 900GW (including both onshore and offshore projects) as shown in Figure 1-1, aligning with the trends observed in Europe. Although the annual growth rate has slightly declined in recent years, it is projected to exceed 15% in the next five years, from 2022 to 2026 [3]. While these figures are highly positive, for wind energy to become more independent of incentives and attract greater investment, there are still challenges to overcome. These include reducing the cost of energy and enhancing performance to maximize profitability throughout the service life of wind projects.



Figure 1-1 Cumulative wind energy capacity in the world, adapted from [3].

A very common metric to calculate the cost of energy is through LCOE (Levelized Cost of Energy). LCOE should be thought as the ratio between the total production and total costs during its lifespan, considering financial costs, time value of the money, and some profits to investors. The total cost of implementation is known as Capital Expenditure (CAPEX), while during the operational lifetime, there are O&M and management costs, also known as Operational Expenditure (OPEX) [4]. At the same time, it is during this period that the benefits are achieved through the electricity produced and sold. Finally, after the nominal service life period and a potential service life extension, Decommissioning Expenditure (DECEX) will take place and relevant costs should be considered. Figure 1-2 summarizes all these costs and benefits. It is important to mention that some return might come from the disposal of materials and equipment after decommissioning, and, for that reason, the disposal is represented in blue and has a question mark. Studies show that recycling can cover up to 20% of offshore decommissioning costs [5].



Figure 1-2 Costs of wind turbine lifespan – adapted from (Sathler 2013).

Even though one might consider that reducing the CAPEX value might be an adequate approach to reduce the total life cycle cost of a wind power project, it can be a rather simplistic solution. As shown in the first scenario of Figure 1-3, a poor implementation choice can affect the whole operational performance, increasing the total cost. Thus, it is important to strike a balance between all costs during the project design phase, as prioritizing the reduction of implementation costs alone may negatively impact future operational costs and result in higher overall costs, as illustrated in the second scenario of Figure 1-3, where higher implementation costs result in lower overall costs.





Towards reducing LCOE, there are mainly two approaches to be followed: increasing production or reducing the total costs. Traditionally, after installation, connection to the grid and commissioning, wind farm operators focus on maximize production and availability, while reducing its costs. However, in modern wind farms, operators consider more sophisticated KPIs, such as the maximization of profitability. This is because there comes a point where producing additional electricity may incur additional costs that outweigh the benefits. This situation is more prevalent when projects are not operating under fixed prices or when costs become elevated [8]. In such cases, operators may choose to curtail the energy production to avoid incurring excessive costs. This strategic decision can help optimize the overall profitability of the wind farm.

Either way, the operational period plays a critical role in determining the success of the entire project and subsequently the potential of an extended service life can be anticipated or postponed. This decision is made based on the decrease of profitability, reliability and performance [9]. Some studies have found that the operational costs of wind turbines could jump from two to four times throughout their service life [10]–[15]. While others identified the impact of the degradation of the system with time, or ageing, which could reduce the relativity production capacity from 5 to 20% in 20 years [16]–[18]. Some countries have already registered yearly curtailment levels (which includes energy rejections and lack of demand) from 5 to 17% [19]–[21]. Therefore, investigate the impact of these losses along project's lifespan and develop an efficient metric to measure real productivity of a wind farm can be crucial to reduce risks and support decisions and a continuous improvement culture.

1.2 Aims and Objectives

The first aim of this work is to investigate the operational losses in wind energy assets and gather them as a unique and solid metric. This metric is an adaptation of the well-known in manufacturing industry, OEE (Overall Equipment Effectiveness). Additionally, the second aim of this work, the impact of operational loss trends in wind energy viability through economic models will be investigated and performed. Thus, the specific objectives of the work are summarized below:

- 1. Review of operational losses throughout the entire process, including wind penetration into the grid, and classify them into groups following OEE concepts to help operators to identify specific areas for improvements.
- Apply OEE to wind energy following framework previously developed and perform quantitative analysis according to literature figures and real cases to evaluate wind energy effectiveness.
- 3. Conduct data analysis to develop different models and procedures to monitor and estimate OEE.
- Investigate in literature how some operational losses increase along the lifespan of wind energy assets and evaluate its impacts on costs and viability.
- Perform stochastic analysis to check viability, costs, and risk of economic failure of wind energy assets along its lifespan in different scenarios, combining different losses trends and some proposed solutions to reduce operational losses identified in literature.

1.3 Justification

Operational losses can vary depending not only on project particularities, but also on how they are accounted. Wind energy assets are surrounded by uncertainties and affected by external factors, which contributes for the complexity of the activity of measuring and monitoring them. Therefore, a simple and reliable metric that accounts for all possible losses is important for both academia and industry to effectively compare and assess assets deployed in different settings, and to investigate proposed solutions thoroughly. This is particularly vital, as improvements in one specific area may negatively affect other parts of the system. Additionally, some studies have shown the increase of losses and the decrease of productivity of wind assets along its lifespan. Thus, identifying and monitoring these operational losses trends beforehand can be critical for a more accurate financial model and for reducing risks.

1.4 Thesis Structure

This thesis has been organized as follows:

Chapter 1 provides a comprehensive summary of the motivations, aims, justification, and presents the overall framework of the present work.

Chapter 2 presents briefly OEE concepts and a critical literature review of main operational losses in wind energy assets. In this chapter, it was defined a classification criteria following the concepts of the tool.

Chapter 3 estimates the average OEE of onshore and offshore deployments, considering some figures found on the literature. Additionally, a real case is presented to compare results.

Chapter 4 investigates how to monitor and evaluate wind farms through traditional machine learning techniques and SCADA data analysis.

Chapter 5 identify losses trends in the literature and two economic analyses are performed. First, considering average offshore costs to understand the impacts of each scenario. Second, a simulated case compares the adoption of different turbine classes.

Chapter 6 introduce to some common solutions and briefly discuss its impacts on the costs. Additionally, a multivariate Monte Carlo is developed.

Finally, Chapter 7 presents a discussion of the main findings of the thesis while the chapter 8 concludes the thesis, by summarizing each chapter, stating the limitations, and suggesting future work.

Figure 1.4 gives an overview of the thesis structure and content.



Figure 1-4 Thesis Structure.

1.5 Related Publications

The following papers were developed along the PhD and was already published or under review at the moment this thesis was written. Parts of this work are extracts of them.

K. P. B. Sathler, A. Kolios, S. Al-Sanad, and J. Parol, "Application of the Overall Equipment Effectiveness Concept in Wind Energy Assets," 2020, doi: 10.3850/978-981-14-8593-0.

K. P. B. Sathler and A. Kolios, "The Use of Machine Learning and Performance Concept to Monitor and Predict Wind Power Output," *Int. Conf. Electr. Comput. Energy Technol. ICECET 2022*, no. June, pp. 20–22, 2022, doi: 10.1109/ICECET55527.2022.9873076.

K. P. B. Sathler, K. Salonitis, and A. Kolios, "Overall Equipment Effectiveness as a Metric for Assessing Operational Losses in Wind Farms: A Critical Review of Literature," *Int. J. Sustain. Energy*, 2023, doi: 10.1080/14786451.2023.2189490.

K. P. B. Sathler, B. Yeter, and A. Kolios, "Impact of operational losses on the levelized costs of energy and in the economic viability of offshore wind power projects," [Manuscript submitted for publication] 2023.

K. P. B. Sathler, A. Kolios, and B. Yeter, "Effect of Energy Losses on Onshore Wind Turbines Techno-economic," [Manuscript submitted for publication] 2023.

2 OEE & OPERATIONAL LOSSES IN WIND TURBINES¹

There are mainly three sources of loss that can be expected from a wind energy converter: Aerodynamics-related, electromechanical, and operational. Whilst the first two are related to manufacturing and design, which can be reasonably estimated, the same argument can hardly be made for the operational losses. Wind turbines are surrounded by uncertainties, including factors such as the wind features inputs, grid integration challenges, and degradation of the system, which makes operational losses assessment a hard task. Other industry has dealt with similar problems and developed different tools to help operators and managers to identify weak points and the main causes of losses. One of these tools is OEE, widely used in manufacturing systems to enhance equipment efficiency by reducing operational losses. Therefore, Section 2.1 will briefly present the metric. In section 2.2, the classification criteria and the literature review are presented, and the finds are discussed in Section 2.3.

2.1 What is OEE?

In the early 1970s, the Japan Union of Scientists and Engineers (JUSE) has developed a maintenance strategy called TPM (Total Productive Maintenance), where the goal was to achieve maximum performance in its production considering all phases related to the production. In order to check its efficiency, a metric called OEE (Overall Equipment Effectiveness) was introduced, where all possible causes of losses and the main six losses are identified and classified in three main elements, named availability (A), performance (B) and quality (C), as shown in Figure 2.1.

¹ This chapter is based on publication by Sathler, Salonitis and Kolios [288]



Figure 2-1 Six Main Losses.

Any change in the process can affect one or more elements. For that reason, OEE became an important productivity tool, since it considers the overall result and efficiency, helping to identify where losses are more frequent, and hence targeting improvement interventions. The OEE index is obtained through the multiplication of the three elements, and it represents the overall performance of the equipment. This index is considered an important metric to help managers and operators to make decisions, increasing the productivity of the equipment or process.

$$OEE = Availability (A) x Performance (B) x Quality (C)$$
(2-1)

2.1.1 Availability (A)

Availability is calculated considering the planned operating time discounted by the period that the equipment is not available to operate, known as downtime [22]. There are two main types of losses in the availability category which can cause downtime, as shown in Figure 2.1. The first accounts for breakdowns, which is generally related with maintenance or failures and the time spent to fix the interruption cause. The second type accounts for adjustments and set-ups, where we refer to pauses in production which are not related with breakdowns. Relevant examples include planned maintenance interventions and adjustments of the equipment for new products. Even though some losses are expected, it is important to quantify them and understand their influence to key output indicators, in order to identify areas of improvements of the process. The basic formula to calculate operational availability is:

$$A = \frac{Planned \ Operating \ Time - Downtime}{Planed \ Operating \ Time}$$
(2-2)

It should be noted that in practice different formulas are adopted for different types of availability, such as inherent, time-based, revenue-based etc, so it is important to ensure that the right metric is adopted.

2.1.2 Performance (B)

Losses related to performance can be the most challenging ones to identify since they are considered through instances where the equipment is performing outside the specification limits set. This type of losses can be related to reduced speed, meaning that for any reason a part of the equipment is running with lower performance, which can be caused by a damage, a not-well-lubricated bearing or lack of alignment, for instance. Another reason for performance losses is minor stoppages, where faults cannot be measured, but production performance is affected. An example is when in a cycle for any reason a motor is taking one second more to start due to a mechanical or electric fault, which is not easy to be recognized by operators. However, it can become a significant loss when accumulated throughout every operational cycle. A performance rate control can warn operators when something is wrong and needs to be investigated. Performance is calculated over planned operating time minus downtime, so the availability loss is not considered twice:

$$B = \frac{Standard Production Rate x Parts Produced}{Planned Operating Time - Downtime}$$
(2-3)

2.1.3 Quality (C)

Finally, quality is related to the final product as a result of a process or operation of equipment. Any producing process should ensure that the final product meets the end users' requirements or the client. The first type of losses in this factor accounts for defects, i.e., when the product is out of specification, and it should be discarded. The second element is rework, when minor defects are identified, and extra work is required in order to recover the product. According to the OEE concept, this is also considered as a loss because the time and the resources spent to fix it could be used to produce a new product or they can reduce the operational life of the process. The formula to calculate quality only considers products that were produced on the period assessed:

$$C = \frac{Units \ Produced - Defective \ Units}{Units \ Produced}$$
(2-4)

While OEE is widely utilized in the manufacturing industry and incorporates major losses in an intuitive manner, it is important to acknowledge its drawbacks and limitations. Firstly, this metric lacks detailed insights into the causes of operational losses, requiring additional effort to accurately identify and correlate each measured loss with its root cause. Additionally, OEE does not encompass critical aspects such as costs and potential investments. Moreover, the tool can potentially lead to misguided actions if the operational team fails to measure the elements accurately or compares results across different deployments without considering project-specific factors. However, these limitations can be largely overcome through the implementation of automation or a systematic approach to calculate OEE, coupled with proper training of the operational team. It is also important to recognize that OEE functions as a complementary metric and should be used in conjunction with other performance metrics such as reliability, maintainability, Failure Tree Analysis, FMEA (Failure Mode and Effects Analysis).

2.2 Operational Losses in Wind Turbines

2.2.1 Classification Criteria

As demonstrated in Section 2.1, the OEE focus is to identify, classify, and quantify operational losses. To adapt it to wind farms projects, some considerations need to be done. The flowchart shown in Figure 2.2 illustrates the assumptions considered in this work to classify the losses found in literature and what the authors believe would be a suitable approach to adapt the tool in wind energy assets. It is important to notice that the decision element in the flowchart started with the preposition "from" because each index is considered from the result of

the previous one, avoiding losses being accounted twice during the process analysis.



Figure 2-2 Flow Chart Losses in Wind Power according OEE tool.

Another important observation in the flowchart is related to the final result. Although the decision question was included in the flowchart, it is very unlikely that no losses is registered in an industry. This could be achieved in a short-term period, but considering long term periods, losses are expected and considered in all projects. For instance, an OEE of 85% is considered world class benchmark [23], representing that, even in reliable projects, there are losses. The following subsections detail the losses and how the scientific community has been aiming to minimize them, especially in the operational perspective. Also, it is important to mention that OEE results cannot be fully compared between deployments. Even turbines from the same company may not have the same OEE, as the equipment productivity relies not only in operational system, but also on the management commitment, involvement of the O&M team, maintenance efficiency, wind farm location, deployment environment and other particularities that any project has.

To identify the operational losses, before categorizing them, an extended literature review was performed. The focus was on papers published from 2010 to nowadays that had key words or expressions such as "operational losses", "quality losses", "production losses", and "performance losses", together with "wind energy" or "wind power", in their titles and/or abstract. Then, a carefully reading was performed to check if important information could be retrieved and if the paper was really related to wind power and operational losses. To a better illustration of the sort of solutions or discussion was provided in the studies, assessed, five categories was also defined. "Investigation" refers to papers that assess the operational losses and discusses it, through reviews, numerical models, trends, or data analysis. "Decision Support" refers to when a framework or a new methodology is created or adapted, which resolves in important information that could help operators to minimize losses. "Controllers" refers to the development of a controller to reduce losses, find optimal point, or change the premises and settings of traditional controllers. "Machine Learning" refers to papers which use any machine learning technique to perform predictions or find correlations among inputs and outputs. And, finally, "Others" refers to solutions that are not listed before, including technical changes or addition of components or gadgets in the system.

2.2.2 Availability Losses

The first aspect to be considered is the set time contemplated in the OEE calculations. In the wind energy industry, turbines are designed to operate all year long, so considering the entire calendar period as a set time base is realistic. The

first source of downtime mentioned in OEE is breakdowns. This item is related to any time that the turbine is not available to produce due to unexpected downtimes, such as failures and corrective maintenance. To address availability losses, increase turbine reliability, and extend their lifespan, important tools such as Reliability Centred Maintenance (RCM) [24] and Reliability, Availability, and Maintainability (RAM) can be utilized. Within the framework of RCM, tools like Fault Tree Analysis (FTA) [25] and Failure Mode Effect Analysis (FMEA) [9][26] [27], and specific studies on failure rates [28][29] and others reliability-based methods [30] can assist operators in identifying critical failure pathways and evaluating potential impacts on system performance. These methods exemplify useful approaches for informing decision-making, optimizing maintenance strategies and increase general reliability of the turbine and its components.

The second downtime factor mentioned in OEE refers to adjustment and set-ups. Differently from the traditional manufacture industry, wind turbines do not need to change worn out tools or adjust their process for new products. Therefore, the only "expected downtime" for wind turbines is preventive maintenance. Many papers that cover availability discuss both downtime cases, corrective and preventive maintenance, which makes hard to separate them efficiently. However, some paper focus on suggesting strategies to improve preventive maintenance schedule [31]–[33], including the use of machine learning to better predict wind conditions [34], which could be used to affect less the wind energy output [35]. It is worth noting that in some cases, this may result in increased costs due to vessel availability constraints, as many wind farms will likely operate within the same time frame.

Offshore deployments have a particular approach. While onshore wind farms can achieve around 98% of availability [36], offshore wind farms present a lower pattern achieving around 92% [37][10]. Besides the distance to the shore and the need of vessels, also considering safety constraints, accessing the turbine is only possible in appropriate climate conditions. Therefore, the cost may be increased affected by this dependency, since some maintenance or changing of components needs to be done in advance, in the appropriate time, instead of

15

the best and more effective time, wasting some of the components remaining lifetime [38]. A logistic maintenance review for offshore operations is presented in [39] and maintenance cost reduction review for offshore farms in [40].

Table 2.1 presents the main causes of losses by availability in wind turbine. The second column gives some examples of the losses cause, and the third column includes studies in which losses were assessed or quantified, suggested solutions, or compared different approaches. The category of the papers, as mentioned in Section 2.2, is also pointed in Table 2-1.

<u>Losses</u>	<u>Example</u>	Related Papers	Investigation	Decision Support	Controllers	Machine Learning	Others
		FMEA - [9], [26], [27], [41]	Х	Х		Х	
Breakdowns	Failures Corrective Maintenance	Reliability and Failure Rate Analysis - [24], [28]–[30], [42], [43]	х	Х			
		Fault Tree Analysis on floating offshore turbine - [25]	х				
		Fault Predictions/Detection - [44]– [51]		Х		Х	
		Uncertainties in O&M models - [52], [53]	х				
		Human Impact on maintenance - [54]	Х				
		Fatigue and Failures related to weather - [55]–[61]	х	Х		Х	х
Preventive Maintenance	Preventive Maintenance	Maintenance Cost Review for Offshore [40], [62]	х				
		Method for better maintenance scheduling - [31]–[33], [35], [39], [63]–[67]		x			
	Inspections	CBM - [27], [68]	Х	Х		Х	
		Reliability Monitoring - [34], [46], [47], [49], [69]		Х		Х	

 Table 2-1 Main Cause of Losses by Availability in Wind Power.

2.2.3 Performance Losses

Differently from the traditional manufacturing industry, wind turbines have different performance rates to be assessed, since the production output depends directly on the wind features, especially wind speed and density. Therefore, before discussing about the losses, the standard rate needs to be commented. The most usual way to assess wind production is through the wind power curve, which defines the production according to the wind speed, normally tested in a lab and later confirmed in a Power Curve Test, according to IEC 61400-12 standard and some local regulations [70]. Some researchers have proposed more accurate power curve considerations, including other factors such as wind direction [71]–[73], turbulence [74] or air density [75] and controllers [73].

Even though these approaches are good for a better production prediction, according to OEE, this can hide some opportunities for improvement. To illustrate this, one common problem related to wind direction is the wake effect. Although wake effect losses are expected, some researchers have proposed solutions to minimize them during the operational phase, such as intentional misalignment of yaw controller [76][77] or changing individual controllers to farm controller [78][79]. In other words, although some wind features cannot be controlled, considering it in the best performance rate can mislead the results and do not incentivize operators to find ways to minimize them in case of a high impact.

Some additional observations regarding the standard rate, as discussed in section 2.1.2, merit further discussion. While there is no rigid rule, and operators can adapt these concepts to their specific needs, it is important for the standard rate to be as simple as possible to minimize human errors and misinterpretations. Unlike other performance metrics, OEE aims to closely align with the best possible performance. Therefore, instead of using manufacturing curves that typically consider average production within a given wind speed bin, the standard rate in OEE should consider the best achievable rate. The goal for operators then becomes minimizing the gap between the best performance and the actual production. As a general guideline, if the actual performance frequently exceeds 100%, it suggests that the standard rate is underestimated. On the other hand, if

17

the actual performance is far from reaching the standard rate, or if it has never reached the standard rate even for a short period, it indicates that the standard rate is overestimated.

With respect to performance losses, two main causes were pointed: reduced speed and minor stoppages. As mentioned in Section 2.1.2, performance index considers losses that cannot be measured as easily as availability, so comparing production output at the same conditions could be the best way to identify reduced speed and minor stoppages, including faults and failures in the system that does not send alerts to operators. Since wind turbines are complex equipment and exposed to hard and uncontrolled environments, the performance losses can be caused by several factors. Some of these are related to the equipment itself, while others are related to the environment.

Even though climate features are not controlled by the operator, they need to be considered in order to better understand the performance behaviour. Some papers relate differences in performance due to seasonal conditions or periods of the day [80], humidity [81], turbulence [82], and other papers are looking for a way to minimize losses due to rain [83] and icing [84]–[86], even using machine learning [45]. For offshore wind farms some additional issues can be considered in this category, such as wave impact due misalignment of the turbine [56][57] and platform motion that can affect the performance of other controllers [87]–[91].

The ones related to the system can be influenced by the condition of other components. Usually, the increase of temperature, vibration, or abnormal effort can affect productivity. Thus, these can be considered examples of reduced speed caused by damaged bearing [92], lack of lubrication, wear outs in components or even ageing [16]. For instance, [58] indicates a performance decrease before failures. Another important loss that is usually neglected is the time spent to start the generation of energy. Every time the turbine is shut down, due to safety reasons, maintenance, or lack of wind, the equipment spends time to gain inertia, start rotation and generate electricity. Thus, a more efficient "starting up" time can directly affect the production rates throughout the year. In some situations, the time needed to achieve the operational rotation can be
affected by wind speed, as demonstrated in [93], which tested this in small scale wind turbines. However, some innovative solutions have been proposed to deal with this problem, such as engaging a motor to increase production range and reduce loss due to starting up [94].

Finally, another cause of losses that were not mentioned before are controllers' systems (which includes sensors and actuators). They could be related to reduced speed or minor stoppages due to malfunctioning or faults. However, in this paper, controllers were considered separately because some researchers focused on improving production by changing controller's models, settings and/or premises. Besides reducing wake effects, as mentioned in the second paragraph of this section, yaw systems can be used to increase production [95]–[99]. The same stands true for other controllers, such as pitch control [100][101], stall [102], and other PI controllers [103]. Artificial Neural Networks, Machine Learning and new algorithms to control or find optimum sensor placement are studied as well [76][104]–[106].

To summarize, it is important to keep OEE calculation as simple as possible, so using maximum performance in a certain range of wind speed can be appropriate as a start. Obviously, this does not indicate to the operator the reason of the loss, but it shows that something is not functioning well and, depending on the level of the loss, the operator can decide if some action needs to be prioritized. Table 2-2 gathers the main losses identified and papers related. Some of losses can be classified in different criteria, but the most important is to have a reliable and simple index that does not account the same loss twice.

Table 2-2 Main Cause of Losses by Performance in Wind Power (*Not fullyresponsibility of operators **Only offshore deployments)

Losses	<u>Example</u>	<u>Related Papers</u>	Investigation	Decision Support	Controllers	Machine Learning	Others
	Wind Features: Turbulence	Power curve models (important to define Standard Rate) and Output Prediction - [71], [74], [75], [107]–[112]	х	х		х	
	Direction Air Density	Investigation of the impact of climate and wind conditions:					
	Rain	Turbulence - [82]	Х				
Climate	Humidity	Air Density - [73]				Х	
Conditions*	Season	Rain - [83]	Х				
	High	Humidity - [81]	Х				
	Temperature	Period of the day - [80]	Х				
	Period of the day Waves**	Seasons - [113]	Х	Х			
		Direction - [114]	Х	Х		Х	
		Waves - [57], [88], [89]	Х				
Reduced Speed	Ageing Blades Factures/Erosion Icing Dust Wake Effects Low Speed of components Start-up	Losses due ageing - [16], [18], [115], [116]	Х				
		Losses due fractures/erosion - [117], [118]	Х				
		Icing losses detection and estimation - [45], [84]–[86], [119], [120]	Х		Х	Х	
		Investigation on wake effects - [79][121][122][114][123][124]	Х	Х			
		Reduce wake effects - [76]–[78], [105], [106], [125]–[128]		Х	Х		
		Losses due impact of wave loads - [56], [87], [89]		Х	Х		Х
		Improving performance - [129]–[131]	Х		Х		
		Balance between load and output - [132]			Х		
		Reduce cut in and minimize losses - [94]					Х
Minor Stoppage	Small Failures (don`t stop	Identifying malfunctioning - [92], [107], [129], [132], [133]	х	Х		Х	
	production) Defects	Fault tolerant identification - [134]		Х			

Losses	Example	Related Papers	Investigation	Decision Support	Controllers	Machine Learning	Others
	Misinterpretation	Yaw Controller - [95]–[99], [105]	Х		Х	Х	
Controllers	of signals	Stall Controller [102]			Х		
	Faults Controller`s	Pitch Controller - [91], [100], [101], [104], [106]	Х		Х		
	setting	PI Controller - [103]			Х		

2.2.4 Quality Losses

Quality assessment in wind energy production poses significant challenges due to the complexity of calculating and classifying all losses that occur after the electricity is produced by the generator. To address this challenge while ensuring simplicity and effectiveness, this study defines quality losses as any losses occurring between the generator and the grid. Unlike traditional manufacturing industry, where the outcome is a physical product that can be reworked, wind energy production cannot "fix" electricity once it is generated. It is important to note that some quality losses in wind energy systems are directly influenced by turbine operational conditions and operator decisions, highlighting a partial connection between quality and other elements. This will be presented and discussed along this section.

As mentioned in Section 2.1.3, defects refer to when the outcome does not achieve the client's requirements. In the wind industry, the client can be considered the grid, so quality in this study refers to grid requirements. Due to the intermittent and uncontrolled input, wind power suffers from several variance and fluctuations. Some of the problems related are flickers, harmonic variance, impedance, resonance, and frequency fluctuation. It is out of scope of this study to discuss each of these problems, but it is important to mention that they can vary according to each grid's characteristics or country regulations. Further grid problems related to quality, including local issues in different countries, are discussed in relevant literature[107][135]–[137].

Some of the quality problems are related to efficiency of intermediate equipment or design solutions [138]–[141] Since some of the energy output can be partly controlled, some researchers are studying ways to minimize them, such as control frequency [142] or harmonics [101] through pitch angle, and flickers and voltage fluctuation through yaw and stall control [102].

Another problem related to the grid which could affect the quality index is the grid availability. As mentioned before, the input in wind energy cannot be controlled, so if the grid cannot receive the electricity, the generation is disconnected, and this becomes an important loss. This can happen due to safety reasons, which include ramps, unstable electricity, grid faults or by lack of demand. Some operational measurements can reduce these losses as well. To minimize ramps, a paper suggests new controller approaches [143], while other works identify safety problems and relates them to other variables [144][145][100], which could be strategic for operators knowing when instability is more likely to occur. Curtailment issues have become a widely discussed topic [19][146], with some proposed solutions related to better production predictions [147][148], expand grid capacity [149] or strategically increase demand during high production [150].

Finally, the last problem related to quality element is due to transmission. This includes basically cabling and intermediary equipment. The transmission system is designed in the project phase and some technical losses are assumed, but it can be difficult to modify it after implementation. However, monitoring transmission losses can indicate when abnormal behaviour or wear outs occur in cables [151]–[153], or when intermediary equipment lose their effectiveness throughout the time. Also, the quality of the electricity produced can cause losses during transmission, as pointed in [154]. This information can guide some decisions made by operators, since the increase in the losses could justify some more extreme interventions. Table 2.3 outlines the main quality losses identified in wind turbine.

Table 2-3 Main Causes of Losses by Quality in Wind Power (*Not fullyresponsibility of operators).

Losses	Example	Related Papers	Investigation	Decision Support	Controllers	Machine Learning	Others
		Fluctuations in output - [141], [155]–[157]	Х				
		Loses due frequency - [142], [158], [159]			Х		Х
	Frequency	Flickers - [102], [155]	Х				
	Voltage	Losses in quality due to wave	v				
Out of	Harmonics	misalignments - [89], [90]	X				
Requirements	Flickers	Harmonics - [101], [154]	Х		Х		
	Converters` fault	Voltage fluctuation - [102], [138], [160], [161]	х		х		
		Losses due power flow controller and converter's fault - [162]–[164]			х		
	Curtailment	Estimation and investigation of curtailment in different countries - [19], [146], [149], [150]	х				х
		Proposed method to reduce curtailment - [147], [148], [165]–[167]	х		х		Х
Grid Availability*	Security	Reducing/Monitoring ramps - [168], [169], [143]	х				
	Grid Faults Ramps	Hybrid system to reduce curtailment and instability - [170]–[172]	х				
		Instability in grid, reduce inertia - [100], [113], [135]–[137], [144], [145], [173]–[177]	х	х	х	х	
Transmission	Cabling	Lifespan and efficiency of cables - [154][152][153][151]	х	х			
	Impedance	Reducing transmission losses - [178][139][179][180][181][182]		х	х		х
	Controllers Equipment	Losses in grid due wind penetration - [183][184][185]	х	х			
	Intermediaries	Impact of impedance and harmonic resonance - [140][144]	х				

2.3 Discussion

Quantifying losses has proven to be an efficient method to identify clear gaps and lead improvement priority decisions in equipment and systems. Even though wind power has many aspects that are not in control of operators, researchers present interesting and promising approaches towards minimizing the losses and improve the production and performance during operational periods. This shows that wind power has still many opportunities for improvements.

In order to ensure a simple and reliable adaptation and implementation of OEE in wind energy, certain assumptions were made. Firstly, unlike many manufacturing applications, wind power is characterized by uncontrolled inputs. As a result, climate features were considered an additional factor contributing to performance losses. In addition, controllers were assessed separately due to the number of factors that they can influence. Finally, about quality, all losses between generator and grid were included, from grid requirements to distribution. For that reason, the six main losses of an equipment can be extended to nine in wind energy assets, as shown in Figure 2-3. It is important to mention that some losses could be classified in different items, but, for efficiency of the tool, the main attention needed is to not consider the same loss twice, following a linear reasoning.



Figure 2-3 Main Operational Losses in Wind Power

While some papers focus on investigating and assessing the losses, some of them propose solutions focusing on minimizing or identifying failures and losses. Typical possible solutions discussed in the papers found by the review activity, could be summarized as follows:

• Understanding lifespan, failure rates, and behaviour of the turbine and its components,

- Machine Learning to identify causalities,
- Increasing performance and minimizing losses through controllers' settings (considering the whole farm instead of individual turbines to reduce wake effects),
- More accurate wind regime prediction, especially short-term, for decision making, including maintenance scheduling and avoidance of curtailment,
- Controllers' settings and the use of energy storage to minimize losses due to fluctuations that can also affect transmission system and grid availability.

From the solutions proposed, most could be implemented during the operational phase, which indicates that regardless of the project design or if the wind farm has already started its operation, developers and operators could still improve their productivity. In addition, some other manufacturing tools could be used to reduce losses. As an example, according to [52], when a failure occurs in an offshore turbine, on average 22% of time is spent for the actual repair activity, while the rest is due to organization, waiting for suitable weather and spare parts management. The papers mentioned in the review investigated how to reduce logistic time and scheduling; however, they are not suggesting solutions to reduce the repair time itself. So, tools such as the Single-Minute Exchange to Die (SMED), in which changeover during maintenance could be reduced drastically, could also be very beneficial to wind power installations. To identify the need of further tools, OEE is pivotal to quantify and identify these gaps, according to a TPM strategy.

As mentioned before, OEE can be used in many situations, such as for comparing before and after changings in the process [186], simulating which scenario has potential for achieving better results [187] or encouraging the continuous improvement culture [188]. The main advantages of using OEE are first, its simplicity and, secondly, the overall analysis, with all possible operational losses included in one single index. To exemplify this last advantage, back in Section 3.2, one of the solutions found to reduce wake losses is through yaw control system. Nonetheless, some researchers used the same yaw system to increase

quality performance by reducing ramps. These are two different outcomes to be managed by the same actuator, where one can affect another. There is a study which suggests an optimum point on these two losses [77], however, it is not clear if these interventions can also affect availability. With that in mind, OEE seems to be a great solution to find overall improvements.

Another important observation about the decision making through OEE is that it can, and should, be related to the financial perspective. Even though it is out of scope of this study, any improvement suggested should find a balance between increased production and extra costs, since the final objective is to reduce LCOE, keeping equipment reliability and power quality high. Some papers, indeed, have proposed new equipment, gadgets, or more intrusive solutions, however, most of those presented in this review focused on changing control principles or using algorithms such as machine learning to find a better performance scenario, which probably does not require significant investments. Due to computational developments, improved processors, and the large amount of reliable data, machine learning has brought numerous possibilities to improve wind sector. In relation to OEE categories, machine learning can enhance availability and consequently profitability by improving wind predictions. It aids performance optimization through more reliable monitoring and effectively detecting abnormalities. And finally, for quality, machine learning algorithms can analyse sensor parameters and find patterns to improve power quality. Moreover, some studies present mixed algorithms, statistics and machine learning with OEE simulation [189].

To sum up, another three possible advantages of using OEE can be identified as follows. First, finding the actual OEE and tracking when the best rate was achieved can help operators and researchers to better understand the equipment. Second, the OEE tool considers that any equipment is unique, which means that the tool is conceptually tailored for each turbine particularities and wind farm location. Finally, some components reduce their performance before breakdown [58], so monitoring OEE has a potential preventive behaviour, by

detecting problems that could potentially affect performance, but do not trigger any fault signs, warning operators for upcoming failures.

2.4 Chapter Summary

In this chapter, the concept of OEE was briefly explained together with its basic equations. A literature review was performed to identify main losses in wind energy assets. Then, these losses were classified in three different elements: availability, performance, and quality, following OEE concepts. Differently from manufacturing, wind energy can have more causes of losses, therefore, an extension of main losses causes was proposed as showed in Figure 2-3, following the assumptions contained in the flowchart in Figure 2-2. In addition, some of benefits of using OEE was discussed in Section 2.3.

3 QUANTITATIVE ANALYSIS OF OEE IN WIND ENERGY

Estimating the operational losses in wind energy assets can be a challenging task, due to the number of uncertainties and particularities that each project carries. The model and size of the turbine, generator type, external environment the turbine was installed, resources available for O&M, experience of the O&M team, type of contract and benefits, regulation of the country, and the capacity and resilience of grid are examples of aspects that can affect its overall productivity. In this chapter, each element of OEE will be discussed and some figures found in the literature will be presented. Later, in the section 3.4, the rates investigated in previous sections will be summarized. Section 3.5 will present a real case scenario and its OEE rate will be investigated.

3.1 Availability Analysis

Although availability losses in wind energy are widely discussed in both industry and academia, there is no consensus on how to account for or treat them. Various methods have been proposed in the literature to calculate this rate, with the production-based and time-based methods being the most common. Other methods that are not still widely disseminated, such as the monetary based method proposed by [190], will not be discussed here. Also, in accordance with the OEE concept, production-based is closely related to the performance element, which will be discussed in the section 3.2. Therefore, this work will focus on time-based method to estimate availability.

As mentioned in the section 2.2.2, the basic formula to calculate availability index needs the planned time for operation and the downtime. This is similar to the time-based method, perhaps the most popular way to calculate availability in any industry. A version of this method is presented and suggested in the standard BS EN IEC 61400-26-1:2019 for wind deployments. The equation (3-1) illustrates this Availability Time Based (A_{TB}), where the uptime means the equipment is available, or operating, and downtime, the equipment is unavailable.

$$A_{TB} = 1 - \frac{Downtime}{Uptime + Downtime}$$
(3-1)

Another alternative to calculate availability, also derived from time-based method and quite presented in the literature, is considering the popular maintenance metrics: MTTR (Mean Time to Repair), MTBF (Mean Time Between Failure), and MTTF (Mean Time to Failure). Figure 3-1 illustrates how these metrics can be connected to uptime and downtime concepts. This method is useful especially for the number of studies that investigates failure rates (which is the inverse of MTBF) and MTTR from main important components in wind turbine in different situations. The formula to calculate the availability considering maintenance metrics (AMM) is shown in Equation (3-2).



Figure 3-1 MTTR, MTBF, and MTTF

The two versions of time-based method are simple to be used, however, they can mislead decisions due to different assumptions. First, considering Equation (3-1), the definition of downtime needs to be clarified. According to the same standard, downtime should include not only maintenance activities, but also forced outage or shutdowns due to external factors. This would include, to name a few, high wind stops, extreme low or high temperature, and grid stops [191]. However, some studies do not account them, one possible reason for this is that, as these problems are not always operators' responsibility, including them would not measure the real efficiency of the O&M team.

Similar problem is found in the second version of the method. It is clear by the Equation (3-2) that only corrective maintenance is directly accounted. This could be justified by two reasons. First, since preventive maintenance can be planned, operators try to work in periods where is less likely to wind, so this would not affect productivity, although the costs can increase due to high demand of vessels. Second, an effective preventive maintenance and failures monitoring could increase MTTF and reduce MTTR, improving availability rate indirectly. Another problem is what is considered in the MTTR, while some authors consider all period where the equipment was not operating, including waiting for spare parts, logistic, weather condition and team availability, others consider only the repair activity itself. In their review, [192] found these incongruency in some data base, especially for offshore projects. This can affect viability analysis and losses estimation in wind energy projects. To exemplify this, the AMM was estimated to be around 97%, considering only the MTTR and failure rates provided [10]. However, the authors included 250 hours logistic time to certain activities and included an estimated calendar-based maintenance. This made the index plummet to about 92%, the value they used in their economic simulation. According to [52], 22.2% of the downtime for offshore is actually spent on repair activity itself, while for [192], the rates are 40% and 20% for onshore and offshore projects, respectively. Therefore, it is important to always consider the entire period the equipment is not available to work.

As mentioned in the introduction of this chapter, the particularities of each project can directly affect the availability. To demonstrate this, Table 3-1 presents a summary of downtime distribution by component of onshore wind energy studies in three different countries, Turkey, Japan, and China. For the situation where more than six components were presented, only the first five were included and the rest was put together as "others". The Japanese study shown considerable low availability when compared to the others. The authors have investigated and compared with several popular databases and found that although their failure rate was similar, their downtime was three times higher. This was due to the high dependency of the importation of spare parts and, in some cases, due to Agricultural Land Act in Japan, which can delay activities by nearly one month, in case they need to use cranes.

[193] - Turkey		[194] - Japan		[195] - China		
Categories	Values	Categories	Values	Categories	Values	
Electric System	29%	Electrical System	38%	690V Cable	29%	
Pitch System	20%	Blade	25%	Pitch System	15%	
Structure	15%	Generator	18%	Control System	14%	
Yaw System	12%	Pitch System	4%	Converter	8%	
Controllers	10%	Gearbox	4%	Generator	7%	
Others	14%	Others	11%	Others	27%	
Availability	96.8%	Availability	87.4%	Availability	98.8%	

Table 3-1 Summary of downtime investigation for onshore projects.

The Turkey study has presented the results of the analysis of the first two operational years from a real onshore wind farm. Considering that the average of onshore projects is close to 98% [28], the availability of this project is a bit lower. This could be due to early life of the project. As shown in Figure 3.2, the failure rate behaviour along a lifetime of an equipment is similar to a bathtub shape, which is usually higher at the beginning of the operation, due to "infant mortality". After this period, the equipment gets constant failure rate most of its life. Conversely, some studies have shown that wind turbine failure rate increases slightly during this period [196], but this will be discussed in chapter 5. Finally, due to wear out and age, the failure rate grows in an accelerate rate, indicating its end of life. Another important aspect to notice is that key components as generator and gearbox have not presented any failure or malfunctioning during this period.



Figure 3-2 Bathtub curve

The main reason for downtime in the Chinese case, presented in Table 3-1, was "690V cable", which is not a common situation for wind energy projects. Nonetheless, this exemplifies how unusual problems could affect availability. The authors estimated that 87% of cable downtime was caused by third part, especially theft attempts of the neutral conductors. Another important aspect is that the study analysed only 6 months data, this could be one of the reasons why this loss got very evident and why the availability was near to 99%. Also, no preventive maintenance was recorded in the period. However, for availability estimation is important to always consider reasons the equipment was forced to shut down.

Although it has been shown that availability rates can vary significantly depending on the method selected, how the losses are accounted, particularities of each the project, resources accessibility and expertise of the team, large databases indicate similar availability rate. Obviously, for onshore and offshore projects, these figures are different. It is important to keep in mind that what is called offshore projects, includes fixed based and floating structures. While the first is closer to the shore, which reduces its weather dependency and consequently its costs and losses, the latter is farther from the coast and dependent on weather condition, which increases costs and losses. This been said, according to [191], considering the age the average losses vary from 1 to 6%, and the average

availability is 97%. [197] investigated several databases and the average found was 96%, however, recent deployments in Germany are performing on average above 98.5%. Conversely, offshore deployments are considerably lower. [198] and [199] considered 90% availability rate, while [37], [10], and [200], a fair estimation would be 91.2%, 92.2% and 94.5%, respectively. A study developed by [201] demonstrated solutions to improve offshore availability, achieving 96% in their model. However, [202] shown that, considering the entire life and the size of the farm, offshore availability can be as low as 82.5%.

3.2 Performance Analysis

In the context of OEE, the performance of an equipment should be the real performance over its theoretical or maximum one, as demonstrated in the equation 2.3. This can be compared to the production-based availability index, suggested by the standard BS EN IEC 61400-26-1:2019, where the rate would be the actual production, frequently called as net production, divided by the potential production, known as gross production. Nonetheless, gross and net production can vary depending on the reference, what makes the values fluctuate considerably. A mix of the several rates will be discussed in this section, however, since one of the goals of OEE is measure production stability, alternative methods will be discussed in the section 4.2 and 4.3.

The climate condition, first main cause of performance losses discussed in the section 2.2.2, can be controversial. While some researchers do not consider them by not being fully responsibility of operators, others consider as a way to compare performance and indicate opportunities to improve performance. The author of this work believes that the latter would be the best. To exemplify this, by comparing the performance of a real farm, [133] discovered a reduction in performance when ambient temperature was above 35 degrees, as the equipment reduces performance for system protection. Rather than accepting this issue, they conducted an investigation into how to improve nacelle conditions, leading to the changing on the specification of some filters that allowed for an increase in temperature protection settings. As a result, yearly production improved by 1.2%.

Considering all aspects that affects performance is important not only to increase the production, but also to identify them. Icing, blades erosion, dust, and insects' contamination can decrease lifetime of components, due to imbalanced operation, and reduce power output [191]. [133] found a reduction in production of up to 1.45% in the wind farm assessed due to excess of dust in the nacelles and soiling in the turbine blades. [203] found a reduction in the maximum lift coefficient up to 35% due to insect contamination. [85] proposed a method to reduce losses from icing accretion through changes in the TSR (Tip Speed Ratio) and achieved 7 to 23% reduction. Other climate conditions can also impact aerodynamic performance, leading to operational losses. For example, according to [83], rain can reduce performance by up to 5% in certain conditions. In more severe scenarios, depending on the rain and wind speed, this impact can increase to as much as 25%.

Wake effects, which is the turbulence or downwind speed reduction caused by operation of nearby turbines, is another controversial example of losses not directly related to operators. While some authors do not account these losses to performance, others measured and found interesting solutions to improve overall productivity. By considering the entire farm, instead of individual turbines, [76] developed a model causing intentional yaw misalignment to increase production by 2.73%. [121] presents a quantitative review on wake effects problems and available solutions in literature, where some studies achieved an improvement above 20% in losses reduction due to wake effects.

Besides the climate conditions, performance can be affected by other factors, as shown in the table 2.2. Some of these losses can be unseen by operators if the performance is not followed closely. Although previous paragraph has mentioned a technique considering yaw misalignment to improve production, [204] states that a misplaced vane by 5 millimetres, could cause 5% annual energy loss and misalignments can cause up to 2% losses. [105] shown that an optimization of yaw increased 7% of productivity. [76] points that production losses due yaw can be underestimated, because they are optimized using static wake models and in

real life due to rapid fluctuations yaw correct position cannot be effectively tracked.

[130] investigated production losses in a real wind farm and found that 6% of losses was related to turbine errors, which include alerts or faults in the system. [94] included a motor in the system to reduce cut in speed (speed where turbines start producing energy), increasing total productivity by prolonging the working period and energy produced in early speed ranges. [205] uses fuzzy regulators to reduce torque fluctuations and increase productivity closely to 1.7%. Considering floating offshore wind turbines, [89] investigated several scenarios, and found extra losses close to 7% in power generation, due to wind-wave misalignment.

To define the performance efficiency, many aspects need to be accounted. As mentioned earlier, these losses can be accounted for in different ways, making comparison difficult. While some treats gross production as the maximum theoretical provided by the wind, others include several losses reducing the gap between gross production and net production. Nevertheless, as demonstrated in this section, there are several operational solutions capable to improve wind farm productivity. Thus, not considering this can hide possibilities for improvement in the performance. For this quantitative performance analysis, studies that account both assumptions will be considered, and the difference and possibilities of improvement will be presented as the range of uncertainty.

To [37], 90% efficiency would include all losses by wake effects, transmissions, and other performance losses. To [199], the losses only due to performance would be 11.8%. [206] estimates the net and gross production from offshore projects and found that performance fluctuates from 95% to 98%. However, they also calculate scenarios including 10% uncertainty, which drop this rate significantly. [207] consider 3% performance loss for onshore projects. For [208], the average loss of offshore projects is 16.5% (from which half is due to performance losses) and 15% to onshore.

3.3 Quality Analysis

Different from manufacturing industry, energy is not a tangible integer product, what makes quality estimation a difficult task. When compared to other means to produce energy, renewables, which includes wind energy, has some particularities. First, wind energy is considered a non-dispatchable source, which means that operators cannot fully control the output to match electricity demands [209]. Another problem is related to the high and unpredictable fluctuations of its production, due to weather dependency. Since the grid operator tries to find balance between production and demand, unexpected energy fluctuations could cause undesirable disturbances in the grid. And finally, wind energy provides electricity through inverters, which do not offer inertia to the grid, and therefore, do not allow a time lag for grid adjustments.

It is important to understand that energy relies on the balance between production and grid demand. In order to keep the grid stable and reliable for consumers, there are some requirements that need to be followed. Otherwise, the energy produced can be rejected or, in some cases, the producer can be penalized financially by causing disturbances in the grid. For that reason, grid operators usually limit the power penetration from wind energy from 20 to 35% [210]–[212]. There are some situations where higher penetration levels were reported, especially in short-term, but to this be more reliable and frequent, changes in the operating practices is necessary, including demand response or storage systems [19]. Power Quality is the common term used to check the stability of the electricity in the grid. Wind turbine can usually exhibit the following power quality problems [213][214]:

- Steady-stage voltage level fluctuations
- Voltage variations and flickers
- Transients
- Harmonics
- Power frequency variation
- Grid protections

It is out of scope of this work to describe all items cited before. Nonetheless, to simplify, each grid network has its tolerance level to allow wind energy penetration. Some grid network requirements in Europe can be seen in [215]. It is important to mention that before the construction of a wind farm, the project designer needs to consider the capacity and how weak or strong is the grid. According to [155], although 10% fluctuation in wind energy voltage is considered a usual limit, 0.3% variation can bring nuisance in weak grids. Also, the study developed by [213] calculates some of operational losses in a wind park located in the southern of Crete Island. They estimated a loss due to grid rejection equals to 4.6%. However, by simulating the same network and doubling the total wind parks, the rejection would be 27%.

Besides the problem of particularities in each project, to find a reliable quality losses index has other adversities. As mentioned before, power quality is related to the balance of production and demand in the grid. Therefore, most of studies that try to identify losses, compares power quality in the grid before and after wind energy penetration. [157] has developed a power quality index and simulated power grid and wind penetration in different situations, while [216] and [217] used ETAP (a digital twin software able to simulates a grid and its disturbance) to assess the impact of different wind turbines generators.

Another challenge in dealing with these losses is how they are treated. For quality losses in wind energy, the available energy can be either rejected or curtailed. Curtailment occurs when operators are forced to disconnect turbines or reduce production due to either lack of demand or grid availability. Many studies in the literature use the term 'curtailment' for both cases, but there are exceptions, as demonstrated in [21]. According to this report, out of the total 12.1% losses registered in 2020, 51.2% were due to curtailment, while the rest were due to wind energy rejection caused by issues such as poor quality or grid safety.

According to [20], China achieved an average of 13% curtailment during the period from 2010 to 2017, where its peak was in 2016, achieving 17%. In 2020, Ireland had 11.4% of its energy from wind deployments curtailed, while North Ireland, had 14.8% [21]. Other countries faced better this problem, in 2018,

Germany, UK, and Italy got around 4% curtailment, while in Spain, this value was close to 1% [218]. It is important to note that comparing the curtailment percentage across countries may be unfair, as they depend on the level of renewable energy integration in the system and how aggressive were the investments in renewable energy.

Transmission is another common reason for losses in energy industry. These losses are estimated in the designing phase, where a percentage of losses from cables, which usually represents the bigger part of losses [219], and intermediary equipment, are already included in the project. The voltage level of the network, type of conductors, or current transformation, and distance need to be considered as well. [151] and [153] are examples of studies that investigate the best layout considering the minimization of losses and costs in offshore projects. According to [213] transportation losses varies between 3 to 5%, however, when connection network is greater than 100km, this could be higher.

Nonetheless, considering the operational period, which is the focus of this work, these losses can be higher than expected. According to [154], deviations in power factor, voltage, and harmonic distortions could not only reduce the operational life of cable and intermediaries' equipment, but also lead to additional losses. In their study they simulated several scenarios considering different levels of disturbances, and in the worst-case scenarios, the losses tripled, when compared to base scenarios. [180] investigated the efficiency of transmission cables from far offshore wind farm, and showed 9% losses reduction, only by adjusting voltage levels.

In addition to the expected losses from transmission, power quality issues can also increase transmission losses. This is mainly due to impedance in the line, which is directly affected by harmonic levels [140]. Furthermore, high capacitive impedance can lead to a high quantity of reactive power and cable losses, according to [220]. The difficulty in correlating losses with their causes arises because power quality can be affected by external aspects, such as turbulence [221], or abnormalities in the turbine, as shown in [222], where power quality was compared with CMS (Condition Monitoring System) signals.

3.4 OEE Analysis

In this section, the average OEE expected for onshore and offshore deployments, according to the losses and values discussed in the previously sections, will be summarized and presented. Obviously, these values work as reference, since the real OEE will depend on the particularities of each project. Some extreme scenarios as the Japanese availability of 87% or the 17% curtailment loss in China will be treated as outliers, therefore, they will not be included.

Figure 3.3 shows the expected rates of OEE and each element, considering the most frequent value as the average and the range is a summary of what was discussed in previous sections. For availability, 98% is a reasonable value for most onshore deployments, however, in some scenarios this can reduce to 95%. Although in the literature authors includes different loss in their assumptions, 93% for performance seems a fair rate. The range was defined by more optimistic values, which generally ignores the wake effects, and the improvements estimated in alternatives proposals to increase productivity. Quality is the most difficult to find reliable information, however, 95% seems a reasonable rate, considering wind energy rejection, and further electrical and transmission losses. The limits were defined by removing part of wind rejection for the best scenario, while the worst included 3% curtailment. Therefore, the average OEE for onshore is around 87%, which is higher than average world benchmark, 85%, discussed in section 2.2.1.





As expected, OEE for offshore is lower than from onshore. As shown in Figure 3.4, the average availability for offshore is 92%. As demonstrated before, some studies proposed ways to improve it to 96%. Although some studies show rates below 90%, they are not very frequent in the literature especially for new deployments, so 89% was set as the likely worst scenario. About the performance, it is expected that the losses level is higher, due to the harsher environment. As demonstrated by [29], the wet and more corrosive environment contributes to higher failure and misinterpretation of sensors, and the wind-wave misalignments can reduce the performance [89]. All this will have impact not only on performance, but also in the quality. The higher level of instability, vibration, and loads can affect Power Quality, as discussed in section 3.3. Another possible reason for higher losses level in quality is the longer transmission lines, so in total, the lower level considered was 88%.



Figure 3-4 Analysis of OEE for offshore and its elements according to literature.

3.5 Case Study

In this section, the OEE of a real onshore farm will be assessed. For confidentiality agreement, the identification and technical details cannot be disclosed. The farm has 17 wind turbines and two different datasets already filtered from one entire year were provided, that were already pre-processed and ready to be used. First, a weekly summary of gross production, net production, and average wind speed in each wind turbine is presented. The second spreadsheet is the list of faults and downtimes, which contains date and time the failure occurred, the category, description, if causes downtime in the turbine or if it was only an alert, and finally what sort of intervention was needed. Table 3.2 summarizes the content of both datasets provided.

 Table 3-2 Summary of Dataset

Dataset	Spreadsheet 1	Spreadsheet 2
Туре	Average per Week	Per Fault or Event
Content	 Week Number Turbine Average Wind Speed Gross Production Net Production 	 Date & Time Wind Turbine Event Category (Ambient, Grid, Turbine, or User) Log Number & Fault Description Type of Intervention (Follow, Error, In Service) Sum of Alert Time Sum of Downtime

To assess the availability the "sum of downtime" was used, considering the A_{TB} (Equation 3.1). For this analysis, the "Grid Events" was considered as quality losses, since it is not clear what really happens when the production in generations stops due to grid. In the case of the turbine is only disconnected from the grid and keeps running, consuming its components operational life, this cannot be considered as availability loss. Therefore, Figure 3.5 presents the total downtime (close to 2120 hours) in the first year considering typing of intervention. To make easier to understand, all operation that was not due to failure was called as scheduled Service. The Ambient category includes high wind speed, safety stop activated without failures related and untwist of cables. Figure 3.6 presents the number of alerts activated that did not interrupt the production, which was around 5500 hours. Although, this would not affect the availability itself, it is important to follow them, since this could explain some losses performances.







Figure 3-6 Fault Alerts and Failures Downtime Breakdown

Since the production data provided was collected on a weekly basis, there is a risk of bias in the estimation of the system's performance. As mentioned earlier, an accurate performance calculation should consider the maximum production achieved under identical conditions. However, the provided dataset lacks certain elements necessary for a more reliable analysis. Nevertheless, this method was still employed to maintain consistency throughout the analysis. A more comprehensive discussion on performance calculation methods and their limitations will be presented in Chapter 4.

In Figure 3-7, the manufacturer power curve is presented, along with real data and maximum performance. The performance index was calculated by comparing the real production to the maximum production, while accounting for availability on a weekly basis to correct for any discrepancies. Additionally, the difference between gross production and net production, as provided in spreadsheet 1, was taken into account during the calculations. An outlier that could potentially impact the analysis was identified and subsequently removed. It is important to emphasize that the weekly data provided is far from ideal and may significantly affect the analysis and the shape of the maximum curve. Nonetheless, this example can serve as a valuable reference for future studies.



Figure 3-7 Production Real Data - Outlier highlighted.

As explained before, since it is more likely that the turbine keeps running during grid events, this was considered as quality losses. The first chart in Figure 3-8 illustrates the total downtime caused by grid events and its breakdown by reason. The second chart shows the fault list, which summed five times more hours than the downtime itself. It was not provided any information about wind energy rejection or curtailments, so further analysis was ignored. Moreover, the project estimates 2.5% transmission loss, and this was included considering the energy after all losses calculated.



Figure 3-8 Grid Losses Downtime Breakdown and Fault List.

Figure 3-9 shows the OEE results for this case scenario, which were in line with the values discussed and presented in the section 3.4. The availability rate reached a high level of 98.6%, indicating that the equipment was operating effectively and with minimal downtime. The performance rate, subject to some limitations in calculation, scored 94.6%, still above the literature average of 93% (Figure 3-3). Finally, Quality was slightly above the literature upper limit of 97%, but only included grid faults and transmission. Further losses such as curtailment or wind rejection could bring the value to the estimated range for onshore projects. The final OEE measurement for the assessed year was 90.7%, indicating high productivity and efficiency.



Figure 3-9 OEE Analysis of Wind Farm Case Study.

3.6 Chapter Summary

In this chapter, a quantitative analysis of wind energy of each element of OEE was discussed. The usual formulas and losses rates were presented. This activity showed how hard estimate and monitor losses can be. Different projects and studies calculate their rates considering different equations and assumptions. Nonetheless, an attempt of putting all figures together was presented in section 3.4. Figure 3-3 presented the range of each element and the total OEE expected for onshore deployments, while Figure 3-4 for offshore. Section 3.5 presented a OEE analysis of a real farm in its early age. The result was coherent with what was found in the literature, despite of the lack of detail in part of the data provided.

4 MONITORING and EVALUATING WIND FARMS

It is important to monitor and evaluate wind farms effectiveness as frequent as possible. This could help to identify abnormalities and also to understand better the equipment. As mentioned before, the estimation of losses can vary between different projects, and this is not different for monitoring. One of the main objectives of OEE is to measure the stability of the system. So basically, best performance in same operational condition should be used for this activity and this is what will be proposed in this chapter. In section 4.1, a simple performance index will be defined, and traditional machine learning algorithms are applied to predict and monitor performance and power output through MET MAST DATA. In section 4.2, a data analysis considering the SCADA DATA provided and its OEE analysis will be performed. For this, some assumptions to calculate OEE elements had to be done, including a proposed method to estimate wind rejection by the grid. Finally, section 4.3 will present alternatives ways to monitor the performance.

4.1 Predict and Monitoring Performance through Machine Learning²

The most common way to monitor and predict wind power output is through the power curve, which in most cases is provided by the manufacturer. As defined by BS IEC 61400-12-1-2017, the power curve is based on the average power produced in a predetermined wind speed bin [109]. Although very useful, these curves do not usually consider the external features and some of the possible operational losses. Therefore, during the design phase a rate is considered in order to calculate the net production. According to Ioannou et al. [37] 90% is a reasonable estimation to be used in the design phase. However, in the operational phase, this approximation does not help operators to understand and identify what is causing the operational losses and fluctuations in production.

² This section is based on publication by Sathler and Kolios [112].

Wind energy output is mainly calculated through the amount of Kinect energy flux from the wind taken by the rotor, considering the density of the air, wind speed, rotor area and the power coefficient. The power coefficient is what determines how much energy can actually be captured from the wind and it is related to some wind features and rotor features, which include the tip speed ratio and blade pitch angle [223]. According to Betz's law, the theoretical maximum possible performance is equal to 16/27, i.e., 59.3% of the kinetic energy in wind.

External factors such as climate, wind conditions and topography can clearly affect the outcome and could be the reason for high fluctuation. Recent studies have been trying to create alternative curves to increase accuracy in prediction and comprehension of production. In the literature there are studies creating curves adding more inputs, such as air density [75], humidity [81], wind direction [71], turbulence [74], and periods of the day [80]. Also, machine learning has been largely used to predict wind power output, as shown in [224], [73], [158], [225] and also to create a model of day-ahead prediction [35], very important for market biding. Even though these models are very useful and beneficial for operators, the great number of curves can make decision making more complex, thus, they do not seem to offer a totally tailored approach.

A solution given by Sathler et al. [110] is the creation of a new way to monitor performance and track fluctuations, where the variance of production is calculated by dividing each value by the maximum registered on the related bin. As a result, the index itself has proved to have similar efficiency in predicting production and, when used together with wind speed, the results were higher than 96% when considering the entire farm. The results also showed that some periods with similar wind can have drastically different production when relying on the index created. Even though these are promising results, the study does not bring solutions for calculating performance in an effective way.

As mentioned before some of possible reasons for fluctuation in production are due to external factors. To monitor the external factors, farms have a separate tower with sensors to measure wind and climate features. Therefore, the goal of this study is to check if the information provided by this tower, also known as Meteorological Mast Data (Met Mast Data), can be used to predict the performance index of the turbines and check if they provide a reliable power output prediction to assist operators in monitoring their turbines. To execute all the steps outlined in the algorithm, including pre-processing, method selection, hyperparameter optimization, and final model creation, Python 3.8 was utilized along with the open-sources libraries: NumPy, pandas, and scikit-learn. Figure 4-1 summarizes the process proposed in this work and the following sections will explain each step of the pipeline in details.



Figure 4-1 Flowchart of the model proposed.

4.1.1 Methods

Machine learning has become very popular in recent years; this is due mainly to the advances in technology, especially those related to storage systems and faster processing of data. There are many types of algorithms and methods inside machine learning, which can help to find patterns in their outcomes, connecting inputs to outputs to make predictions with lower errors. In this study, regressionsupervised models will be used. They are classified as supervised because the algorithm is trained based on historical data previously available, and as regression since production and performance are considered quantitative continuous values.

It is considered a good practice to start with simpler methods before testing more sophisticated ones to avoid waste of computational resources. According to [226], decision trees, support vector machines and naive bayes are the traditional methods for supervised learning. Therefore, to design both models proposed, production output and performance prediction, four traditional methods were chosen: Linear Regression (LNR), Decision Tree Regression (DTR), Support Vector Regression (SVR) and Random Forest Regression (RFR). Table 4-1 gives a short explanation on each method with references for further interest.

Model	Model Description	
Linear Regression (LNR)	Linear Prediction method. Look for best fit with lower errors, considering straight linear equation.	
Decision Tree Regression (DTR)	Decision Tree Regression (DTR) It is a nonlinear supervised prediction method, which creates conditional statements.	
Support Vector Regression (SVR)	Based on Support Vector Machines, it uses, hyperparameters with a tolerance to minimize errors.	[228]
Random Forest Regression (RFR)	Construction of multiple decision trees. A probabilistic analysis is made among those decision tree to select the best prediction.	[229]

	Table 4-1	Description	of Machine	Learning	Methos	Selected.
--	-----------	-------------	------------	----------	--------	-----------

The goal behind of the selection of these methods is to keep the models as simple as possible, avoiding the need of high computational resources, as mentioned before. However, each of them has its own set of pros and cons. LNR is a simple and interpretable method that provides insight into the relationships between variables, but it may struggle to capture complex nonlinear patterns. DTR, on the other hand, is capable of modelling intricate relationships and handling both numerical and categorical features, yet it can easily overfit the data and lacks interpretability for larger trees. SVR excels at handling high-dimensional data and can effectively handle outliers, but it can be very challenging and time consuming with regards to finding the right hyperparameter combination. Finally, RFR combines multiple decision trees to mitigate overfitting, provides feature importance rankings, and handles a wide range of data types, but it may be more challenging to interpret compared to individual decision trees.

The selection of a specific technique depends on the specific problem at hand, the available data, and the desired balance between interpretability and predictive performance. Although other traditional methods as Gaussian Process has proved to be a good alternative for wind energy assets [230], due to the amount of data and inputs used here this method was discarded. Gaussian Process involves inverting matrix which makes it computationally expensive as the dataset grows larger. Also, as demonstrated in the section 4.1.2.2, the methods selected provided accurate predictions, which makes the use of more advanced methods unnecessary for the aim of this activity.

A normal problem in machine learning is the selection of the data, which can affect results positively or negatively. So, to avoid any misleading in the conclusions or any bias due to data selection, k-fold Cross Validation (CV) will be used. In this method, the data are split into "k" equal parts, where each of these parts is used as a test set, while the rest of data is used as a training set, so the model runs k-times. This process provides "k" different outcomes, so further statistical analysis will lead to evaluating each regression model, and the one that fits best for the purpose of this study will be selected.

Finally, to assess the accuracy of the method proposed, four different metrics were selected. These metrics were not only the most popular ones for evaluating regression accuracy but also offer unique perspectives when used together. First,

the coefficient of determination, also known as R-squared (R²). It measures the extent to which the dependent variable explains the independent variable. R² is calculated by considering the squared sums of the differences between predicted and observed values and the total sum of squares, as shown in formula 4-1. The results vary from 0 to 1, where 1 means perfect correlation, in other words, the dependent variables explain 100% of the variance in the independent variable. Conversely, a result equal to zero means there is no correlation.

$$R^2 = 1 - \frac{SSR}{SST} \tag{4-1}$$

where SSR is the sum of squared residuals and SST total sum of squares.

The other metrics are Mean Absolute Error (MAE), Mean Percentage Absolute Error (MAPE), and Root Mean Squared Error (RMSE). As suggested by its name, MAE calculates the average absolute difference between predictions and actual values (formula 4-2). It is less sensitive to outliers compared to other metrics. Similarly, MAPE (formula 4-3) considers absolute values, but also divides the error by the actual value, providing a ratio that reflects its accuracy. On the other hand, RMSE (formula 4-4) penalizes poor predictions by squaring the errors, offering valuable insights into the accuracy of the model. In equations 4-2 to 4-4, y_i is the tested value and \hat{y} is the predicted one.

$$MAE = \frac{1}{n} \sum_{i=1}^{n} 1 - \frac{SSR}{SST}$$
(4-2)

$$MAPE = \frac{1}{n} \sum_{i=1}^{n} \left| \frac{y_i - \hat{y}_i}{y_i} \right|$$
(4-3)

$$RMSE = \sqrt{\frac{1}{n} \sum_{i=1}^{n} (y_i - \hat{y}_i)^2}$$
(4-4)

4.1.2 Data Analysis

To test the model proposed, SCADA (Supervisory Control and Data Acquisition) data from five onshore wind turbines of 2MW and Met Mast Data from the farm tower were used. For confidentiality reasons the location of the farm and model of turbines cannot be disclosed. It was provided information from 172 days with 10 minutes range, totalling 24,768 records per turbine. From the SCADA data was retrieved production and wind speed. The Met Mast Data, on the other hand, includes wind speed measured at five different heights, direction of the wind at four different points, humidity and temperature, both from two distinct points, and finally ambient pressure.

4.1.2.1 Pre-processing

From these data some abnormalities, such as negative outputs, incomplete data, and periods where wind speed was out of the production range were removed. Even though the cut-in speed of this turbine is 3.5 m/s, recordings below 5 m/s were not considered, because of the high fluctuations caused by the starting-up of the turbine. Production below 100kW was eliminated as well for the same reason. Hence, these values are more likely to be an error and could mislead the model. The goal of this work is to create a model to predict production and performance. Therefore, outliers can minimize the accuracy of the model. Considering that, an interval of confidence of 99% was calculated and the values out of this range were considered to be outliers.

As mentioned in the introduction, the performance index was calculated by the division of each validated data to the maximum production recorded in the same bin. The wind speed range of 0.5 m/s was selected for this study, following the recommendation from BS IEC 61400-12-1-2017. From now on, all the processes will be presented only considering the first wind turbine generator (WTG01) in order not to be repetitive. But the process explained here is the same applied to all turbines and in the same farm. As a result, from the original 24,768 recordings provided, 15,141 were used to assess the methodology proposed after pre-processing in WTG01.
Figure 4-2 illustrates the pre-processing evolution in three different stages. The first graph represents the total amount of data provided; it can be seen that the production fluctuates significantly during the period assessed. In the second graph, the first criteria for data reduction, wind speed below 5 m/s and production below 100kW, was undertaken and the interval of confidence of 99% was calculated. In this graph it is clear how outliers, especially those above the upper limit, can affect results. Since the performance is a rate between each value and its maximum, the outliers would affect all values in that bin, creating distortions in the model. It is important to note that this outlier was occasional, since there are no other points around it in the range, which justifies its removal. Finally, the last graph includes a colour map with the performance index calculated. The black line is the power curve provided by the manufacturer.





4.1.2.2 Model Selection

In the model selection activity, basic procedures and parameters were used. Nonetheless, a deep investigation of parameters was developed to run the methodology proposed and it is presented in the subsection 4.1.2.3. Hence, in few words, to find the best LNR model, Ordinary Least Square method was used. To train the RFR, one hundred trees was used, and the kernel selected to the SVR was the "Radial Basis Function". In all methods square error was used as a metric to the loss function and the input features were scaled through Standardization.

To avoid bias, a cross-validation was conducted to select the optimal regression method. In this analysis, k = 10 was considered, and the metric used was R^2 . This study is divided into two different models: one to calculate the power output considering wind speed and performance (MODEL 1), and the other to predict the performance of production using Met Mast Data (MODEL 2). To provide a visual comparison of the results, a box plot from each method will be presented, illustrating the interquartile range, whiskers denoting the minimum and maximum values (excluding outliers), and individual data points. Additionally, a separate box within the graphs displays the mean value and standard deviation for each method, providing a summary of their performance. This concise representation facilitates a swift evaluation of prediction accuracy and variability across the different machine learning methods. Before defining the models, a similar procedure was done considering only wind speed to assess the advantages of including performance as an input, as illustrated in Figure 4-3.



Algorithm Comparison - CV K_Fold_5 - WTG01

Figure 4-3 Box Plot of CV of Power Output Prediction with only Wind Speed

Figure 4-4 presents the results of the models to calculate production considering wind speed and performance. Even though considering only wind speed has some good values (Figure 4-3), the addition of performance significantly

improved the accuracy in all methods. The high result is not a surprise since the performance index identifies the fluctuation of production in a certain bin, in other words, where exactly the production output will be. Although the visible curvature, LNR, obtained a high result, around 0.94, which illustrates and reaffirms the importance of the performance index as an input. DTR and RFR, which are nonlinear regression conditional models, achieved an exceptional correlation, with RFR demonstrating a slightly higher result (0.99988) compared to DTR (0.99980). This marginal difference suggests that the two models performed similarly, indicating a strong and nearly indistinguishable correlation.



Algorithm Comparison - CV K_Fold_5 - WTG01

Figure 4-4 Box Plot of CV of Power Output Prediction with Performance and Wind Speed (MODEL 1)

The biggest challenge of the model proposed is the prediction of the performance index. Figure 4-5 shows the results of MODEL 2, where the performance was predicted using Met Mast Data as input. In this scenario SVR and RFR had the best results, achieving an average R^2 of 0.866 and 0.892 respectively. Considering the best outcome, RFR, the result is very consistent since the standard deviation was lower, around 0.0049, which means that around 89% of wind turbine performance can be explained by the Met Mast Data or external interferences.



Figure 4-5 Box Plot of CV Performance Prediction through Met Mast Data Prediction Model. (MODEL 2)

4.1.2.3 Power Output Model

Based on the model selection analysis, RFR demonstrated superior performance compared to other methods in both proposed models, leading to its selection for predicting the performance index (MODEL 2) and calculating the power output (MODEL 1). RFR, with its versatile architecture and numerous parameter options, offers extensive possibilities for optimization. In this study, six widely adopted parameters used in optimizations were investigated. 'Bootstrap' enhances model robustness, while 'max depth', 'max features per leaf', and 'min samples per leaf' control tree complexity to prevent overfitting. 'Min samples split' captures meaningful patterns by determining the minimum number of samples required to split a node. The 'number of estimators' impacts overall performance and stability by determining the ensemble's tree count. These parameters are the most common and have the potential to enhance the accuracy and effectiveness of the wind energy prediction model. Further details on the parameters can be found in [231].

To tune the best set of parameters, two routines were created. Firstly, a range of commonly used parameter values was explored through random selection. Then, based on the best results obtained, a more focused range was retested, considering all possibilities. Each iteration used a fivefold cross-validation

58

approach. Table 4-2 shows the parameters and criteria assessed at each stage, along with the selected parameters. Figure 4-6 shows the box plot of MODEL 2 after the parameters redefinition, resulting in 0.65% improvement in average R². As MODEL 1 achieved R² equals to 0.999, no further analysis was conducted on its basic structure.

	Random	Single	Selecte
Bootstrap	['True', 'False']	['False']	['False'
Max. Depth	None & linspace (20,110,4)	['None', 20, 80]	['None'
Max. Feature per Leaf	[1, 2, 5, 10, 14]	[1, 2, 5]	[1]
Min. Samples per Leaf	[1, 2, 4]	[1, 2]	[1]
Min. Samples Split	[2, 4, 10]	[2, 10]	[2]
Number of Estimators	linspace(10, 1010, 11)	[110, 210, 410]	[210]
Total Possible Combinations	4950	108	-
Total Tested	1000	108	-

Table 4-2 RFR Optimization parameters – WTG01

Algorithm Comparison - CV K_Fold_10 - WTG01



Figure 4-6 Boxplot of MODEL 2 after RFR tuning.

Finally, from the total data provided, 30% was set aside to validate the model, while 70% was used to train the models. To increase reliability on the models proposed, increment results and reduce the risk of bias, the selection of the validation group was random, and the procedure were run five times. It is important to mention that the validation data were separated before the performance calculation. The goal behind this strategy is to emulate a real

scenario, where the performance index is calculated only with provided data and the performance index predicted is unknown. It is important to mention that to calculate the complete farm, the whole process was redone adding a new input variable to identify each one of the five turbines. In other words, "All Farm" scenario is not the average analysis of the five turbines individually, but it is a new simulation.

4.1.3 Results and Discussion

The proposed approach proved to be efficient to predict the wind power output. Table 4-3 presents the results from the five iterations considering turbine WTG01, where the predictions achieved an average of 0.24% error when compared to the real production. Regarding the metrics, the results were also very consistent, independently of the iterations. The R² was close to 100% and the average RMSE was around 77.9kW; since this is a 2MW turbine, this error is acceptable considering the benefit the model can bring. Gross production, i.e., the one considering the manufacturer power curve, was presented as a reference as well as the net production, considering 90.0% performance.

Table 4-3 Results from WTG1 with 15,141 data points (RP = Real Production, MP = Model Prediction, GP = Gross Production, NP = Net Production).

It	eration	RP (kWh)	MP (kWh)	GP (kWh)	NP (kWh)	R ²	MAE	MAPE	RMSE
1	Values	883,364.64	885,941.83	972,915.38	875,623.84	0.9852	52.6931	6.00%	78.1422
	Δ	-	0.29%	10.14%	-0.88%				
2	Values	884,423.00	886,833.65	976,478.52	878,830.67	0.9848	53.4556	5.97%	78.6284
	Δ	-	0.27%	10.41%	-0.63%				
3	Values	888,095.65	890,885.79	978,283.22	880,454.90	0.9851	52.9711	5.94%	78.3595
	Δ	-	0.31%	10.16%	-0.86%				
4	Values	881,308.79	882,689.29	971,379.26	874,241.33	0.9854	52.2486	5.85%	77.2820
	Δ	-	0.16%	10.22%	-0.80%				
5	Values	889,771.90	891,025.18	977,694.58	879,925.12	0.9857	51.4527	5.77%	77.1723
	Δ	-	0.14%	9.88%	-1.11%				

It is important to note that the net production has achieved a good result as well; however. this value is more appropriate for the design phase. By using a fixed rate, this value does not help the operators to understand the production behaviour, during the operational phase. As mentioned before, there are many factors that affect the turbine performance, many of which are related directly to the external factors and climate features. Therefore, the use of Met Mast Data to monitor performance and estimate production can be considered reliable. Some of the known operational losses are due to turbulence, air density, or wake effects, for example, and they can be linked to the differences in wind speed in different points, temperature, and wind direction, respectively, and all this information is provided by the Met Mast Data.

Another advantage of the model is that it gives extra information to operators about possible losses in the equipment's efficiency. Components of the turbine tend to reduce its performance before breakdowns [58], but considering the high fluctuation in production, it is hard to identify if and when this possible loss is due to wind features or an equipment issue. Monitoring the expected performance through Met Mast Data could work as a reference to operators to check if the fluctuation is normal, considering the environment characteristics, or if this could be a mechanical or electrical problem.

While the model seems very useful to monitor production and performance in real time, future predictions were not evaluated until here. Knowing that it is very unlikely to have accurate forecasts within a 10-minute range, as calculated so far, an additional simulation of performance prediction considering daily average results was done and the results are shown in Figure 4-7. From the 172 days of data provided, the month of January, i.e., 31 days, was separated out as a test set, and the other 141 days as training. In this simulation the R² of the performance prediction was 0.89 and RMSE was 0.04, which means the model can provide a good accuracy even considering larger ranges. This can be useful, especially in order to plan maintenance in advance, since in some periods the total production can be lower, even though the wind speed is the same.



Figure 4-7 Performance Prediction X Real Performance – WTG01 – January

Although it is possible to find in the literature similar results in terms of accuracy, the model presented in this study has the advantage of use less computational resources when compared to more advanced techniques, such Artificial Neural Network or Gaussian Process as mentioned before. For real-time monitoring this can become an important advantage. Also, since the model here proposed uses data from diverse sources, this can avoid redundancies and provides a new independent input to operators. Nonetheless, it is expected an improvement in the performance prediction with more data, a deeper analysis and elimination of abnormalities (fault alarm data were not provided by the operators in this study), and a feature analysis and reduction.

Table 4-4 presents the average results achieved in each turbine and when the entire farm is assessed together. Considering all simulations, the average error between the real production and from the model was 0.16%. The MAPE and the RMSE was respectively, 5.68% and 72.83kW. The entire farm obtained a slight better result, with an average RMSE of 64kW, which could be due to the amount of data or by the identification of the turbine as input. It is expected that the difference in individual performance can be better tracked when they are assessed together, especially considering the wake effect loss.

Param.	WTG1	WTG2	WTG3	WTG4	WTG5	All
Error	0.24%	0.07%	0.16%	0.08%	0.18%	0.21%
R ²	0.9852	0.9871	0.9863	0.9860	0.9858	0.9905
MAE	52.5642	49.4654	49.5505	51.1040	50.8078	41.4567
MAPE	5.90%	5.54%	5.71%	5.83%	5.93%	5.00%
RMSE	77.9169	71.1578	73.7286	74.6535	75.4473	61.5918

Table 4-4 Average Results from Simulations

To sum up, the use of Met Mast Data to predict performance can be very beneficial: First because they contain most of the external information that directly affects wind power production, so it could reduce the use of several curves, which assess these inputs individually, to only one model. Second, for the authors, the turbine behaviour is unique, which means that although known, the influence of each factor on production losses can vary from farm to farm due to the topography, wind conditions and climate. So, the use of the Met Mast Data related to real performance provides an increasingly tailored model. In addition, a daily average model was created and proved to be as effective as using a 10-minute range, which means the model can be used for short- or medium-term predictions, depending on the accuracy of climate and wind feature forecasts or a historical database. To end up, the model proposed proved to be very effective in helping operators with decision making as a tool to monitor performance and predict production in real time or for future predictions.

4.2 OEE Analysis through SCADA³

Different from previous section, where MET MAST data was used to monitor the performance, in this section this element will be assessed considering only SCADA data. The goal behind this is activity is not to predict performance, but measure in a reliable and coherent way the real performance and productivity of the wind farm. Additionally, each of the OEE element will be discussed and measured. Due to limitation of the dataset, some assumptions needed to be considered. To not be repetitive, only the first wind turbine will be demonstrated

³ This section contains extracts from the publication by Sathler et al [110].

in following subsections, however, in the subsection 4.2.5, a summary considering the entire farm will be presented.

4.2.1 Availability

The data provided did not show when abnormalities and failures occur, neither if the turbine was shut down by the operator or in test/maintenance mode. Although this affects availability analysis, all missing data will be considered as downtime as well as when the wind speed was higher than 5m/s and the production was below 50kW. These abnormalities can be seen in Figure 4-1a. To calculate availability, the time-based method (A_{TB}) was selected, according to the equation 3-1.

4.2.2 Performance

The power output from wind turbine is directly related to the wind features as discussed before. The equation to calculate the power output is:

Power
$$Output = \frac{1}{2} \rho A C_p(\lambda, \beta) V^3$$
 (4-5)

where,

 ρ is the air density [kg/m³].

A is the rotor area [m²].

 C_p is a performance constant, and it is related to the tip speed ratio (λ) and pitch angle (β).

V is the wind speed [m/s]

As shown in Equation 4-5, the power output is directly related to the air density. Three weather features can influence the air density, the local pressure, humidity, and temperature. These values are connected to each other, what makes the estimation of production through a simple power curve less effective. To correct this, a normalization of the measured values is recommended in the standard BS EN IEC 61400-12-1:2022. In this method, air density normalization is applied to the SCADA data provided before the calculation of performance. This method is

briefly described below. First, the relative air density is calculated, through the Equation 4-6:

$$\rho_{10min} = \frac{1}{T_{10min}} \left(\frac{B_{10min}}{R_0} - \phi P_w \left(\frac{1}{R_0} - \frac{1}{R_w} \right) \right)$$
(4-6)

where,

 $\rho_{10\text{min}}$ is the derived 10 min averaged air density.

 T_{10min} is the measured absolute air temperature average over 10 min [K]; B_{10min} is the air pressure corrected to hub height [Pa];

R₀ is the gas constant of dry air, 287.05 [J/kgK];

 Φ is the relative humidity [%]

R_w is the gas constant of water vapour, 461.5 [J/kgK];

 P_w is the vapour pressure equal to $2.05 \times 10^{-5} \exp(0.0631846 T_{10min})$ [Pa].

Then, the wind speed measured is normalized according to the Equation 4-6:

$$V_n = V_{10min} (\frac{\rho_{10min}}{\rho_0})^{\frac{1}{3}}$$
(4-7)

where,

Vn is the normalized wind;

V_{10min} is the averaged wind speed measured;

 ρ_0 is the reference air density, here was assumed 1.225kg/m².

Additionally, the Vn is corrected with nacelle position:

$$V_m = V_n \cos\left(\varphi\right) \tag{4-8}$$

where φ is the misalignment angle between the nacelle and the average wind speed and V_m is the equivalent wind speed.

To calculate the performance analysis, the maximum production in each wind speed bin (0.5m/s) was identified and a rate of real value over maximum was considered its actual performance. However, in this activity, two extra components were added to straight the gap and makes the maximum performance assessment more realist. Besides the wind speed bin, the wind

direction (WD30) and the wind turbulence intensity (TI) was considered. The first was grouped as 30° bin, while the later was calculated following Equation 4-8, according to BS IEC 61400-1:2019:

$$TI = \frac{\sigma_{10min}}{V_{10min}} \tag{4-9}$$

Where σ_{10min} is the standard deviation and V_{10min} is the mean wind speed calculated within 10 minutes. Some tests were performed before the classes was defined, following the impact in results and similar amount of sample to avoid bias in the analysis. Therefore, the three classes considered was:

- Class 1 TI < 5%
- Class 2 5% \leq Tl \leq 10%
- Class 3 TI > 10%

4.2.3 Quality Estimation

As discussed in subsection 2.2.4, there are various causes of quality losses in wind energy deployments, ranging from the power quality of generated energy to transmission loss levels. Unlike in the manufacturing industry, which produces tangible goods, energy involves some subjectivity, making loss estimations a more challenging task. Unfortunately, to this analysis only SCADA data was provided, which limits the scope. However, some of quality losses can be calculated by considering the difference in the production output.

One of the main problems of wind energy is its high level of fluctuation. As wind energy is weather dependent, operators have limited control over its output. Grid operators must maintain a stable frequency by balancing active power in the system, and extreme energy variations can pose a problem. This effect, also known as ramp rate, is widely discussed, and some countries have included requirements in their national grid codes to better address this issue [232]. High increases (ramp-up) and high decreases (ramp-down) are both problematic, but while in the first, operators can minimize the fluctuation by reducing power output through the pitch angles, the same cannot be said for ramp-down. Either way, in both cases, wind farms can suffer financial losses due to curtailment or rejection

of energy production or penalties, depending on the contract. Therefore, to estimate some possible quality losses, a ramp rate analysis will be conducted.

There are different ways to define the ramp rates, which could be by variation, duration, and/or change rate [168]. A rate considering the difference of power output in a defined period of time is the most common. Some grid codes define the acceptable rate of variance, usually 5, 7.5, or 10% [233], which allows an estimation of energy curtailed. Here, the grid limits were not provided, but to simulate some of these losses the ramp rate will be calculated considering the 10 minutes data available, and a deduction from the energy produced will be accounted as wind energy rejection. The following equation and rules summarize how the energy loss (EL) was calculated in the present simulation; the rated power considered was 5%:

$$\begin{cases} if |P_t - P_{t+10}| \le 0.05 \ x \ RP \longrightarrow P_{t+10} \\ if |P_t - P_{t+10}| > 0.05 \ x \ RP \longrightarrow EL = P_{t+10} \ x \ (1 - \frac{|P_{t+10} - P_t - 0.05 \ RP|}{RP}) \end{cases}$$
(4-10)

where *RP* is the Rated Power, P_t is the Active Power, and P_{t+10} is the Active Power measured 10 minutes later.

4.2.4 Partial Results and Discussion of WTG01

To simplify the analysis, Table 4-5 presents the summary of results per week. The number of the week is the official one, according to the calendar and it was sort in chronological order. The average OEE was 70%, which for an onshore turbine can be considered a low value, as discussed in section 3.4. However, as mentioned before, it was not provided further information about faults or mechanical problems. This information could explain the reason behind low availability from the weeks 27 to 34. Ignoring this period, the availability would be 98%, close to the average found in the literature.

Week	Average of Availability (A)	Average of Performance (B)	Average of Quality (C)	Average of OEE
26	1.000	0.641	0.966	0.619
27	0.706	0.755	0.928	0.495
28	0.522	0.816	0.946	0.403
29	0.307	0.862	0.939	0.248
30	0.612	0.895	0.963	0.527
31	0.704	0.802	0.965	0.545
32	0.870	0.768	0.952	0.636
33	0.906	0.765	0.960	0.665
34	0.486	0.732	0.977	0.348
35	1.000	0.757	0.959	0.725
48	1.000	0.915	0.975	0.892
49	1.000	0.876	0.985	0.863
50	0.982	0.868	0.976	0.833
51	1.000	0.790	0.987	0.779
52	1.000	0.868	0.969	0.842
53	1.000	0.771	0.989	0.763
1	1.000	0.850	0.969	0.823
2	1.000	0.852	0.983	0.837
3	0.897	0.849	0.974	0.741
4	0.923	0.853	0.964	0.759
5	0.921	0.834	0.976	0.750
6	0.979	0.808	0.969	0.767
7	0.996	0.874	0.965	0.840
8	0.999	0.833	0.960	0.799
9	1.000	0.763	0.962	0.734
10	0.922	0.803	0.963	0.713
11	0.998	0.882	0.955	0.841
12	1.000	0.812	0.967	0.786
Total	0.867	0.830	0.967	0.699

Table 4-5 Summary of the results of WTG01

Table 4-5 shows an average performance rate of 83%, which is below the average reported in the literature. However, unlike availability, performance did not fluctuate significantly over the assessment period. With the exception of weeks 32 to 35, where the performance was approximately 75%, the rate varied from 80% to 90% for most of the period. To illustrate one advantage of the proposed analysis, assuming that the low availability recorded in weeks 28 to 31 was due to maintenance and the performance below 80% was seasonal, the

overall productivity of the farm would be higher if the maintenance was done in this period, without any substantial changing in the project. In terms of quality, the values were consistent with the literature, although other losses such as transmission were not considered. While further testing is necessary to validate the criteria and rules used, the results are promising.

Other analyses can be conducted using the proposed OEE assessment. An investigation of performance variance revealed significant differences in situations with similar wind features. Table 4.6 exemplifies this observation, demonstrating that even with similar wind speed bin (WS), wind direction bin (WD30), TI class (TI_Class), and season (both were during summer and nighttime), the performance varied by approximately 23%. The average performance during the first period was nearly 100%, while in the second period, it dropped to 77%. It is important to mention that OEE helps to identify abnormalities in the system, but further investigation is necessary to understand the root cause of this performance difference. This information is crucial for identifying possible solutions to minimize such variations and optimize wind energy productivity, ensuring efficient and cost-effective operation. The data provided does not allow further investigation, which could include even production reduction due to curtailment. However, it is important to consider this potential loss in the OEE assessment to ensure the metric captures the overall performance accurately and accounts for any operational limitations or curtailment effects.

69

Month	Day	Time	ws	WD30	TI_Class	Average of Performance (B)
July	30	02:00	12.5	330	1	0.992836
July	30	02:10	13	330	1	0.999962
July	30	02:20	13	330	1	0.99995
July	30	02:30	13.5	330	1	0.994367
July	30	02:40	13.5	330	1	0.994346
July	30	02:50	13.5	330	1	0.994366
Month	Day	Time	WS	WD30	TI_Class	Average of Performance (B)
August	20	21:00	13	330	1	0.748819
August	20	21:10	13	330	1	0.75835
August	20	21:20	13	330	1	0.764873
August	20	21:30	13	330	1	0.772976
August	20	21:40	13	330	1	0.785902
August	20	21:50	13	330	1	0.793626

Table 4-6 Comparison of two different periods with different performance

Other patterns can be found when the elements are grouped according to different criteria. Table 4-7, 4-8, 4-9, 4-10, and 4-11 summarizes the results grouped by temperature range, month, TI class, wind speed bin, and wind direction bin, respectively. About the temperature criteria presented in Table 4-7, the results are clearer correlated, but when the temperature is above 35° C, the performance decay is more evident. According to the manufacturer's manual, above 35°C, the production is reduced for equipment's safety. This could also explain the performance variance when assessed per month (Table 4-8). As expected, higher turbulence caused lower performance and lower quality (Table 4-9), and the performance is higher when wind turbine works in its rated power (Table 4-10). Regarding wind direction presented in Table 4-11, only the angle bins 240 and 270 had a significant difference in the result, probably by the increased wake losses. It is worth mentioning that the rate considers the maximum in the same bin, so the reason for lower stability needs to be better investigated. Although the availability has varied among the tables, no conclusion can be done, due to the limited and size of the sample, as already shown in Table 4-5.

Temp (°C)	Average of Availability (A)	Average of Performance (B)	Average of Quality (C)
5-15	0.963	0.898	0.973
15-25	0.978	0.839	0.972
25-35	0.977	0.847	0.966
35-45	0.739	0.808	0.964
>45	0.614	0.752	0.945
Total	0.867	0.830	0.967

Table 4-7 Results of WTG01 grouped per Temperature Range.

Table 4-8 Results of WTG01 grouped per month.

Month	Average of Availability (A)	Average of Performance (B)	Average of Quality (C)
January	0.944	0.844	0.972
February	0.991	0.843	0.965
March	0.972	0.823	0.962
July	0.571	0.835	0.952
August	0.777	0.759	0.961
December	0.996	0.858	0.979
Total	0.867	0.830	0.967

Table 4-9 Results of WTG01 grouped per TI Classes.

TI_Class	Average of Availability (A)	Average of Performance (B)	Average of Quality (C)
1	0.919	0.834	0.977
2	0.906	0.844	0.962
3	0.762	0.784	0.938
Total	0.867	0.830	0.967

WS_5	Average of Availability (A)	Average of Performance (B)	Average of Quality (C)
5	0.939	0.735	0.981
5.5	0.884	0.768	0.977
6	0.911	0.751	0.975
6.5	0.900	0.771	0.970
7	0.883	0.783	0.962
7.5	0.905	0.784	0.958
8	0.888	0.787	0.947
8.5	0.877	0.790	0.946
9	0.885	0.775	0.939
9.5	0.901	0.794	0.945
10	0.919	0.863	0.966
10.5	0.865	0.930	0.978
11	0.839	0.960	0.985
11.5	0.831	0.969	0.990
12	0.796	0.972	0.995
12.5	0.806	0.967	0.994
13	0.785	0.961	0.996
13.5	0.798	0.943	0.989
14	0.816	0.925	0.979
14.5	0.798	0.948	0.987
15	0.745	0.954	0.994
15.5	0.667	0.960	0.976
16	0.649	0.948	1.000
16.5	0.833	0.996	0.999
17	1.000	0.974	0.992
17.5	1.000	1.000	1.000
18	1.000	1.000	1.000
18.5	1.000	1.000	0.996
19	1.000	1.000	1.000
21.5	1.000	1.000	0.896
Total	0.867	0.830	0.967

Table 4-10 Results of WTG01 grouped per Wind Speed Bin.

WD30	Average of Availability (A)	Average of Performance (B)	Average of Quality (C)
0	0.872	0.931	0.938
30	0.908	0.812	0.968
60	0.928	0.847	0.956
90	0.770	0.846	0.948
120	0.830	0.853	0.961
150	0.926	0.845	0.969
180	0.933	0.827	0.972
210	0.943	0.824	0.965
240	0.924	0.730	0.949
270	0.915	0.747	0.957
300	0.875	0.846	0.970
330	0.816	0.833	0.973
360	0.790	0.842	0.962
Total	0.867	0.830	0.967

Table 4-11 Results of WTG01 grouped per Wind Direction Bin.

4.2.5 Results of the Wind Farm

Table 4-12 summarizes the results considering the entire farm. The first aspect to notice is that on average all elements performed slightly better when the entire farm is assessed. The total OEE of the period assessed was 73%. The availability only improved 3%, however, the low value was achieved in the same period, which means some abnormality or event might have occurred in the entire farm.

Week	Average of Availability (A)	Average of Performance (B)	Average of Quality (C)	Average of OEE
26	0.705	0.714	0.986	0.496
27	0.685	0.773	0.970	0.516
28	0.696	0.760	0.977	0.522
29	0.664	0.772	0.980	0.510
30	0.677	0.813	0.981	0.547
31	0.732	0.812	0.981	0.587
32	0.793	0.785	0.974	0.608
33	0.846	0.781	0.971	0.641
34	0.804	0.785	0.980	0.617
35	0.907	0.769	0.968	0.674
48	1.000	0.923	0.980	0.906
49	0.974	0.887	0.988	0.855
50	0.977	0.875	0.979	0.838
51	0.990	0.807	0.988	0.789
52	0.973	0.879	0.972	0.834
53	1.000	0.781	0.991	0.774
1	0.980	0.858	0.972	0.819
2	0.962	0.863	0.985	0.815
3	0.890	0.872	0.976	0.753
4	0.984	0.862	0.969	0.822
5	0.987	0.841	0.978	0.813
6	0.998	0.814	0.969	0.788
7	0.952	0.873	0.967	0.801
8	0.951	0.830	0.963	0.756
9	0.930	0.818	0.967	0.733
10	0.990	0.852	0.967	0.816
11	0.960	0.880	0.960	0.813
12	1.000	0.818	0.976	0.797
Total	0.891	0.832	0.975	0.727

Table 4-12 Results considering all wind farm.

4.3 Alternative of Performance Analysis

The performance analysis discussed in the subsection 4.2.2 considered only external factors, as turbulence and some weather features. However, it is also possible to evaluate performance by considering internal factors such as the generator rotational speed or the pitch angle. In this section, some alternative methods to measure performance will be presented.

4.3.1 Wind Speed x Generator Speed

Similar procedure as shown in subsection 4.2.2 was adopted for this analysis. However, instead of the wind direction and the turbulence, the best performance in each wind speed was found considering the generator rotational speed. The generator speed was grouped in bins of 50 rpm. To facilitate comparison and visualization, the average results are presented on a weekly basis in Table 4-13.

Week	Average of Performance (B)	Week	Average of Performance (B)
26	0.757	1	0.898
27	0.832	2	0.925
28	0.881	3	0.903
29	0.899	4	0.919
30	0.902	5	0.926
31	0.836	6	0.900
32	0.816	5	0.926
33	0.791	6	0.900
34	0.760	7	0.905
35	0.810	8	0.904
48	0.921	9	0.878
49	0.927	10	0.890
50	0.935	11	0.906
51	0.870	12	0.898
52	0.905		
53	0.887	Total	0.888

Table 4-13 Summary of Performance Analysis - Wind Speed x Generator Speed

The result of this method was 0.888, 0.058 higher than the previous method, which considers only weather features. By adding the generator speed the performance was more stable, however, the value is still lower than the average performance found in the literature. As mentioned before the techniques used to measure performance can vary a lot, which makes comparisons less precise. Unfortunately, there is not enough information to interpret these results properly, but machine learning techniques have provided a couple of solutions to explain black box models that can give some tips on what might be happening.

There are several methods for interpretability of machine learning models, including feature importance, partial dependence plots, permutation importance,

LIME (Local Interpretable Model-agnostic Explanations), SHAP (SHapley Additive exPlanations), and integrated gradients. After careful evaluation, SHAP values were chosen as the preferred method for interpretability in this model.

SHAP is a comprehensive and unified method that provides detailed explanations of feature impacts on individual predictions. It utilizes Shapley Values, a game theory concept, to distribute predictions fairly among features. In this study, the open source SHAP library [234], available for Python users, was employed. Given the high accuracy achieved by RFR in the performance analysis of wind turbines, presented in section 4.1, the 'tree explainer' method recommended in the SHAP documentation was selected for this analysis. The assessment metric for the model was R².

The R² achieved in this activity was 0.85, indicating a strong correlation. The bee swarm graph in Figure 4-8 provides a summary of the SHAP analysis. Each point represents an instance, positioned vertically based on feature importance. The swarm's centre represents the central tendency, indicating the impact on the prediction (positive or negative). The colour signifies the feature's value, indicating its correlation with the final accuracy. The width of the swarm represents the uncertainty or spread associated with the SHAP values. From the analysis of the Figure 4-8, it is evident that Stator Active Power had the highest contribution to the performance, followed by Rotor Speed, Generator Speed, Ambient Temperature, and Pitch Angle.

Given that this is a DFIG (Double-Fed Induction Generator), the significant impact of Stator Active Power on performance can be attributed to its direct connection to the grid, aiding in reducing disturbances. As previously discussed, ambient temperature plays a crucial role in performance, particularly when temperatures exceed 35°C. Some studies have proposed solutions to increase productivity in such scenarios without compromising equipment safety, as mentioned in [133]. In this study, modifications to existing dust filters and enhancements in nacelle ventilation allowed for an increase in the start protection temperature from 35 to 40°C. The operational temperature of several components had a marginal impact

76

on the performance, but should be investigated, since this could indicate some abnormality in the equipment.



Figure 4-8 SHAP results of the performance.

4.3.2 Power Coefficient (PC)

As demonstrated in Equation 4-5, the power coefficient (PC) is directly related to the power output, and it is function of the tip speed ratio (TSR) and the pitch angle (β). Both values can be retrieved from the data provided. The tip speed ratio is the rotor speed divided by the wind speed, so this could be calculated following by taking the rotor speed measured by SCADA and converting the speed, as follows:

$$TSR = \frac{Rotor Tip Speed}{Wind Speed} = \frac{Rotor Speed \ x \pi D}{60.V_{10min}}$$
(4-11)

D is the rotor diameter.

To assess the best performance, the tip speed bin and pitch angle bin was rounded as integers numbers. Table 4-14 presents the results of this analysis,

where the average value was 75.4%. By considering the pitch angle and the rotor speed it was expected a better efficiency of the equipment, but the result was lower than the previous methods selected.

Week	Average of B	Week	Average of B	
26	0.669	1	0.751	
27	0.776	2	0.759	
28	0.763	3	0.759	
29	0.821	4	0.749	
30	0.836	5	0.743	
31	0.801	6	0.729	
32	0.754	5	0.768	
33	0.719	6	0.736	
34	0.715	7	0.708	
35	0.725	8	0.736	
48	0.823	9	0.782	
49	0.767	10	0.740	
50	0.752	11	0.751	
51	0.696	12	0.759	
52	0.767			
53	0.683	Total	0.754	

Table 4-14 Summary of Performance Analysis – CP.

4.3.3 Discussion

Regardless of the method selected, the most important thing is to not narrow too much the criteria for assessing performance. For the author, considering only external factors would be the best way to identify the equipment and operational losses, as demonstrated in subsection 4.2.2. Since, conceptually, if the wind conditions and weather features are the same, the equipment should provide same power output in normal situations. The problem behind narrowing the criteria too much is that this would mislead operators and managers about possibilities to improve the equipment productivity. In chapters 2 and 3, some simple solutions to improve the performance of the equipment, found in the literature, were briefly presented. However, according to OEE concept, this means that the equipment was not extracting power output at its maximum capacity.

4.4 Chapter Summary

This chapter was divided in two main parts. The first, section 4.1, a model was proposed to monitor and predict wind power output through traditional machine learning techniques. MET MAST Data was used as input and provided a high accuracy, even in daily average forecasts. The second part, sections 4.2 and 4.3, presented a data analysis of the same farm considering only SCADA data. The assumptions to calculate availability was discussed in subsection 4.2.1 and a method to estimate wind energy rejection was proposed in subsection 4.2.3. How to monitor and measure the performance was first presented in subsection 4.2.2, where only wind features was considered. Alternatives methods using internal information, such as generator speed and/or rotor speed, was presented and discussed in section 4.3. Finally, OEE was estimated, and its rate was grouped considering different categories for a better understanding of productivity and losses behaviour. This was presented in subsection 4.2.4 and 4.2.5.

5 LOSSES TRENDS & COST IMPACTS⁴

It was found in the literature three main reasons for projects to not achieve financial return along its entire operational life. They are the increase in OPEX more than expected in medium and long term, the decline of wind turbine performance due to the ageing of components, and finally, the curtailment and its tendency to be higher in the future. Therefore, the first section will introduce each of these problems with more details. Later, in section 5.2, an economic analysis is performed, considering the losses discussed previously. Finally, section 5.3 will present a cost analysis of a hypothetical onshore farm. Besides the loss trends discussion, this simulation will also check and compare the impact of different wind turbine classes in same location and layout.

5.1 Losses Trends

5.1.1 Failure Rate

As demonstrated in the Figure 3.2, the life cycle of an equipment can be represented by the bathtub curve, which represents the three main phases of an equipment life. First, the equipment starts with a high failure rate, also known as infant mortality, then, during the operational life, its failure rate stabilizes, and finally, failure rate starts increasing again, until running the equipment is not considerable viable anymore. Nonetheless, for some authors, the parameter shaper of the operational life period of wind turbines can be slightly higher than one [15], which indicates a slow constant increase in failure rates.

Some studies have shown this behaviour. According to [13], maintenance is far from being uniform across projects. As demonstrated by [235], annual failure rates of wind turbines rated above 1MW tends to increase significantly with time. In worser scenarios, the average annual failure rate starts close to 0.25, but it overpasses 2 failures per year in seven years. [236] and [237] have shown similar trends and [238] shown an increase in downtime during the operational life of wind turbines. Since the increase of failure rates directly affects the OPEX and

⁴ This chapter was based on the publication by Sathler, Yeter and Kolios [289].

the availability of an equipment, investigating these aspects can help to confirm if this is a consistent trend.

According to [239], the leading cause of the increase of costs are the more frequent failures with age. [14] found that operational costs of onshore farms can quadruple during its lifespan, similar scenario was pointed by [15]. Although offshore wind projects are newer and relies on more robust technology and engineering, the behaviour is comparable. In 2012, [12] affirmed the costs are likely to be 3.6 times higher after 20 years. [239] estimates the costs are likely to double in 10 years, without considering transmission costs. Regarding the availability index, [28] estimates a decay in the availability close to 95% after the 11th operational year for onshore projects. [240] investigated several databases, and found old projects, with 30 years, can have availability close to 80%.

It is worth mentioning that some of database investigated in the literature are old and can bring some bias in the analyses. However, although the technology has improved during the last years, it is difficult to confirm if this trend has been minimized, since most of companies, do not provide this information in detail. According to [13] which has followed the costs of some farms, the overall costs have indeed reduced, however, they still seem to increase with the age. In their study, newer or older projects have almost doubled its costs in five years, though the sample size of their investigation was limited.

5.1.2 Ageing

Any engineering system is at the risk of degradation during its lifetime. For system such as wind turbine this scenario can be worse, due to uncontrolled environment in which they are located. Besides the wear out of components, problems such as corrosion or blades erosion are examples of reasons why the performance of a wind system can deteriorate with the age and replace them can be costly and time consuming. According to [191], the increase in failures and downtime cannot explain the decreasing trend in capacity factor and expects that at least two thirds of the performance reduction is due to worsened efficiency, particularly resulting from the loss of aerodynamic efficiency caused by worn-out blades. In other

words, an older turbine tends to produce less than expected in the same wind regime, when compared to its early operational years.

In a study conducted by [16], for oldest turbines the yearly relative loss was on average 0.53% during the first 10 years, then this rate increased to 1.23%. The study concluded that this was due to the end of subsidies from the government, hence, companies tend to compensate for costs by reducing maintenance activities to keep the profit margin at the same level. For newer projects, they had the data from only the first 10 years, and the relative degradation level was 0.17% per year. These results illustrate the importance of a good maintenance routine and how newer turbines are more prepared to resist degradation.

Other studies have also investigated the theme. [115] assessed the performance of one farm through SCADA for one year and found no significant loss, but they found an impact on the temperature and vibration of the system. [17] investigated performance losses in wind farms in Germany from 2000 to 2014 and found that yearly ageing losses are, on average, 0.63%. [191] calculated an early decrease rate of 0.15%, which would account a total of 6% reduction in performance until the end of the turbine's life. A worser rate was identified by [18], where farms from UK were investigated from 2002 to 2012, and the average relative ageing loss found was 1.6%. Although this study was well conducted, it is more likely that this rate is an outlier, when compared to other studies investigated.

5.1.3 Curtailment Trend

Curtailment is when the production needs to be interrupted or reduced, regardless of the wind regime and the availability of the turbine. This could be by operational problems, when production is out of requirements and could put the system in risk, or by external factors, such as bird migrations, shadow flickers, or lack of demand. The last one is particularly a big problem, since they are unexpected and in case of the project assumes these losses, this could affect the viability and the final price of the energy. As discussed in section 3.3, while some countries experience curtailment rates below 5%, others have reported energy dispatch curtailment of more than 15%. This issue has raised concern among several countries.

According to [149], who investigated curtailment in Nordic countries, although there are no big concerns nowadays, this can become a problem in the future if new transmission lines are not constructed. [19] had similar conclusion, where in their simulation 15.5% of the wind energy produced in USA in 2050 could be curtailed without expansion of transmission lines. [241] simulated various scenarios to assess the impact of increasing renewable energy penetration, and their findings showed that curtailment loss in Great Britain could reach 17%. However, they also examined how power grid limitations could mitigate these losses.

Given the global push to increase renewable energy sources rapidly, there is a genuine risk of increased curtailment levels. As noted in [220], the installation time for wind farms is usually shorter than that of transmission networks. Although there are some solutions (discussed in 6.3) to reduce curtailment and increase wind power penetration, it might not be feasible to use or store all excess of energy generated [212]. With higher levels of wind energy penetration, the cost of grid strengthening may increase significantly.

5.2 Cost Impacts

5.2.1 Methods

There are several methods to assess a project viability, profitability, and its total costs. In this section, the three methods selected to develop this work will be presented. LCOE, to check the average cost expected to produce one unit of energy and NPV, to assess its viability. Additionally, a Monte Carlo Simulation and Sensitivity Analysis are included for a better analysis of results.

5.2.1.1 LCOE

The levelized cost of energy (LCOE) is one of the most commonly used metrics for estimating the cost of energy production, allowing for comparison of different energy sources or projects within the same energy source. In brief, the LCOE takes into account the present value of all costs and expenses over the lifespan of the project, from its conception to decommissioning, and divides it by the discounted sum of all energy delivered over the same period. The main formula and its adaptation to the wind power model being developed are shown below:

$$LCOE = \frac{Life\ Cycle\ Cost}{Total\ Energy\ Delivered} = \frac{\frac{CAPEX + DECEX + \sum_{i=1}^{t} OPEX}{(1+r)^{t}}}{\frac{\sum_{i=1}^{t} Anual\ Energy\ Output}{(1+r)^{t}}}$$
(5-1)

Where r is the discounted rate and t corresponds to the lifespan of the project.

5.2.1.2 NPV

The net present value (NPV) provides insight to confirm the viability, profitability, and value added by the project. Unlike LCOE, this metric considers the revenue value, comparing the cash flow expected throughout its lifespan. The projects that are worthwhile to invest in, need to have a positive value at the end of the project's lifespan, while the project with a negative value should be rejected. Figure 5.1 illustrates a diagram with the cash flow analysis. It is essential to check that even if the project has some negative years, as illustrated in year 4, the overall analysis of the project could still be positive in the end. When this method is used to compare different projects, the one with the highest value, where the expected return is higher, should be selected. The NPV formula is as follows:

$$NPV = \sum_{i=1}^{t} \frac{Yearly \ Cash \ Flow}{(1+r)^t} - Investment \ Capital$$

$$= \sum_{i=1}^{t} \frac{\{(Year \ Production \ x \ Revenue) - OPEX\}_i^t}{(1+r)^t} - CAPEX$$
(5-2)

Where r is the discounted rate and t corresponds to the lifespan of the project.



Figure 5-1 Diagram with cash flow analysis

5.2.1.3 Monte Carlo Simulation

It is very difficult, if not impossible, to develop a fully accurate model for any longterm project. During the design phase, some assumptions need to be made, which affect the results, in turn misleading investors' decisions. In order to address this issue, several methods are available to create a more reliable financial model, one of which is the Monte Carlo Simulation (MCS). In a Monte Carlo Simulation, the range of uncertain inputs is randomly selected, and the model is run a significant number of times, producing the range of all possible outcomes. Therefore, MCS helps investors to understand the variability of financial outcomes, providing a better insight into the risks associated with the viability of an offshore wind project.

5.2.2 Cost Assessment

This simulation will focus on offshore projects. Usually, the distribution of costs of an offshore deployment are on average 20 to 30% of costs from OPEX, 70 to 80% to CAPEX, and 1 to 3% from the DECEX. In the next subsections, the values found on literature related to these elements will be presented. Additionally, some other essential elements will be discussed, the strike price, capacity factor, and operational losses to determine the production and incomes of the project; and, finally, the discounted rate, a key component used in long-term economic analysis.

5.2.2.1 CAPEX

Going further away from the shore requires more complex engineering solutions, consequently, the cost of energy produced is expected to be higher and thus potentially unviable. However, even though offshore wind is more expensive than its onshore counterpart, operating in better wind regimes makes it a competitive energy source when considering cost per unit. According to [242], offshore projects can be cheaper than onshore wind or even compete with some gas plants in few years.

According to [243], the average CAPEX in 2019, considering projects delivered in the UK, Netherlands, France, and Norway was around 3.7m£/MW. Although more expensive, this value is not far from the models developed by [10], [244], [198], achieving respectively, 3.1m£/MW, 3.3m£/MW, and 2.8m£/MW. Two other important renewable energy agencies, NREL [245] and IRENA [246], registered respectively, 2.7m£/MW and 2.2m£/MW, while the UK government predicts that this cost could be around 1.6m£/MW in 2025 [247].

5.2.2.2 OPEX

The share of OPEX in the total costs of offshore projects, can be as high as five times, when compared to onshore [248]. Besides the harsher environment, which can cause more fatigue loads combined with degradation mechanisms, the bottleneck regarding supply vessels and the dependency on weather conditions for maintenance activities are the main reasons why operations can be quite costly. Storm conditions could make the journey to the turbine riskier. According to [52], in offshore only 22.2% of the mean time to repair are related to repair activity itself, and the rest is due to preparations and the waiting for appropriate conditions and spare parts. Moreover, according to [38], components might be changed before its ideal life cycle because of limited interventional opportunities, which could also affect the cost.

The values regarding OPEX found in literature vary significantly. Most agency reports have shown a lower cost, from 50 to 60k£/MW per annum, while in economic simulations, prices were shown to be higher than 100k£/MW. This could be explained by the fact that there are ambiguous assumptions about the distance. Furthermore, the cost associated with offshore wind would vary for each country due to salaries, taxes, bureaucracy, and infrastructure affecting all operational activities. Another problem is that most companies do not provide the actual costs, considering it confidential and strategical.

5.2.2.3 DECEX

Wind farm projects are designed with a specific operating period in mind, and after this period, it is expected that the investor will decommission the project and clean the area, leaving the environment as close to its original state as possible. Although projects can be extended, the DECEX needs to be taken into consideration when the project is no longer financially viable. [249] conducted a bottom-up analysis and found that the DECEX would fluctuate from 175 k£/MW to 480k£/MW. This range is close to what was found in another research, such as cost of 462k£/MW [10], 244k£/MW [244], and 135k£/MW [198]. As presented in Figure 1-1 it is likely that part of these costs will be covered by the recycling of residuals, but this was not considered in this analysis.

5.2.2.4 CF

The expected wind speed varies considerably throughout the year, and due to other inefficiencies in offshore wind operations, the actual energy output can differ from the theoretical maximum energy output from offshore wind assets. The difference between the actual energy output and the theoretical maximum output is represented by a ratio called the "Capacity Factor (CF)", which provides a more realistic estimate of expected energy output from an offshore wind site and is used in energy resource and feasibility assessments. For accurate capacity factor estimations, it is crucial to know the wind regime, site characteristics, and technical information from wind turbines.

There is an optimistic expectation that technological advances, larger wind turbines, and offshore wind sites with higher wind potential will allow for much

higher capacity factors. In this regard, some agency reports forecast capacity factors surpassing 60% in the next few decades [247]. However, the present study only considers the actual capacity factors available in the literature, rather than what is expected in the future. Thus, the capacity factors used in the analysis vary from 42.5% to 53.5%.

5.2.2.5 Strike Price

The revenue estimate is a vital part of the economic viability analysis. This value corresponds to the strike price of the energy delivered and depends on the contract details of each project. Most wind farm projects have an agreement about their prices and do not compete directly in the market, which has been an effective way to incentivize offshore wind development. Nonetheless, the strike prices have been reducing drastically, as shown by The Low Carbon Contracts Company (LCCC) and the Electricity Settlements Company (ESC) projects. At this moment, they manage 40 offshore projects through contract for difference (CfD), where the current strike prices to offshore projects fluctuate from £45.73 to £176.57 [250].

A similar difference is described by [242] where they state that between 2017 and 2020, the average value from offshore dropped from £167 to £112 per MWh, and the average future values expected is £59.25/MWh. Other researchers estimated to their simulation £140/MWh [10], £74/MWh [198], and £87.60/MWh [251].

5.2.2.6 Discounted Rate

To perform a long-term economic analysis, the time value of money must be considered. As money can depreciate over time and future expenses can have a lower value in the present, investments made in different periods must be corrected to eliminate the effects of time. According to [252], the discount rate varies from 5 to 7% in most developed countries, while this value can be higher than 10% in other emerging countries. [239] reports that the discount rate in subsidized wind projects fell from 8% in 2009 to 4% in 2019. Other studies, such as [253], [247], and [10] which estimated 8.06%, 6.3%, and 6.15% respectively.

5.2.2.7 Summary

In light of the discussion above, estimating the costs and production of offshore installations can be challenging. The distance to the shore, structural basis, logistics, policies in each country, and the expertise and size of the company are the main reasons why costs fluctuate significantly.

Table 5-1 summarizes the values found in the literature, which will be the cost basis for this study. Regarding the assumption outside Table 5-1, the mean value shown in Figure 3.4 will be considered, So, the availability, performance, and quality rates used will be 92%, 90%, and 95%, respectively. The discounted rate assumed is 6%, according to values discussed in the subsection 4.2.2.6. The lifespan assumed is 20 years, since this is the technical life most wind turbines are certified for [254] [255].

<u>OPEX</u>	DECEX	<u>CF</u>	Strike Price	<u>Reference</u>
£112,300.00	£243,770.00	0.425*	£140	[10]
£66,122.24		0.44		[246]
				[256]
£180,218.75	£461,655.81	0.535		[244]
£53,250.00	£135,000.00	0.49	£74	[198]
	£327,500.00			[249]
£100,018.86				[257]
£54,750.00			£87.60	[251]
				[243]
£54,170.40*		0.51		[247]
£146,200.00				[258]
			£167 - £47	[242]
			£155 - £59	[250]
	<u>OPEX</u> £112,300.00 £66,122.24 £180,218.75 £53,250.00 £100,018.86 £54,750.00 £54,170.40* £146,200.00	OPEX DECEX £112,300.00 £243,770.00 £66,122.24 £180,218.75 £180,218.75 £461,655.81 £53,250.00 £135,000.00 £100,018.86 £54,750.00 £54,170.40* £146,200.00	OPEX DECEX CE £112,300.00 £243,770.00 0.425* £66,122.24 0.44 £180,218.75 £461,655.81 0.535 £53,250.00 £135,000.00 0.49 £100,018.86 £54,750.00 0.51 £54,170.40* 0.51	$\begin{array}{c c c c c c c c c c c c c c c c c c c $

Table 5-1 Costs from the literature review

Conversion Rate (2021) £ = 0.73US\$ £ = 0.86€

*Stipulated by the authors

5.2.3 Scenarios

As discussed in the section 5.1, there are three main risks to be considered during operational life. The rates for each scenario will be discussed in following subsections. After that, the economic model proposed will be briefly explained and a summary of the scenarios, parameters and costs considered in the simulations will be presented.

5.2.3.1 Scenario A - Failure Rate Increasing

OPEX includes all operational costs, from administrative to maintenance. As the increase in the costs are related to the increase of failures, only maintenance costs will be considered to rise in this simulation. The other costs are going to be considered as constant throughout the lifespan, as inflation are not incorporated in this work. In the model developed by [10], around 50% of the OPEX are due to maintenance. While to [251], [244], [259], the share is 58%, 45%, and 43% respectively. Considering the average of these values, this work will consider that 49% of OPEX costs are due to the maintenance.

Following the trends found and discussed in the section 4.1.1, the increase in the maintenance costs will be 3 times during its entire life, which means an increment of 6% per year. It is important to notice that more failures result in more production losses. In this study, it will be considered that half of availability losses, or downtime, are planned (preventive maintenance); while, the other half are considered unplanned, so failures and breakdowns. Thus, for each increase in cost, a further deduction in the production following the same rate will be added. Since it is difficult to determine when these failures will occur, these losses are multiplied by its CF, as an attempt to be closer to real life scenario.

5.2.3.2 Scenario B – Ageing

According to subsection 5.1.2, the yearly relative ageing rate fluctuates from 0.17 to 1.6%. Great part of the studies assessed focused on onshore deployments and in some cases older deployments. To convert this to offshore to aspects needs to be considered. First, offshore wind turbine can be considered a recent technology, especially the ones that go further to the shore. In that case, assuming a lower ageing rate could fit better to offshore farms. However, offshore are exposed to a harsher environment and face more loads in its operational periods. So, it is reasonable to consider the range presented, therefore a constant relative ageing rate of 0.5% will be considered.
5.2.3.3 Scenario C – Curtailment and Quality Losses Trends

It is difficult to determine how much energy will be curtailed in the future. The energy can be curtailed by several reasons, including lack of demand and power quality. As described in the section 2.2.4, this work will consider all losses after the turbine as a quality loss. It is expected that the losses in cable and the power quality reduces together with the degradation of the turbine. Therefore, in this scenario, the initial quality loss of 5% will triple during the entire life. Considering all risks involved and the curtailment rates discussed in the subsection 5.1.3, this assumption is not only consistent, but also a conservative, since this loss can occur by several reasons.

5.2.4 Economic Model Application

The proposed model is divided into two parts. The first part presents a projection of the real-time behaviour and extracts key financial metrics such as LCOE, NPV, critical year, and breakeven year. These metrics are calculated based on average values. The critical year helps determine when the costs will exceed the benefits, while the breakeven year represents the point at which the accumulated benefits surpass the total costs. The concept of the breakeven year is similar to that of the discounted payback year, which takes into account the time value of money.

It is important to note that the assessment of the critical year considers the fixed instalment payments of the CAPEX. Therefore, it is possible for a project to have a critical year before its end of life, while the investment breakeven can still occur during the operational life, as the analysis focuses on cumulative cash flows without considering investment obligations. This distinction will become clearer in the subsequent sections when discussing the results of the simulations. Figure 5-2 provides a visual representation of how the critical and breakeven years are determined, aiding in the understanding of these key concepts.



Figure 5-2 Diagram of Breakeven and Critical year analysis – Part I

The second part of the model proposed is the Monte Carlo Simulation (MCS). In addition to the scenarios presented before (Scenarios A, B, and C), a reference scenario is created in which no loss trends occur. This is more common to find in the literature, where costs and benefits are considered constant throughout the project life. Additionally, since it is more likely that the scenarios defined will occur at some level together, a mix of scenarios is investigated. To do this, all possible combinations are explored: Scenarios A+B, A+C, B+C, and A+B+C.

Figure 5-3 summarizes the economic model proposed and illustrates the MCS. For each simulated scenario, 100,000 iterations were randomly run, selecting the defined cost range, one positive and one negative standard deviation, and dividing them into 25 possible values with the same interval. There is no fixed rule or concept for determining the exact range size in Monte Carlo Analysis. However, a larger number of values generally enables a more comprehensive exploration of the variable space, creating a greater range of feasible scenarios. It is important to balance this with computational resources and time constraints. In this case, considering the standard deviation and average values, a range of 25 values was chosen, which was deemed sufficient to capture variability and provide robustness to the model.



Figure 5-3. Flowchart of Economic Model and Monte Carlo Simulation – Part II

Table 5-2 presents the costs and all considered assumptions to run this model. The values correspond to the mean value, μ , found in the literature. The standard deviation, σ , was calculated assuming the values have a normal distribution. The assumptions regarding initial losses and performances, the operational period, and the discount rate are the same in all simulations. The parameters used in each scenario are presented in Table 5.3.

	μ	σ
CAPEX	£2,793,945.03	£608,267.09
OPEX	£95,878.78	£44,813.62
DECEX	£291,983.95	£119,396.26
Revenue	£111.40	£37.43
CF	0.49	0.04
Availability	0.92	-
Performance	0.90	-
Energy Losses	0.05	-
Operational Period	20 years	-
Discount Rate	6%	-

Table 5-2 Average Costs and Basic Assumptions to the model

Table 5-3 Scenarios investigated

Scenario	Parameters
Reference	-
Scenario A	OPEX Increase rate = 6% per year.
Coonano / (49% of OPEX is due to maintenance, or variable.
Scenario B	Year Ageing Loss = 0.5%.
Scenario C	Curtailment/Quality losses Increase rate = 6% per year.

5.2.5 Results and Discussion

The scenarios have been sorted from the most positive to the most negative. To avoid repetition, only scenario A is presented graphically. The results from the economic projection of all scenarios are presented in Table 5-4, which corresponds to Part I of the model discussed in subsection 5.2.4.

Table 5-4. Projected cost throughout lifespan considering average values.

	LCOE (£/MWh)	NPV (£/MW)	NPV (£/MW) Critical Year	
Reference	£98.93	£501,609.17	-	15
Scenario C	£102.41	£349,487.39	-	15
Scenario B	£103.64	£298,076.55	20	16
Scenario B + C	£107.16	£157,326.81	12	17
Scenario A	£110.89	£19,773.91	10	18
Scenario A + C	£114.69	-£124,287.95	9	-
Scenario A + B	£116.34	-£183,758.72	8	-
Scenario A + B + C	£120.19	-£316,448.53	7	-

In Figure 5-4, a real-time and accumulated projection for scenario A is presented. The first aspect to notice is that, although the NPV of this Scenario was positive, the critical year was in the tenth year - half of the lifespan expected. This may be considered counterintuitive; however, this result is supported by some studies. According to [214], wind turbines require increased maintenance costs and significant refurbishment around the 11th year of operation. The breakeven year was in the 18th year, which is closer to the end of turbine's operational life. These results can be explained by the small margin projected, where the average LCOE (£110.89/MWh, 12% higher than the reference) was only £0.51 lower than the strike price assumed.



Figure 5-4 Results Scenario A

The impact on costs and viability is found to be lower in the other scenarios when assessed individually. While the increase in LCOE for Scenario C is around 3.5%, it is around 4.7% in Scenario B. This slight difference of 1.2% in Scenario B is enough to raise the breakeven year to the 16th year and bring the critical year to the last year of operation. Nonetheless, in both scenarios, NPV was higher than zero, meaning that the techno-economic analysis suggests it is worthwhile to invest in the project. In contrast, in the mixed scenarios, the results are not very positive. Apart from Scenario B+C, where the results are better than Scenario A alone, Scenarios A+C, A+B, and A+B+C appear to have negative net present values and do not achieve the investment breakeven year during the projected lifespan.

As discussed before, in addition to the impact on operational loss trends, Part II of the analysis aims to understand how assumed costs could affect the LCOE and the project's viability. Therefore, Table 5-5 summarizes the results from MCS,

where the variation in LCOE is presented, considering a 95% confidence interval, together with the percentage of positive NPV values obtained in the simulation. Figure 5-5 presents the histogram with both analyses of Scenario A. The costs fluctuate from £81.64 to £140.65, and more than 50% of simulations considering Scenario A resulted in an NPV higher than zero. This means that despite the increased OPEX costs, there is still a considerable chance of the project being profitable for its investors.

	Lower Limit	Average	Upper Limit	NPV > 0 (%)	
Reference	£73.81	£99.08	£124.34	0.65986	
Scenario C	£76.44	£102.62	£128.81	0.61316	
Scenario B	£77.33	£103.79	£130.26	0.59954	
Scenario B + C	£80.02	£107.43	£134.84	0.55164	
Scenario A	£81.64	£111.14	£140.65	0.50332	
Scenario A + C	£84.46	£115.04	£145.61	0.45415	
Scenario A + B	£85.52	£116.63	£147.74	0.43399	
Scenario A + B + C	£88.36	£120.49	£152.62	0.38728	

Table 5-5. Results from Monte Carlo Simulations in all scenarios.



Figure 5-5 Results of Monte Carlo Simulation of Scenario A

Scenarios B, C, and B+C are also very positive for investors, achieving NPV higher than zero in more than 50% of the simulations. On the other hand, Scenarios A+B, A+C, and A+B+C are considered riskier to investors. Besides reassuring an elevated increase in costs, found in Part I, less than half of simulations are considered worthy of investing. The worst case is Scenario A+B+C, where the average LCOE is £120.49/MWh, 21.6% higher than the reference, and only 38.7% of simulations has got a positive NPV.

It is worth mentioning that these scenarios were investigated to understand how the operational loss trends would affect the projects financially. Nevertheless, operators and managers are expected to interfere in these projects before the situation becomes irreversible. Therefore, the proposed study not only provides an estimate of costs and risks but also serves as a guide for managers. Such guidance helps decision-makers monitor whether the project is going in the right direction and what the consequence would be if nothing is done to mitigate the risk.

Furthermore, conducting a sensitivity analysis is crucial for understanding the relationship between the defined rates and the output. This analysis helps identify which factors are more critical and sensitive to the overall outcomes. In this study, in addition to the three previously discussed scenarios, the discount rate was also investigated, as it is a key element in cost models. The range of values considered encompassed both more likely range and extreme scenarios, with the previously determined average value serving as the central value. The chosen step size for the investigation was 0.02, except for Scenario B, where a smaller step of 0.002 was used. This step size adequately covers the relevant assumptions, and further narrowing of the step is not necessary. Figure 5-6 presents the results for the levelized cost of energy (LCOE), while Figure 5-7 depicts the percentage of net present value (NPV) higher than zero. Both investigations utilized the same methodology as shown in Figure 3, but with 1000 iterations conducted instead.

97



Figure 5-6 Results of the sensitivity analysis on LCOE.



Figure 5-7 Results of the sensitivity analysis when NPV>0.

The first thing to notice is that the discount rate variation had the highest impact in the economic simulations, where the costs in the worst case, Scenario A+B+C, go from £97/MWh with a tax rate of 2% to £129/MWh with a tax rate of 10%. The result indicates that financial incentives are still important for the development of wind energy technology. OPEX rate variation was the second worst case. Although lower than the discount rate, it is evident by the graph that the impact due to the increase in OPEX has an exponential effect, so with higher rates is likely that OPEX becomes the most significant influence on LCOE costs. Variations of ageing and curtailment rates affected the costs less. In most cases, the percentage of positive NPV was higher than 50%, even considering worse rates. To sum up, the model proposed showed that the inclusion of the scenarios discussed in Section 5.2.3 can explain why some projects are failing economically in the medium and long term. Nonetheless, this study does not consider any intervention by operators to reduce these losses, which is very unlikely to be true. For considering operators' interventions, analyses which include refurbishment and new maintenance costs, should be developed to compare the financial advantages of such interventions. Nevertheless, the results obtained in the present work confirm what was found in the literature. The increase in operational losses might be the main reason for wind energy projects to have a shorter financial lifespan. In appendix A, a summary considering return tax of 4% and 8% is presented.

5.3 Case Study⁵

The present study case is a hypothetical onshore wind farm model built in Meteodyn WT, a commercial CFD-based software which makes use of the turbulence flow method RaNS (Reynolds averaged Navier Stokes), transport equations, and thermal stability to model wind flow [260][261]. It is out of scope from this work to discuss how CFD works. Some inputs cannot be disclosed due to confidentiality concerns, such as the modelled wind farm's location, corresponding wind regime, and wind turbines' model. The provided wind regime covers an entire year averaged by 10 minutes of reading. The motivation behind this simulation is to compare the performance and costs of different classes of turbines in the same conditions and the impact of the trend losses discussed in the previous sections. Therefore, the layout was not optimized; however, a minimal distance between rotors is considered, following manual requirements. Figure 5-8 presents the average wind speed analysis calculated in the software and the farm's layout. This farm is located in a relatively open area, so there is no significant interference from the surroundings. In this case, ambient wind speed variation is stable within the farm.

⁵ This section was based and contains extracts from the publication Sathler, Yeter and Kolios [290].



Figure 5-8 Mean Wind Speed analysis of the hypothetical farm in Meteodyn.

Figure 5-9 presents the power curve and the coefficient of thrust (CT) curve from each turbines assessed. which are 2 MW wind turbines. The classes tested were IA, IIA, IIIA, and IIB. Also, the hub height considered was the same for all turbines, 80 m. The CT is necessary for calculating the downwind speed reduction with higher precision. It is crucial to notice that the cut-out of the wind turbine IIB is 20 m/s, while the others are 25 m/s. Although in this case wind did not achieve higher value than 20m/s, by the distribution is very likely that this will eventually happen. For safety and durability of the turbine, the standard BS EN IEC 61400-1:2019 defines the basic parameters and criteria according to the environmental factors. There are three different wind classes, I, II, and III, and four different turbulence classes, A+, A, B, and C. Class S is defined by designer and do not follow the standard. Table 5-6 presents the criteria of each class according to the standard.



Figure 5-9 Power Curve and CT Curve from turbines assessed.

	Table 5-6 Basic Param	eters for wind turbines	[IEC 61400-1:2019]
--	-----------------------	-------------------------	--------------------

W	Wind turbine class		I	Ш	111	S	
Vave		(m/s)	10				
	(1	m/s)	50	42,5	37,5		
^ℓ ref	Tropical	(m/s) $V_{ref,T}$	57	57 57 57			
A+		$I_{ref}(\cdot)$		0,18			
А		$I_{ref}(-)$		the designer			
В		$I_{ref}(-)$					
С		$I_{ref}(\cdot)$	0,12				
The parameter values apply at hub height and							
Vave	is the ann	ual average v	wind speed;				
V _{ref}	is the refe	erence wind s	peed averag	je over 10 m	in;		
V _{ref,T}	V _{ref,T} is the reference wind speed average over 10 min applicable for areas subject to tropical cyclones;						
A+	A+ designates the category for very high turbulence characteristics;						
А	A designates the category for higher turbulence characteristics;						
B designates the category for medium turbulence characteristics;							
С	C designates the category for lower turbulence characteristics; and						
I _{ref}	is a refere	ence value of	the turbulen	ce intensity	(see 6.3.2.3).		

It is possible to calculate the operational safety limits expected from each turbine using these basic parameters. Three aspects were assessed in this work. The Rayleigh distribution of the wind speed distribution, P_r , turbulence intensity model, TI_{M} , and the extreme turbulence model, TI_{ETM} . These parameters are estimated as follows:

$$P_r(V_{hub}) = 1 - \exp\left[-\pi \left(\frac{V_{hub}}{2V_{ave}}\right)^2\right]$$
(5-3)

$$TI_{M} = \frac{I_{ref}(0.75V_{hub} + b)}{V_{hub}}; b = 5.6m/s$$
(5-4)

$$\sigma_{ETM} = cI_{ref} \left(0.072 \left(\frac{V_{ave}}{c} + 3 \right) \left(\frac{V_{hub}}{c} - 4 \right) + 10 \right); c = 2m/s$$
(5-5)

$$TI_{ETM} = \frac{\sigma_{ETM}}{V_{hub}}$$
(5-6)

 V_{hub} is the real average wind speed at the hub height.

Figure 5-10 presents the average wind distribution in each turbine class. Figures 5-11 to 5-14 present the average result of turbulence intensity analysis and extreme turbulence model of the turbines IA, IIA, IIA, and IIB, respectively. The blue curve in the figures refers to the limit stablish by IEC standard, as mentioned before, while the red one is the one measured through Meteodyn software.

It is vital to notice in figure 5-10 that none of the wind distributions fits the wind classes perfectly, however, from the rated wind speed any of the curves were above the limit. Class III was the curve with closest shape, although the average wind speed in the hub height is more frequent. Also, it is worth noting that considering the rated speed of around 10 m/s, any of the curves are above the limit, which means less loading on the blades. Not meeting the standard means that the turbine is not working in the conditions in which the extraction of power output is maximized according to the turbine class.



Figure 5-10 Wind Distribution of each class.



Figure 5-11 Turbulence Analysis of Wind Turbine IA.







Figure 5-13 Turbulence Analysis of Wind Turbine IIIA.



Figure 5-14 Turbulence Analysis of Wind Turbine IIB.

Regarding the turbulence intensity, the turbulence analyses are presented in Figures 5-11 to 5-14. The results suggested that only turbine IIB was out of standard limits. Since this behaviour is only in the Extreme Turbulence Model, reconfiguring the layout should reduce or solve this issue. The wind turbine manuals often note that if the turbulence intensity is high, the fatigue loads on the wind turbine are expected to increase; in turn, the expected turbine life decreases. Although the correct action would be to discard wind turbine IIB as an option, for educational purposes and comparison reasons, the complete analysis of the turbine IIB will be kept; however, a higher ageing and failure rate will be assumed, 0.7% and 8%, respectively.

Besides the estimation of the Annual Energy Production (AEP) and the wake losses, the software allows the creation of criteria for performance losses and curtailment losses through temporal analysis. In this context, two rules were created to curtailment. The first one is a power limit of 75% of the rated capacity during early mornings from 2:00 a.m. to 5:00 a.m. to simulate a lack of demand. During peak times (weekdays from 7:00 p.m. to 9:00 p.m.), 10% of the energy is curtailed, simulating grid operators' preferences during critical times and allowing some flexibility to deal with inertia. This consideration may eventually happen in real world, allowing operators to assist grid if more energy is needed. For the performance, two criteria were created. As mentioned in section 4.2.2, temperature higher than 35°C and turbulence intensity higher than 15% can cause performance restrictions, so a further 10% reduction is considered. In this paper, high ambient turbulence intensity is above 15%. Finally, quality loss was simulated similarly to what was discussed in 4.2.3, but the limit was 10% and the interval investigated was 1 hour. Table 5-7 presents a summary of the output from the software for each turbine class.

Turbine	IA	IIA	IIIA	IIB
Diameter (m)	80	88	97	100
Rated (kW)	2000	2000	2000	2000
AEP (/KW _{installed})	2551	2773	3462	3537
Wake Effect (%)	5.48%	5.98%	5.94%	5.89%
Ambient Performance (%)	2.08%	2.09%	2.11%	2.11%
Quality Losses - Ramp (%)	3.41%	4.16%	4.29%	4.50%
Curtailment (%)	1.44%	1.79%	2.14%	2.28%

Table 5-7 Summary of Turbines and Meteodyn Results.

For the economic analysis, the three main components of costs, CAPEX, OPEX, and DECEX, discussed in subsection 5.2.2, will be considered. For CAPEX, the costs will be divided into two main parts. The first part is the wind turbine, whose cost depends on its class, hub height, and rotor diameter. The wind turbine cost can be calculated through a tool called Wind-Plant Integrated System Design and Engineering Model (WISDEM) [262], tool developed by National Renewable Energy Laboratory (NREL). WISDEM correlates material mass and costs from several deployments, and through regression analysis generates easy-to-use equations to estimate wind turbine costs.

The second part is the Balance of Station Cost (BOS), which includes civil works, assembly, installation, engineering services, permits, electrical, and connections costs. The NREL repository is also used to calculate BOS following the design and scaling model developed by [263]. OPEX can be considered into three parts: Land lease, O&M, and replacement. These costs (in USD) can be calculated in USD using the expressions given in the following [263]:

$$0\&M \text{ costs in USD} = 0.007 \text{xAEP}$$
 (5-7)

Land Lease Cost in
$$USD = 0.00108xAEP$$
 (5-8)

Replacement Cost in
$$USD = 10.7xMachine Rating$$
 (5-9)

DECEX, as discussed before, is still hard to estimate effectively due to a lack of experience [264]. DECEX depends on the age of the project, the availability of

crane offers (second-hand market), and the price volatility of recyclable materials [265]. Although 5% of CAPEX was deemed as a reasonable estimate in [264], a recent study [266] estimated higher costs, ranging from 200,000 to 532,000 USD per turbine. The present work assumes that 10% of CAPEX covers the DECEX cost assumption, making it assumptions slightly less than 2% of the total costs, which is reasonable according to the literature. Additionally, the initial availability considered was 98% to all turbines, as demonstrated in section 3.4, while the economics elements assumptions, such as the discount rate and the operational life, are taken as 6% and 20 years, respectively (following subsection 5.2.2.7).

Table 5-8 summarizes the cost analysis of each element of costs considering all the turbines investigated. As expected, turbines with larger rotors result in higher costs. In this study, the O&M is considered as "variable OPEX", while the Cost associated with land and replacements are considered to be "fixed OPEX". Finally, the costs were converted to pounds following the rate of $\pounds 0.73 = USD1.00$.

Turbine	Turbine IA IIA		IIIA	IIB
Foundation	£40,512.53	£43,753.20	£47,331.91	£48,510.37
Transportation	£62,692.40	£62,692.40	£62,692.40	£62,692.40
Civil Work	£71,861.20	£71,861.20	£71,861.20	£71,861.20
Assembly	£42,035.70	£47,010.70	£52,702.00	£54,620.02
Electrical	£116,011.60	£116,011.60	£116,011.60	£116,011.60
Permits	£32,555.08	£32,555.08	£32,555.08	£32,555.08
BOS - Total	£365,668.51	£373,884.18	£383,154.20	£386,250.66
Turbine/MW	£587,940.28	£618,864.86	£685,054.98	£708,994.32
CAPEX	£953,608.79	£992,749.04	£1,068,209.18	£1,095,244.98
DECEX	£95,360.88	£99,274.90	£106,820.92	£109,524.50
OPEX	£22,389.74	£23,345.56	£26,311.90	£26,632.44
0&M	£13,037.39	£14,171.72	£17,692.07	£18,072.47
Land Lease	£2,011.48	£2,186.49	£2,729.63	£2,788.32
Replacement Cost	£15,622.00	£15,622.00	£15,622.00	£15,622.00

Table 5-8 Summary of Cost Analysis of each Turbine Class.

Conversion Rate (2021) £ = 0.73US\$

Table 5-9 summarizes the result of the economic analysis. the LCOE analysis of each turbine and each scenario. The turbine IIIA is found to be the best option in terms of cost-benefit, while the scenario "IA" is rated as the worst. This is explained by the fact that the smaller diameter wind turbines extract less power from the wind, which is not compensated by appropriated local wind regime.

Turbine IIB is estimated to be the second cheapest average LCOE despite higher loss trend rates. However, this can be considered riskier to investors since the uncertainty related to LCOE is the highest. The reference scenario shows that IIIA would still be the best turbine for this site, regardless of the slightly smaller diameter. The reason behind this is the higher losses estimated through Meteodyn (Table 5-7), which was higher for curtailment, performance, and quality. Another negative aspect to the wind turbine IIB, it is that its cut-out speed of 20m/s, which could limit the production during good wind conditions.

Scenarios/Turbine	IA	IIA	IIIA	IIB	
Reference	£52.37	£51.05	£44.77	£44.92	
С	£54.15	£53.22	£46.84	£47.13	
В	£54.71	£53.34	£46.78	£47.83	
Α	£57.15	£55.89	£49.59	£52.27	
B+C	£56.51 £55.54		£48.88	£50.08	
A+C	£58.08	£58.26	£51.88	£53.54	
A+B	£59.71	£58.42	£51.83	£55.69	
A+B+C	£61.67	£61.67 £60.81 £54.15		£58.28	
Avg. LCOE (£/MWh)	£56.79	£55.82	£49.34	£51.22	
Std. LCOE (£/MWh)	£2.85	£3.03	£2.95	£4.26	

Table 5-9	Summary	of Case	Results.
-----------	---------	---------	----------

To sum up, this case has shown how important is to create a scenario considering all losses. Although this did not directly affect the turbine selection, LCOE fluctuated considerably. The average LCOE was around 9.5% higher than the reference scenario for scenarios IA, IIA, and IIIA. For turbine IIB, this difference was 14%. This indicates that the costs can vary more than expected, which could explain why some projects fail to cover their costs in the medium and long term. The results indicated the importance of investigating different scenarios, turbine classes, and loss trends within the scope of economic analysis. Furthermore,

while the technology is improved to overcome these trends, a better economic analysis is essential to identify the actual costs and de-risk in projects making them more attractive to investors. Although the results can be considered consistent, some assumptions were simplified. An increase in loss rates was considered to demonstrate higher wear of equipment. However, a load fatigue analysis should be performed together for a more precise cost estimation since the impact on costs could be higher than estimated here.

5.4 Chapter Summary

Some wind energy projects are failing to deliver what they were expected in medium and long term. This is mainly due to increase of operational losses and costs along its lifespan. In Section 5.1, the three main losses trends were identified along with relevant figures. In Section 5.2, it was investigated the impact of these trends on LCOE and NPV and the "Critical Year" and "Breakeven Year" were also identified. The cost estimation was based on average values from the literature for offshore deployments. Furthermore, a stochastic analysis using MCS was conducted, followed by a sensitivity analysis. In Section 5.3, a hypothetical onshore farm was modelled in a CFD software. The AEP, wake-effects, and other losses was estimated considering four different classes of turbines. Then, a cost-assessment was performed including the losses trends scenarios. The inclusion of the operational losses proved to be useful for decision makings and monitoring of project risks. It also has the potential to provide insights into the actual returns of proposed solutions aimed at reducing operational losses, as will be discussed in the next chapter.

109

6 SOLUTIONS TO REDUCE OPERATIONAL LOSSES

As discussed in previous chapters, operational losses have a big impact on the costs of wind energy, especially in medium- and long-term analysis. This means that there is space for improvements and opportunities to increase the asset value and attract more investments. The main reasons for operational losses are the increase of failure rates, ageing, and curtailment. In this chapter, some of the main solutions proposed to reduce them will be discussed. Additionally, a brief economic analysis of these solutions will be performed considering the economic model developed in the chapter 5.2. Thus, the costs and scenarios will be the same as the ones presented in Tables 5-2 and 5-3. In case of values with different currencies, they were converted into sterling pounds at the same rate presented in Table 5-1. Finally, in section 6-4 a multivariate MCS including a range of loss trends and the findings and figures from solutions discussed in previous sections is presented.

6.1 CMS

There are three main types of maintenance: reactive, preventive, and predictive. Reactive maintenance occurs when equipment is visibly not working correctly, or when a failure has already happened. Preventive and predictive maintenance aim to prevent equipment failure before it happens. Preventive maintenance is usually performed periodically, with technicians checking equipment health and performing scheduled maintenance routines such as oil and filter changes, general cleaning, tightening bolts, or replacing damaged seals and roller bearings. Predictive maintenance involves the use of gadgets, sensors, or transistors to monitor and identify abnormalities in the equipment that are difficult to be identified by technicians. Figure 6-1 shows a potential failure (PF) curve, illustrating the reduction in performance over time before a failure occurs. It is possible to identify early signals of abnormalities during this period, which may vary depending on the equipment or component. However, in some cases, the start of failure can be identified months or even more than a year in advance [267].



Figure 6-1 P-F Curve [268].

One common approach to monitor the safety and health of a machine is the CMS (Condition Monitoring System). This system is composed by a combination of sensors that can measure in real time different parameters, such as vibration and temperature of several components. Following these parameters could assist operational team to act before the failure happen or at least prepare beforehand to the intervention. As mentioned before, the time spent with repair itself is around 20 to 40% of the downtime, great part of this is due to logistics and availability of spare parts [52][192]. Thus, only by reducing the waiting time of downtime, the productivity could significantly increase. It is important to mention that to be effective, the CMS still relies on interpretability of the parameters and maturity of the team to act at the correct moment. For that, a further investment in training of operators and maintenance team, improvements in the process, or even the use of algorithms to assist in decision makings and develop a reliable historical data repository needs to be considered.

Accessing the real costs and benefits of CMS can be challenging, as they are dependent on each project, and manufacturers often do not share this information, considering it strategic and confidential. However, some studies tried to estimate them. [269] found an increase of almost 10% in the revenue and an increase of one point in the availability rate during the period simulated. [270] developed a Life Cycle Cost (LCC) analysis and estimated a benefit 36% higher than the scenario without CMS. [271] developed a method considering CMS and could reduce on average about 35% of maintenance costs. Nonetheless, the trade-off of CMS cost and possible return are not clear and needs to be carefully investigated.

To [38], it is unrealistic to install transistors to monitor all individual component, because of lack of space and high cost. According to [272], the payback of CMS investment could be from 5 to 7 years considering 45% downtime reduction, but 10 years if 35% is selected. Although [273] recognizes the benefit to the maintenance team of adopting CMS, in some of the scenarios investigated, the economic return is achieved only if 47% of reactive maintenance become preventive. [274] points the importance of integrating results with SCADA, to avoid false alarms or missed faults, and the real performance and cost-benefit is almost unknown. The interpretation complexity of faults report and the difficult to identify false alarms have discouraged operators to use CMS effectively, although most of wind turbines in Europe (with rated capacity above 1.5MW) are fitted with this system [275].

In this thesis, the impact of LCOE will be investigated before and after CMS implementation. To this end, the offshore economic model developed in section 5-2 is used. The investment cost by unit of power installed was £16,607.50/MW [276], £28,666.67/MW [270], and £63,266.34/MW [272]. The increase on O&M costs from the same references was £730.00, £2,580.00, and £2,075.00, respectively. Regarding the mitigation of losses, it was considered a reduction in downtime of 40% and 1%, 0.1% and 1% at the yearly rate from the scenarios A, B, and C, respectively. Finally, the CMS costs assumed were the average of values discussed in this paragraph.

Table 6-1 presents a comparison of the LCOE for the offshore scenarios investigated with and without the use of CMS. The low impact in the reference scenario suggests a low return on investment. However, by including the loss trends in the analysis, the economic benefits of CMS become more evident. With effective maintenance, the equipment is expected to perform at a high level of effectiveness over its lifespan, thereby reducing wear on components. Table 6-2 presents the financial return of this simulation, considering £/MW installed. By factoring in the likely increase in losses over the equipment's lifespan, CMS plays an important role in reducing costs, and the real return on investment could be higher than 5%. This highlights the importance of considering loss trends in any analysis of CMS efficacy.

	LCOE (£/MWh)	LCOE + CMS (£/MWh)
Reference	£98.93	£98.51
Scenario C	£102.41	£101.20
Scenario B	£103.64	£102.13
Scenario B + C	£107.16	£104.86
Scenario A	£110.89	£106.76
Scenario A + C	£114.69	£109.65
Scenario A + B	£116.34	£110.75
Scenario A + B + C	£120.19	£113.67

 Table 6-1. Comparison offshore case with and without CMS.

		Without CMS With CMS				Financial			
	LCC	Revenue	NPV	Return	LCC	Revenue	NPV	Return	Comparison
Reference	£3,979,555.74	£4,481,164.91	£501,609.17	112.60%	£4,036,324.42	£4,564,435.83	£528,111.41	113.08%	0.43%
Scenario C	£3,979,555.74	£4,329,043.13	£349,487.39	108.78%	£4,036,324.42	£4,442,977.77	£406,653.35	110.07%	1.19%
Scenario B	£3,979,555.74	£4,277,632.29	£298,076.55	107.49%	£4,036,324.42	£4,402,488.20	£366,163.78	109.07%	1.47%
Scenario B + C	£3,979,555.74	£4,136,882.55	£157,326.81	103.95%	£4,036,324.42	£4,288,039.77	£251,715.35	106.24%	2.20%
Scenario A	£4,327,118.39	£4,346,892.30	£19,773.91	100.46%	£4,313,865.54	£4,501,285.24	£187,419.69	104.34%	3.87%
Scenario A + C	£4,327,118.39	£4,202,830.44	-£124,287.95	97.13%	£4,313,865.54	£4,382,722.81	£68,857.27	101.60%	4.60%*
Scenario A + B	£4,327,118.39	£4,143,359.67	-£183,758.72	95.75%	£4,313,865.54	£4,339,337.61	£25,472.06	100.59%	5.05%*
Scenario A + B + C	£4,327,118.39	£4,010,669.86	-£316,448.53	92.69%	£4,313,865.54	£4,227,784.81	-£86,080.74	98.00%	5.74%**

Table 6-2 Results from financial return – With and without CMS.

*This result ignored that first scenario had negative return. **This result ignored that both scenarios had negative return.

6.2 Maintenance x Ageing

Good maintenance activities are beneficial not only to increase the availability but also in deaccelerating the ageing effects, extending its lifespan. In other words, it means the equipment can work for a longer period than expected with similar or acceptable performance levels at lower failures rates. [277] shown how an enhanced risk-based maintenance could increase the useful life of elevators from 25 to 35 years. [278] demonstrated how early decisions can influencing greatly the possibility of increase military equipment life. In this study, it was indicated the cost effectiveness should be assess for the decision about extension and how maintenance can contribute to that.

The same behaviour is expected to wind turbines. Although part of the ageing effect is caused by external factors, a bad maintenance can also directly influence on it. High temperature, high vibration level, excess of dust, and/or loose bolts, are example of factors that forces the turbine to operate unbalanced, which can wear components faster and reduce performance of the turbine with age. In this scenario, the productivity is reduced in three different ways. Increasing failure rates and downtimes, shortening equipment lifespan, and reducing performance during operational useful periods.

[16] has found that the decline in performance is also influenced by maintenance cost-benefit trade-offs. After the eligible period in which farms have access to Production Taxes Credits (PTC), the yearly performance ageing rate dropped from 0.53% to 1.27% per year. To [192], there is a strong relationship between operational maintenance expenditure and annual energy production. [16] also found an increase in curtailment during the period that the benefit is not paid, which indicates an attempt to maximize profitability from operators.

The extra credit will depend on the type and size of the project. The full credit value in 2022 was \$26/MWh [279], but some projects could be eligible to \$10/MWh [280]. In UK a similar benefit is the Feed-in Tariffs (FIT), which also is

dependent on the project, and can vary from less than £5 to above 20£/MWh for wind projects [281].

To understand the impact of the maintenance and ageing trade-offs, three different scenarios was defined (Figure 7-2). The values assumed in this analysis were the same presented in Table 5-2. Considering the figures mentioned in previous paragraph, it was assumed a reduction of £10 after the 10th year. This year and the ageing rates were retrieved from the investigation performed by [16]. Therefore, Scenario I, after 10th year, the ageing rates dropped from 0.53% to 1.27% per year, and half of maintenance expenses is cut. Scenario II, the reduction is accepted, without changing the maintenance routine, so the ageing is kept equal to 0.53% the entire period. Scenario III is the intermediate scenario, in which the ageing rate is set at 0.9% and maintenance costs reduced by 25%. It is important to note that maintenance costs are only a part of the OPEX, which was previously discussed as accounting for 49% of the total OPEX.





Table 6-3 presents the results of the simulation. While the reduction in the strike price is significant, nearly 10%, keeping the maintenance activity would result in higher revenue for investors in the long term, as demonstrated in Scenario II. The NPV for Scenario II was above £200,000, which is 50% higher than Scenario I, where maintenance is reduced to compensate for the end of tax credit. Furthermore, the expected performance of the wind farm after 20 years of

operation would be 0.894, indicating a decay of 10% compared to when the turbines were new. For Scenario I and III, this loss would be 27% and 18%, respectively. Thus, Scenario II presents a higher likelihood of having its life extended, making it both viable and profitable.

	Scenario I	Scenario II	Scenario III
LCOE (£/MWh)	£104.84	£102.74	£103.71
NPV (£)	£126,188.59	£201,866.34	£166,185.73
Performance (After 20 years)	0.729	0.894	0.814

 Table 6-3 Results from Trade-off Maintenance x Ageing Investigation.

It is important to note that this analysis does not take into account certain factors that are crucial for investors, such as opportunity costs and investment returns. Additionally, the figures presented in this analysis reflect the entire 20-year lifespan of the project and do not consider the potential risks and reduced margins associated with delaying outcomes from an investor's perspective. However, this simplified simulation serves as a demonstration of the importance of monitoring the financial aspects of the project and exploring various scenarios before making decisions.

Figure 6-3 presents a sensitivity analysis of NPV, examining various combinations of maintenance cost reduction and ageing rates. The graph illustrates the results, with green values indicating increased returns compared to the base scenario (dark blue), and red values representing negative NPV. The ageing rates span from the reference rate (discussed earlier) in the middle, to extreme scenarios with no change on the left and double the estimated rate on the right. Five rates were considered in total, evenly spaced. Additionally, five levels of maintenance reduction were investigated, with a maximum assumed reduction of 50%. It is worth noting that this value represents an extreme scenario and is highly unlikely, particularly for projects with over 10 years of operational life. Overall, this range of values encompasses a wide spectrum of scenarios, covering the most probable situations in real-world settings. For reference,

maintaining the tax credit throughout the entire period would result in an NPV of £335,506.18.

Sensitivity Analysis						
		Ageing After 10th Year				
		1.0053	1.009	1.0127	1.0164	1.021
uo	0%	£201,866.34	£117,915.10	£ 29,647.35	-£ 63,151.92	-£185,183.62
rcți	10%	£221,174.59	£137,223.35	£ 48,955.60	-£ 43,843.67	-£165,875.38
edt	20%	£240,482.84	£156,531.60	£ 68,263.85	-£ 24,535.42	-£146,567.13
t. R	30%	£259,791.09	£175,839.85	£ 87,572.09	-£ 5,227.17	-£127,258.88
ain	40%	£279,099.34	£195,148.10	£106,880.34	£ 14,081.08	-£107,950.63
Σ	50%	£298,407.58	£214,456.35	£126,188.59	£ 33,389.32	-£ 88,642.38

Sensitivity Analysis

Figure 6-3 Sensitivity Analysis of Maintenance Cost Reduction X Ageing rates.

6.3 Power Integration Solutions

As renewables increase its share in energy market, more concerns about level of curtailment are addressed. Energy relies on the balance of production and demand, therefore, as renewables are considered non-dispatchable source, grid operators limit its penetration to avoid stresses in the grid. This creates a clear conflict of interests between grid operators and renewable plant operators. While the first tries to guarantee supply, with required quality and safety in the most economical way, the second needs to enhance power output at maximum to minimize costs [220].

To become economic viable and competitive in energy market, especially offshore projects that are more costly, a high productivity is extremely important. If the energy is not purchased by the utility, the revenue is reduced [213], whether is curtailed or rejected (some exceptions can occur, depending on the commercial arrangement). Therefore, some solutions have been proposed to increase the wind energy penetration, which can vary according to the country and grid. Although in some situations, especially in short-term higher wind penetration might occur, generally, grid operators limit wind energy penetration at rates that varies from 20% to 35% [210]–[212].

The main problem of non-dispatchable energy is its uncontrollable nature and its high fluctuation. This suddenly change in production level can cause disturbances in the grid. One widely discussed solution nowadays to reduce curtailment losses is the Energy Storage System (ESS). In this system, the extra production, when there is a lack of demand, is stored to be used in periods where the production fluctuates significantly. Thus, the ESS keeps the output stable and become able to cover rapid ramp downs. There are different system types to store energy, as pumped hydro, compressed air, flywheel, electromechanical, super capacitors, and finally, through batteries [220][174]. None of them is capable of meeting both energy and power density simultaneously [174], however, battery is becoming popular due to its recent cost reduction [282].

Apart from ESS, there are other solutions to reduce wind energy curtailments, which includes: wind forecasting, demand-side management, electric cars, smart grids, super grids, and the adoption of electric water heating [283]. All these solutions are related at some level to increase the demand, bringing some benefits for consumers to use energy during predicted windy periods. Green Hydrogen is also getting popular, where hydrogen fuel is generated through renewables. According to [284], green hydrogen helps to overcome grid constraints of offshore deployments and it can be also cost-competitive with hydrocarbon fuel, if untaxed. Another common solution is overplanting, where to reduce curtailment, wind farms are built with higher capacity than the electrical infrastructure [165]. In this scenario, the individual turbine's performance is reduced, so operators can regulate power output of the farm to increase overall productivity. This solution also minimizes losses related to availability.

As the reason behind overplanting is to reduce farm energy rejection and curtailment, there are a limit at which overplanting is viable. Obviously, this depends on each project and the grid capacity. According to [165] overplanting from 2 to 8% can reduce total LCOE, and the optimal scenario found in their MCS was 4%. Some countries have suggested overplanting from 5 to 20%, to prevent grid restriction [285]. Nonetheless, the average LCOE reduction was about 0.5% in both cases.

120

To [286], the level of sophistication and costs required to transform wind power into largely dispatchable are controversial. For that reason, it is important to perform reliable economic analysis. It is out of scope of this thesis to investigate all pros and cons from each solution proposed. However, to check some of the findings related to overplanting, a simplified economic analysis was simulated. For this, a reduction of 10% in availability and performance losses at the reference scenario was considered as well as 50% in quality losses. To achieve average 0.5% LCOE reduction, 3.7% of overplanting was needed. This result aligns to what was found in the literature.

6.4 Multivariate Monte Carlo Simulation Case

Wind energy is surrounded by a diverse cause of uncertainties, which includes ambient conditions, wind features, dispatchability to the grid, levels of wear out and its costs. Therefore, a stochastic analysis as MCS is more appropriated than deterministic models to assess its viability and risks. In subsection 5.2.4, an offshore model was developed considering different scenarios of operational losses trends. Although, those rates can be considered reliable, since they were based in strong evidence retrieved from the literature, they can vary according to the project. Another two important aspects that can affect economic viability assessment are: how some early decisions and the use of solutions, as described previously in this chapter, can impact cost models. To this end, a more elaborated MCS, called here as Multivariate MCS (MMCS) will be performed and discussed in this section.

Table 6-3 presents the range of operational losses trends that will be added to the model as well as some rules and assumptions created. Offshore figures can fluctuate considerably according to how far they are from the cost. More distant projects are only viable if the wind regime is higher, so it is reasonable that these projects contemplate higher capacity factor. Therefore, high CAPEX will randomly select a CF from upper limit and an additional rate was inserted to be more realistic. Another rule defined was about OPEX. To [263] a reasonable estimation is an annual OPEX equals to 2% of its CAPEX for offshore projects. However, more recent studies that also consider floating projects has considered higher values. In the literature review developed and presented in Table 5-1, this value is on average 3.2%. A higher investment in maintenance can be related to ageing levels, as demonstrated in section 6.2. Therefore, for relative OPEX costs above 3%, the ageing and failure rate trend will be selected randomly in its lower range.

Losses rates	Range	
Failure increase rate	Linspace(1.02, 1.10, 9)	
Ageing rate	Linspace(1.001, 1.009, 9)	
Curtailment rate	Linspace(1.02, 1.10, 9)	
Criteria	Rule	
If OPEX/CAPEX > 3%	Ageing rate & Failure rate:	
If CAPEX > £3,000,000	F.C. range: + 0.02	

Table 6-4 Operational Losses Trends and Rules for MMCS model.

The adoption of overplanting and the use of CMS was also included in this simulation. For this, a random selection of 'Yes' and 'No' for each category was included in the algorithm. Table 6-4 presents the rules used as discussed in each of them. The overplanting calculated in Section 6.3 was 3.7%, however, this analysis only contemplated the reference scenario. Since overplanting makes sense only if some return is expected, this value was reduced to 3%, which makes less likely to have higher LCOE with overplanting in the model. The introduction of new wind farms in the future projects and/or expansion of transmission lines are not in control of operators, so a low curtailment rate was assumed, varying from 0 to 2% per year. The CMS followed the same assumptions discussed before.

 Table 6-5 Overplanting and CMS Rules

Extra Implements	Rules
If overplanting == 'YES'	CAPEX, OPEX, DECEX = Increase of 3%. Availability Losses: 10% Reduction Performance Losses: 10% Reduction Quality Losses: 50% Reduction Curtailment Rate: linspace(1.00,1.02,3)
If CMS == 'YES'	CAPEX = CAPEX + £36,180.17 OPEX = OPEX + £1,795.00 Availability Losses: 40% Reduction Ageing Rate: -0.001 Failure Rate: -0.01

Figure 6-4 presents the NPV balance and the average LCOE results from 100,000 iterations. Considering all new criterions more than half of scenarios simulated brought economic benefits to investors. The average LCOE was \pounds 107.61, which is around 10% higher than the reference scenario presented in Table 5-5. However, it is important to notice that the fluctuation in cost range is still high. Considering an interval of confidence of 95%, the LCOE can vary from \pounds 82.35 to £133.87.



Figure 6-4 Result from Multivariate MMCSTable 6-6 summarizes the values into four categories: those without solutions, those with CMS, those with Overplanting, and those with both solutions. On average, the use of both solutions achieved the lowest average LCoE, £102.90. By using only one of the solutions, the return was similar to total average. For not using any solution, the result was £112.87, almost 5% higher than the average value.

	Sample Size	Average LCoE
Both	24970	£102.90
Overplanting	24868	£107.32
CMS	24880	£107.29
Without	25282	£112.87
Total	100000	£107.61

Table 6-6 Summary of MMCS per solution type

In conclusion, the model has demonstrated the significant impact of uncertainties on the economic lifespan of wind energy deployment over the medium to long term. It's crucial to investigate the loss trends to mitigate risks and anticipate future patterns before losses become irreversible, leading to premature project end-of-life. This study investigates two popular solutions - overplanting and CMS - to reduce operational losses and increase production, resulting in a reduced LCoE. Unlike other studies that overlook loss trends, the return on investment for these scenarios is higher than what is typically reported in literature. In other words, investing in efficient maintenance and loss reduction solutions may not yield immediate returns but can decelerate loss trends, reduce risks, and increase financial returns over the medium to long term. Appendix B presents the MMCS for return rates of 4% and 8%.

6.5 Chapter Summary

This chapter explores several solutions to reduce operational losses in wind energy deployment. Section 6.1 discusses the Condition Monitoring System (CMS), which can have a low return on investment due to its high investment and requirement for specialized expertise. However, in a long-term analysis that considers the reduction of loss trends, CMS has been shown to have a more significant impact on reducing LCoE. Section 6.2 investigates the relationship between ageing and maintenance activities, finding evidence of correlation between the two. Section 6.3 discusses solutions to increase power integration and reduce curtailment losses. While some of these solutions are gaining attention, they still need a reduction in investment costs to become more popular. All the solutions were analysed using the cost model presented in subsection 5.2.4, and the values align with other studies. To further investigation of the effectiveness of these solutions, a Monte Carlo simulation (MMCS) was performed in section 6.4. The simulation created a range of possible loss trends and included the use of some of the solutions in the model.

7 DISCUSSION

Wind energy is a promising solution to reduce fossil fuels dependency. However, it has been reported that some projects are forced to be shut or have significant refurbishments before completing its projected life, affecting investors return. The findings of this study indicate that the increase of failure rate along turbine's life is the main reason for economic failure of wind energy assets. Although some studies have pointed to this problem, suggesting even a yearly increase in the OPEX on the economic models [255], they seem to be still underestimated. The increase of failure rate affects not only the OPEX costs, but also the availability of the turbine, consequently reducing its production, and this additional loss is not commonly added in economic analysis found in the literature.

The concept of "critical year" developed and presented in the section 5.2.4 tries to predict when the cost would overpass the benefit if operational loss rates remained constant. Although the critical year considering failure rate scenario (Scenario A) was close to the 11th year of operation, period where according to [214] wind turbines require significant refurbish due to increased maintenance costs, more studies are necessary to validate the rates assumed. Apart from the increase of failure rates, other two possible causes were identified, ageing and curtailment. It is likely that all three losses and trends will occur together at some level. In this scenario, the critical year was the 7th year, indicating that possibly some of the assumed rates can be considered high.

The inclusion of operational losses in economic models can be helpful not only to estimate investment risks, but also to understand the real return of some practical solutions. Referring to one of the solutions proposed in Chapter 6, [272] suggests that the investment in CMS could be paid back within 10-year period. Therefore, considering the projected operational life and the extra costs and risks involved to adopt CMS, this might not be very attractive for investors. However, none of the assessed studies reported on the potential reduction of degradation, which could mitigate ageing effects and slow down the growth of failure rates along turbine's lifespan. Similar benefit is demonstrated in the sections 5.3, 6.1 and 6.2.
Throughout the thesis, several economic analyses were conducted. The main findings indicate that the LCOE is underestimated by 10% in the MMCS results in section 6.4 when compared to the reference scenario presented in subsection 5.2.5. Both scenarios used the same cost base. However, by implementing the two solutions discussed to mitigate operational losses and degradation, overplanting and CMS, this difference could be reduced to 5%. In today's highly competitive market, differences of this magnitude can be crucial for the wind energy industry.

Regarding the adaptation of the metric OEE to wind energy assets, the findings was unexpected positive. Despite the uncontrollable environment and uncertainties surrounding wind power, the first case study (presented in section 3.5) achieved an overall productivity rate of close to 91%. According to [23], an OEE higher than 85% is considered world class benchmark in manufacturing industry. This result indicates that the technology can be considered reliable enough to deliver what they are expected. Although the second case study (demonstrated in section 4.2) achieved 73% productivity, the short period assessed could explain its lower result. Also, it is clear that at first half of the data provided some abnormalities have occurred. By eliminating this period, the OEE would be close to 81%.

It's worth noting that both case studies were based on early-age wind projects, which suggests a likelihood of bias in the results. Also, in this thesis, the OEE analysis covered the entire energy production process, not just the turbine. This means that external factors such as curtailment, which may not always be the responsibility of operators, were also taken into account. Although there is no consensus on whether external losses should be included in OEE analysis, the energy industry is inherently interdependent, with production relying on a balance between production and grid demand. Thus, it is crucial to consider these external factors in the OEE analysis as well as in the economic models, particularly because depending on the contract, these losses can reduce revenue and consequently the viability of project. Although in some cases, the government may cover these losses financially, consumers would ultimately bear the costs

through higher energy prices. Therefore, the strategy behind this was to avoid bias in the analysis by checking the real costs and productivity of the assets.

In conclusion, the successful accomplishment of these two main aims, which were to adept OEE to the wind energy context and investigate the underlying factors contributing to occasional economic failures in wind projects, reinforces the importance of carefully assessing operational losses in wind energy for informed decision-making and comprehensive risk assessments. The creation of a unique metric that encompasses all possible operational losses has the potential to provide a more comprehensive overview of the equipment's performance. The novelty lies especially in the inclusion of the quality element, usually neglected or assessed separately in other studies. Also, as discussed in Section 2.3, improving one specific part of the process can have negative impacts on other parts. Therefore, an overall analysis of the entire process, such as the adaptation of OEE, can be crucial in ensuring that the project reaches its full productivity capacity. It is important to mention that to be more effective, academia and the industry should try to use a unique procedure to calculate the OEE, so the results can be fairly compared. In any case, the scheme shown in Figure 2.2 has the potential to be kept as the basis of future analysis, since this would inevitably cover the entire process and consider all operational losses. Additionally, an attempt to estimate quality element was developed in subsection 4.2.3, while in subsection 4.2.2 and section 4.3, the performance element was investigated, where different options were presented. To the best of knowledge of the authors, this is the first practical attempt to adapted OEE in wind energy, which means that there is room for future improvements.

Regarding the economic analysis, as mentioned before, it was identified and investigated three main operational loss trends that could affect productivity along wind energy operational life: the increase of failure rate, ageing, and curtailment. The addition of these operational loss trends demonstrates when and why the project can fail before its projected life. It is very likely that if a project is failing economically, part of these rates has been neglected or underestimated. The study has revealed that ageing and curtailment have minor impacts on the overall

128

costs, despite the latter being a widely discussed topic in recent papers. However, it is important to keep in mind that grid structures and transmission lines usually take longer to be construct [220] than the wind farms, so this could become a big problem in near future. The increase of failure rate has shown to be the most impactful loss trends, which means that OPEX and maintenance are still the main bottleneck to wind energy to become completely independent from subsidies, competing directly in the energy market. It is important to remind that the rates outlined here must be carefully reviewed and tailored to the specific technology and scenario at hand. Although, the economic analysis used on the stochastic analysis was based in figures found on the literature to only offshore deployments, the particularities, country, and type of technology were not considered. As demonstrated in reference [255], which investigates LCOE by country, the costs can vary significantly, with the most expensive option being twice as costly as the cheapest. Also, it is expected that the implementation and operational costs of floating offshore wind turbines (FOWT), for instance, might be higher than from bottom fixed structures, which are closer to the shore. This indicates that the larger cost ranges assumed could be narrowed for specific projects, providing a more precise result.

Beyond the results itself, the main contribution of this work was the development of a framework that can be easily adapted and tailored to different deployments. Additionally, it is expected that the methodology proposed would give a broader overview of possible causes of economic failure and also, considering the adaptation of OEE, a reliable metric to help operators to identify and eliminate wastes in the process, improving productivity and profitability. Although the focus of this thesis was on wind energy assets, the proposed methodology and framework are versatile enough to be applied to different technologies, particularly in the field of renewable energy, such as solar energy. Solar energy shares similar characteristics with wind energy and can thus easily adopt the principles and foundations developed here.

As implied throughout this chapter, there are certain limitations in this thesis that need to be acknowledged and addressed in future studies. The primary limitation

129

is related to limited data due to confidentiality concerns. This is a common problem faced by researchers to develop an economic analysis in wind energy, which can reduce the confidence of results. As demonstrated by [287], the current strike prices from UK auctions are approximated half of the average LCOE predicted by different sources in the literature. This means that either the costs discussed in the literature will reduce drastically in coming years or the strike prices are underestimated, and those future projects might not be viable its entire life. Also, some generalizations had to be done to perform the economic analysis and should be revisited before assuming the rates stablished here. With regards to the estimation of OEE, the quality analysis was simplified by solely taking into account ramp rates and wind energy rejection, as demonstrated in subsection 4.2.3. However, it should be noted that by following only this proposed quality analysis, many identified quality losses discussed in subsection 2.2.4 were disregarded. Different from manufacturing industry, measuring quality of the energy produced involves some subjectiveness and for better use of the tool this should be better defined in future to be efficiently measured and related to operational performance of the turbines. Finally, it would be important to extend the OEE analysis to older projects to observe how the metric changes over time. Moreover, future work could explore other factors that might impact OEE, such as workforce training and equipment maintenance.

8 CONCLUSION

This chapter will conclude the research by summarizing and presenting the main findings related to its aims and objectives mentioned in the introduction chapter. Section 8.1 will present a summary of the chapters. Section 8.2 will discuss the thesis contributions. In section 8.3, the limitations are discussed, while in 8.4 a suggestion of future works.

8.1 Summary of Chapters

This thesis started with an introduction of the theme and the problem that would be investigated (Chapter 1). Some wind energy deployment is failing to deliver what they were expected in medium and long term and becoming economically inefficient before the end of its projected lifespan. Thus, the problem background and the scope of the thesis was defined together with its aim and specific objectives. Later, the structure followed along this work and the works and papers published and submitted are presented.

The goal of chapter 2 was to identify the main operational losses in wind energy assets. The manufacturing industry has developed a metric called OEE to gather all operational losses and assist operators and managers in decision making. Therefore, the first part of the chapter presented this metric. Then an adaptation of OEE to wind energy was defined and finally an extended literature review was performed. The losses identified were classified according OEE elements and the benefits of this metric were discussed.

While the previous chapter focused on the qualitative investigation, chapter 3 estimated and established the general rate losses of each element of OEE. It was clear that these rates are not accounted following a common rule, which can reduce the reliability and make comparisons a hard task. Nonetheless, all these aspects were considered and a general OEE with its likely range was summarized for onshore and offshore projects. In section 3.5, a real case scenario was investigated and presented.

Chapter 4, through real data, some alternatives proposals to monitor and evaluate wind farms performance and productivity were discussed. First, using MET MAST data information, a script using traditional methods of machine learning was presented and have shown higher accuracy, even considering daily average values. Then, the OEE of a wind farm from provided SCADA data was evaluated. Due to the limitations of the data, part of losses was estimated, as the quality rate in which a simple procedure considering ramps rates and wind energy rejection was developed. The values were aligned with what was found in the literature.

Chapter 5 identified possible reasons for the farms become inviable in medium and long term. The main aspects identified were the increase of failure rates, ageing and curtailment. The latter is getting much attention, because with more wind farms built, curtailment tends to be higher without changings in the grid and/or expansion of transmission lines. An economic analysis was performed considering the average of costs found in the literature and the losses trends investigated. In section 5.3, an onshore hypothetical farm was simulated in a CFD software, and its costs were calculated considering the losses and comparing the selection of different wind turbine classes.

Chapter 6 shown some important and popular solutions to reduce the operational losses discussed in the previously chapter. While the thesis presents several interesting proposals such as new controllers, gadgets, and machine learning models, this chapter focuses on discussing CMS, the trade-off between maintenance efficiency and ageing, and solutions to increase power integration. The offshore figures discussed in chapter 5 were used to investigate and demonstrated possible economic return by adopting the solutions. Finally, a multivariate MCS is performed integrating the solutions and different trends rates randomly.

Chapter 7 presents a discussion about the main findings, some of the impacts of this study and some limitations.

8.2 Thesis Contributions

Through different case studies and an extended literature review, this thesis found an indication of main problems that can reduce performance of wind energy assets along its lifespan. The impact of each loss trends was investigated stochastically considering different scenarios. Also, a metric was proposed to monitor and measure the operational losses and productivity stability of wind energy farms. The novelty, soundness, and value from each specific objectives presented in Section 1.2 are summarized in tables 8.1, 8.2, 8.3, 8.4, and 8.5. "Novelty" outlines what is new to academia and/or industry. Soundness sums up the methods used and how they were applied. Finally, the value explains why and to whom the outcomes can be significant.

Table 8-1 Summary of Objective 1

Objective 1: Review of operational losses throughout the entire process, including wind penetration into the grid, and classify them into groups following OEE concepts to help operators to identify specific areas for improvements.

Novelty	Soundness	Value
At best of knowledge of the authors no similar work has been developed so far. Most of reviews identified in the literature treat specific groups of operational losses. In this activity, they were all put together, including some third-party causes, giving a widely overview of the losses and risks of wind energy assets. The focus of this review was on publications starting from 2010.	This extended review used several key words as "operational losses", "quality losses", "performance losses", "availability", "grid losses", together with "wind energy" or "wind power". The losses were classified in three main elements, following basic concepts of the metric OEE. Apart from the classification, it was indicated what sort of discussion and solutions were presented.	This review might be for interests of not only researchers, but also for the industry. Although there are many papers discussing the operational losses, they are not treated together. This review summarizes all possible operational losses and also has presented a reasoning line to accommodate them in the elements of OEE. This criterion has the potential to become the basis of OEE adaptation to wind energy assets. Additionally, some solutions are presented, and they were grouped according to its proposal.

Table 8-2 Summary of Objective 2.

Objective 2: Apply OEE to wind energy following framework previously developed and perform quantitative analysis according to literature figures and real cases to evaluate wind energy effectiveness.

Novelty	Soundness	Value
OEE is a popular metric in manufacturing to measure equipment stability and productivity by assessing its operational losses. The advantage of this tool is that it puts all losses together, which contributes to find the best global solution. What outstands this metric from others is that it includes quality losses, often neglected in turbine performance analysis.	By following a linear reasoning strategy, developed in the review, the losses were grouped in the three elements of OEE. A further quantitative review was proposed to estimate the average rate of each OEE element for onshore and offshore projects. Two real case scenarios from early age projects were also performed to this end.	Any changing in a process can bring consequences in other stages of the process. This argument is also valid for wind energy assets. For the best of knowledge of the author, it was not found any metric where all possible losses are put together for wind energy, especially the ones related with quality. This metric is known as an important tool not only to measure equipment stability, but also to help to identify hidden losses to improve productivity and profitability.

Table 8-3 Summary of Objective 3.

Objective 3: Conduct data analysis to develop different models and procedures to monitor and estimate OEE.

Novelty	Soundness	Value
The adaptation of OEE proposed is a novelty itself, however, there is no procedures regarding the application and how the rates should be accounted. For that, some common metrics was investigated, and data analysis and machine learning methods used.	Availability is already a well-established concept in the literature. Although there are different options to calculate it, it was followed the time-based concept. Performance and quality, on the other hand, had to be developed. From a data analysis activity, the best performance registered in similar operational conditions was determined as reference. To the quality, by limited data, a script considering ramp rates losses and wind energy rejection was created. Therefore, OEE could be estimated in real scenarios. To monitor performance through MET MAST data, a model was developed considering RFR, achieving a high accuracy for wind power output prediction.	Estimate and monitor OEE can be beneficial to operators to identify hidden losses and understand better how the equipment is performing along its life. The main value of this tool is that it includes all elements in one metric, including quality. The ramp rate x wind rejection criteria developed provided coherent outcomes and could become a starting point for a more complete assessment. The cases have shown the potential of the tool, where in similar events the performance, for instance, could be half of expected.

Table 8-4 Summary of Objective 4.

Objective 4: Investigate in literature how some operational losses increases along the lifespan of wind energy assets and evaluate its impacts on costs and viability.

Novelty	Soundness	Value
A review of operational different causes of increase of operational loss were investigated and presented together. Here, common rates of operational losses trends were defined and its impacts on viability and costs were investigated. A diverse number of scenarios were drawn, considering them individually or together, to better understanding if they could be the reason of economic failures in the projects.	The main causes of losses of productivity along wind assets lifespan were investigated through literature review. Scenarios considering likely loss trends were created and added to economic analysis to check the impact in the LCOE and its viability. The critical and breakeven year of each scenario was also assessed. A hypothetical onshore farm was created to assess the impact in costs from different wind turbine classes with different scenarios.	Before deciding to participate in a project, investors want to have the risks drawn as clear as possible. By including the loss trends, a new range of possibles outcomes are created. It is not expected that operators will accept the losses without any action to minimize them. Therefore, the simulations are important to indicate how the costs could be in future and the consequences in case the trends keep growing at same rate. Some of externalities, such as curtailment by lack of demand was also investigated. Although this is not always responsibility of operators, depending on the contract their revenue can be affected, for that reason, they should be considered.

Table 8-5 Summary of Objective 5.

Objective 5: Perform stochastic analysis to check viability, costs, and risk of economic failure of wind energy assets along its lifespan in different scenarios, combining different losses trends and some proposed solutions to reduce operational losses identified in literature.

Novelty	Soundness	Value
Some solutions to reduce losses were investigated, including CMS to reduce breakdown, and overplanting to increase wind dispatchability. Also, the trade-off between maintenance and ageing was considered. Thus, a stochastic analysis assuming not only costs uncertainties, but also, the losses trends rates and some of solutions discussed were performed.	Due to the number of uncertainties in wind energy assets, stochastic analysis is the best approach to assess risk in wind assets. Two analyses were performed here. First, the impact of losses trends rates was investigated through MCS, considering only costs uncertainties. Later, a MMCS was run considering costs uncertainties, different losses trends rates, and finally, the implementation or not of some of solutions previously discussed.	As demonstrated in this thesis, besides uncertainties on wind regimes and operational costs, loss trends rates, curtailment risks, and the return of some solutions can also affect final cost of wind energy. Therefore, the MMCS proposed can give a broader view of the risks to investors. Although the range of costs can be considered higher, the model proposed is easily adapted to attend each specific scenario in real life models. Also, some of solutions found on the literature demonstrates low financial return; however, the reduced degradation level is usually not assessed, which can mislead decisions.

8.3 Limitations

Despite of prominent results achieved in this thesis; some simple assumptions were made due to data limitation. Real costs, productivity data, and level of operational losses, are considered key and confidential by the industry. Thus, the costs and values assumed were based in the literature and some international reports, which usually consider mean values from several deployments. In this case, it might be hard to avoid some sort of bias.

Another approach to deal with the data limitation was to expand the research including data from different types of deployments, turbines, countries, and ages. Although this generalization can be considered good to bring a wide overview, this hinders specific analysis. Wind energy is surrounded by uncertainties, so some subjectiveness and simplifications are acceptable for general analyses as the one proposed here.

The incomplete data also affected the attempt to measure OEE. In the data analysis activity, several assumptions had to be done, including that no abnormalities have happened during the period assessed. This could affect the rates of each element, since it was not informed for example, if the farm was in test or normal operation, if the losses in performance was due to grid restrictions, or if the farm was penalized by any disturbance created in the grid. Nonetheless, to best use of OEE, all losses, even the ones that could be justified, should be accounted.

It is worth to add that one of the main characteristics of OEE is its simplicity. The way OEE was adapted to wind energy in this work tried to be very clear and concise, keeping the fundamental characteristic of the tool. Nonetheless, the definition of the six main losses, which ended up becoming nine, did not consider how the operational losses interact to each other. Quality losses, for instance, can cause downtime, reduced speed, and/or increase wind energy rejection and transmission losses. Thus, considering it just at the end of the process might not be the best approach. Either way, regardless of the main losses' identification, the main purpose of OEE was achieved through the procedure suggested, which

is to gather all losses in a solid metric, helping operators to identify hidden losses. In other words, by following the procedure suggested here, these losses would be included in OEE, despite of its element classification.

Finally, it is important to mentioned that this PhD has started in March of 2020, the same period of Covid outbreak. Part of the decisions and scope had to be changed and adapted, especially during the first two years, when companies, the university and the country had total or some level of restriction. Nonetheless, the overall structure and aims was kept as close as possible for the original goals and the results were satisfactory.

8.4 Future Works

This work has demonstrated the importance of including operational losses trends in economic analysis and the use of OEE for decision makings. However, some simplifications and generalizations had to be done due to the lack of complete data. Therefore, investigating if the considerations assumed here are coherent is an important sequence of this work. Also, as discussed along the thesis, OEE tries to find the best balance between all elements. Although the goal of the tool is to assist operators to find solutions to reduce all losses and not to decide the best configuration, this could still be used for this end.

Relating fatigue with operational behaviour could help to link performance with availability, for example. Also, the use of software that estimates power quality, could be related to performance. Here, only the ramp rate issue was used to estimate wind power rejection. Although the values found was coherent, further investigation is important. Creating a theoretical link between elements of OEE could facilitate comparisons between configurations and/or scenarios. Many studies presented here suggest solutions to solve or reduce losses in each element, but how this affects other parts of the process is rarely discussed. As demonstrated, an increase of performance at the cost of reducing its quality could make overall effectiveness worser than before.

REFERENCES

- WindEurope, "Wind Energy in Europe in 2019," 2020. [Online]. Available: windeurope.org.
- [2] A. Nghiem and I. Pineda, "Wind energy in Europe: Scenarios for 2030," Wind Eur., no. September, p. 32, 2017, [Online]. Available: https://windeurope.org/wp-content/uploads/files/about-wind/reports/Windenergy-in-Europe-Scenarios-for-2030.pdf.
- [3] J. Lee and F. Zhao, "Global Wind Report | GWEC," Glob. Wind Energy Counc., p. 75, 2021, [Online]. Available: http://www.gwec.net/globalfigures/wind-energy-global-status/.
- [4] A. Kolios, J. Walgern, S. Koukoura, R. Pandit, and J. Chiachio-Ruano, "openO&M: Robust O&M open access tool for improving operation and maintenance of offshore wind turbines," *Proc. 29th Eur. Saf. Reliab. Conf. (ESREL 2019)*, pp. 452–459, 2019, doi: 10.3850/981-973-0000-00-0.
- [5] E. Topham, D. McMillan, S. Bradley, and E. Hart, "Recycling offshore wind farms at decommissioning stage," *Energy Policy*, vol. 129, no. March, pp. 698–709, 2019, doi: 10.1016/j.enpol.2019.01.072.
- [6] K. P. B. Sathler, "Análise do Custo de Ciclo de Vida de Parques Eólicos Análise do Custo de Ciclo de Vida de Parques Eólicos," CEFET/MG, 2013.
- [7] M. SAKURAI, Gerenciamento Integrado de Custos. São Paulo: Atlas, 1997.
- [8] D. McInnis and M. Capezzali, "Managing wind turbine generators with a profit maximized approach," *Sustain.*, vol. 12, no. 17, 2020, doi: 10.3390/su12177139.
- [9] M. M. Luengo and A. Kolios, "Failure mode identification and end of life scenarios of offshore wind turbines: A review," *Energies*, vol. 8, no. 8, pp. 8339–8354, 2015, doi: 10.3390/en8088339.
- [10] A. Ioannou, A. Angus, and F. Brennan, "A lifecycle techno-economic model

of offshore wind energy for different entry and exit instances," *Appl. Energy*, vol. 221, no. April, pp. 406–424, 2018, doi: 10.1016/j.apenergy.2018.03.143.

- [11] A. Ioannou, A. Angus, and F. Brennan, "Stochastic financial appraisal of offshore wind farms," *Renew. Energy*, vol. 145, pp. 1176–1191, 2020, doi: 10.1016/j.renene.2019.06.111.
- [12] G. Rajgor, "O&M under control?," *Renew. Energy Focus*, vol. 13, no. 2, pp. 42–46, 2012, doi: 10.1016/S1755-0084(12)70040-6.
- [13] IRENA, "RENEWABLE ENERGY TECHNOLOGIES: COST ANALYSIS SERIES," 2012. [Online]. Available: https://www.irena.org/publications/2012/Jun/Renewable-Energy-Cost-Analysis---Wind-Power.
- [14] R. Gasch et al., Windkraftanlagen. 2005.
- [15] C. a Walford, "Wind turbine reliability: understanding and minimizing wind turbine operation and maintenance costs," *Energy*, no. March, pp. SAND2006-1100, 2006, [Online]. Available: http://prod.sandia.gov/techlib/access-control.cgi/2006/061100.pdf.
- [16] S. D. Hamilton, D. Millstein, M. Bolinger, R. Wiser, and S. Jeong, "How Does Wind Project Performance Change with Age in the United States?," *Joule*, vol. 4, no. 5, pp. 1004–1020, 2020, doi: 10.1016/j.joule.2020.04.005.
- [17] S. Germer and A. Kleidon, "Have wind turbines in Germany generated electricity as would be expected from the prevailing wind conditions in 2000-2014?," *PLoS One*, vol. 14, no. 2, pp. 1–16, 2019, doi: 10.1371/journal.pone.0211028.
- [18] I. Staffell and R. Green, "How does wind farm performance decline with age?," *Renew. Energy*, vol. 66, pp. 775–786, 2014, doi: 10.1016/j.renene.2013.10.041.
- [19] J. Jorgensen, T. Mai, and G. Brinkman, "Reducing wind curtailment through

transmission expansion in a Wind Vision future," no. January, p. 38, 2017, [Online]. Available: https://www.nrel.gov/docs/fy17osti/67240.pdf.

- [20] Y. Qi, W. Dong, C. Dong, and C. Huang, "Fixing Wind Curtailment with Electric Power System Reform in China," 2018, [Online]. Available: https://www.brookings.edu/wp-content/uploads/2018/04/fixing-windcurtailment-with-electric-power-system-reform-in-china.pdf.
- [21] EirGrid, "Annual Renewable Energy Constraint and Curtailment Report 2020," no. May, pp. 1–28, 2021, [Online]. Available: http://www.eirgridgroup.com/site-files/library/EirGrid/Annual-Renewable-Constraint-and-Curtailment-Report-2020.pdf.
- [22] M. N. Scheu, A. Kolios, T. Fischer, and F. Brennan, "Influence of statistical uncertainty of component reliability estimations on offshore wind farm availability," *Reliab. Eng. Syst. Saf.*, vol. 168, 2017, doi: 10.1016/j.ress.2017.05.021.
- [23] D. H. Stamatis, *The OEE primer: Understanding overall equipment effectiveness, reliability, and maintainability.* 2017.
- [24] K. Fischer, F. Besnard, and L. Bertling, "Reliability-centered maintenance for wind turbines based on statistical analysis and practical experience," *IEEE Trans. Energy Convers.*, vol. 27, no. 1, pp. 184–195, 2012, doi: 10.1109/TEC.2011.2176129.
- [25] J. Kang, L. Sun, and C. Guedes Soares, "Fault Tree Analysis of floating offshore wind turbines," *Renew. Energy*, vol. 133, pp. 1455–1467, 2019, doi: 10.1016/j.renene.2018.08.097.
- [26] M. N. Scheu, L. Tremps, U. Smolka, A. Kolios, and F. Brennan, "A systematic Failure Mode Effects and Criticality Analysis for offshore wind turbine systems towards integrated condition based maintenance strategies," *Ocean Eng.*, vol. 176, no. October 2018, pp. 118–133, 2019, doi: 10.1016/j.oceaneng.2019.02.048.
- [27] H. Li, A. P. Teixeira, and C. Guedes Soares, "A two-stage Failure Mode

and Effect Analysis of offshore wind turbines," *Renew. Energy*, vol. 162, pp. 1438–1461, 2020, doi: 10.1016/j.renene.2020.08.001.

- [28] S. Faulstich, B. Hanh, and P. J. Tavner, "Wind turbine downtime and its importance for offshore deployment," *Wind Energ.*, vol. 14, pp. 327–337, 2011.
- [29] P. J. Tavner, D. M. Greenwood, M. W. G. Whittle, R. Gindele, S. Faulstich, and B. Hanh, "Study of weather and location effects on wind turbine failure rates," *Wind Energ.*, vol. 16, pp. 175–187, 2013.
- [30] M. Leimeister and A. Kolios, "A review of reliability-based methods for risk analysis and their application in the offshore wind industry," *Renew. Sustain. Energy Rev.*, vol. 91, no. April, pp. 1065–1076, 2018, doi: 10.1016/j.rser.2018.04.004.
- [31] R. Zheng, Y. Zhou, and Y. Zhang, "Optimal preventive maintenance for wind turbines considering the effects of wind speed," *Wind Energy*, vol. 23, no. 11, pp. 1987–2003, 2020, doi: 10.1002/we.2541.
- [32] X. Li *et al.*, "A decision support system for strategic maintenance planning in offshore wind farms," *Renew. Energy*, vol. 99, pp. 784–799, 2016, doi: 10.1016/j.renene.2016.07.037.
- [33] N. Y. Yürüşen, P. N. Rowley, S. J. Watson, and J. J. Melero, "Automated wind turbine maintenance scheduling," *Reliab. Eng. Syst. Saf.*, vol. 200, no. March, 2020, doi: 10.1016/j.ress.2020.106965.
- [34] X. Yin, X. Zhao, J. Lin, and A. Karcanias, "Reliability aware multi-objective predictive control for wind farm based on machine learning and heuristic optimizations," *Energy*, vol. 202, p. 117739, 2020, doi: 10.1016/j.energy.2020.117739.
- [35] L. Duchesne, E. Karangelos, A. Sutera, and L. Wehenkel, "Machine learning for ranking day-ahead decisions in the context of short-term operation planning," *Electr. Power Syst. Res.*, vol. 189, no. April, p. 106548, 2020, doi: 10.1016/j.epsr.2020.106548.

- [36] B. Hahn and H. Jung, "Improving Wind Turbines Availability by Reliability Based Maintenance.," *DEWEK*, 2006, [Online]. Available: https://www.researchgate.net/profile/Berthold-Hahn/publication/229014918_Improving_wind_turbine_availability_by_reli ability_based_maintenance/links/5c4ab2ec299bf12be3e1aa4c/Improvingwind-turbine-availability-by-reliability-based-maintenance.pdf.
- [37] A. Ioannou, A. Angus, and F. Brennan, "Parametric CAPEX, OPEX, and LCOE expressions for offshore wind farms based on global deployment parameters," *Energy Sources, Part B Econ. Plan. Policy*, vol. 13, no. 5, pp. 281–290, 2018, doi: 10.1080/15567249.2018.1461150.
- [38] W. Yang, "Condition monitoring of offshore wind turbines," in *Offshore Wind Farms: Technologies, Design and Operation*, Elsevier Ltd, 2016, pp. 543–572.
- [39] M. Shafiee, "Maintenance logistics organization for offshore wind energy: Current progress and future perspectives," *Renew. Energy*, vol. 77, no. 1, pp. 182–193, 2015, doi: 10.1016/j.renene.2014.11.045.
- [40] M. I. H. Tusar and B. R. Sarker, "Maintenance cost minimization models for offshore wind farms: A systematic and critical review," *Int. J. Energy Res.*, no. October, pp. 1–27, 2021, doi: 10.1002/er.7425.
- [41] H. Li, C. G. Huang, and C. Guedes Soares, "A real-time inspection and opportunistic maintenance strategies for floating offshore wind turbines," *Ocean Eng.*, vol. 256, no. April, 2022, doi: 10.1016/j.oceaneng.2022.111433.
- [42] U. Bhardwaj, A. P. Teixeira, and C. G. Soares, "Reliability prediction of an offshore wind turbine gearbox," *Renew. Energy*, vol. 141, pp. 693–706, 2019, doi: 10.1016/j.renene.2019.03.136.
- [43] T. N. Santelo, C. M. R. de Oliveira, C. D. Maciel, and J. R. B. José Roberto,
 "Wind Turbine Failures Review and Trends," *J. Control. Autom. Electr. Syst.*, 2021, doi: 10.1007/s40313-021-00789-8.

- [44] G. Helbing and M. Ritter, "Deep Learning for fault detection in wind turbines," *Renew. Sustain. Energy Rev.*, vol. 98, no. January, pp. 189–198, 2018, doi: 10.1016/j.rser.2018.09.012.
- [45] L. Chen, G. Xu, Q. Zhang, and X. Zhang, "Learning deep representation of imbalanced SCADA data for fault detection of wind turbines," *Meas. J. Int. Meas. Confed.*, vol. 139, pp. 370–379, 2019, doi: 10.1016/j.measurement.2019.03.029.
- [46] Z. Lin, X. Liu, and M. Collu, "Wind power prediction based on highfrequency SCADA data along with isolation forest and deep learning neural networks," *Int. J. Electr. Power Energy Syst.*, vol. 118, no. November 2019, p. 105835, 2020, doi: 10.1016/j.ijepes.2020.105835.
- [47] Y. Zhang, H. Zheng, J. Liu, J. Zhao, and P. Sun, "An anomaly identification model for wind turbine state parameters," *J. Clean. Prod.*, vol. 195, pp. 1214–1227, Sep. 2018, doi: 10.1016/j.jclepro.2018.05.126.
- [48] A. Koltsidopoulos Papatzimos, P. R. Thies, and T. Dawood, "Offshore wind turbine fault alarm prediction," *Wind Energy*, vol. 22, no. 12, pp. 1779– 1788, 2019, doi: 10.1002/we.2402.
- [49] N. Martin, C. Mailhes, and X. Laval, "Automated Machine Health Monitoring at an Expert Level," *Acoust. Aust.*, vol. 49, no. 2, pp. 185–197, 2021, doi: 10.1007/s40857-021-00227-4.
- [50] T. Han, L. Ding, D. Qi, C. Li, Z. Fu, and W. Chen, "Compound faults diagnosis method for wind turbine mainshaft bearing with Teager and second-order stochastic resonance," *Meas. J. Int. Meas. Confed.*, vol. 202, no. September, p. 111931, 2022, doi: 10.1016/j.measurement.2022.111931.
- [51] J. McMorland *et al.*, "A review of operations and maintenance modelling with considerations for novel wind turbine concepts," *Renew. Sustain. Energy Rev.*, vol. 165, no. April, p. 112581, 2022, doi: 10.1016/j.rser.2022.112581.

- [52] A. Ioannou, A. Angus, and F. Brennan, "Informing parametric risk control policies for operational uncertainties of offshore wind energy assets," *Ocean Eng.*, vol. 177, no. February, pp. 1–11, 2019, doi: 10.1016/j.oceaneng.2019.02.058.
- [53] L. Yang, R. Peng, G. Li, and C. G. Lee, "Operations management of wind farms integrating multiple impacts of wind conditions and resource constraints," *Energy Convers. Manag.*, vol. 205, no. May 2019, p. 112162, 2020, doi: 10.1016/j.enconman.2019.112162.
- [54] A. Mentes and O. Turan, "A new resilient risk management model for Offshore Wind Turbine maintenance," Saf. Sci., vol. 119, no. June 2018, pp. 360–374, 2019, doi: 10.1016/j.ssci.2018.06.022.
- [55] O. Gözcü and M. Stolpe, "Representation of wind turbine blade responses in power production load cases by linear mode shapes," *Wind Energy*, vol. 23, no. 5, pp. 1317–1330, 2020, doi: 10.1002/we.2488.
- [56] G. M. Stewart and M. A. Lackner, "The impact of passive tuned mass dampers and wind-wave misalignment on offshore wind turbine loads," *Eng. Struct.*, vol. 73, pp. 54–61, 2014, doi: 10.1016/j.engstruct.2014.04.045.
- [57] J. T. Horn, J. R. Krokstad, and B. J. Leira, "Impact of model uncertainties on the fatigue reliability of offshore wind turbines," *Mar. Struct.*, vol. 64, no. June 2018, pp. 174–185, 2019, doi: 10.1016/j.marstruc.2018.11.004.
- [58] M. Reder, N. Y. Yürüşen, and J. J. Melero, "Data-driven learning framework for associating weather conditions and wind turbine failures," *Reliab. Eng. Syst. Saf.*, vol. 169, no. January 2017, pp. 554–569, 2018, doi: 10.1016/j.ress.2017.10.004.
- [59] S. Fæster, N. F. J. Johansen, L. Mishnaevsky, Y. Kusano, J. I. Bech, and M. B. Madsen, "Rain erosion of wind turbine blades and the effect of air bubbles in the coatings," *Wind Energy*, vol. 24, no. 10, pp. 1071–1082, 2021, doi: 10.1002/we.2617.

- [60] J. Gao, B. Sweetman, and S. Tang, "Multiaxial fatigue assessment of floating offshore wind turbine blades operating on compliant floating platforms," *Ocean Eng.*, vol. 261, no. August, p. 111921, 2022, doi: 10.1016/j.oceaneng.2022.111921.
- [61] T. Zheng and N. Z. Chen, "Time-domain fatigue assessment for blade root bolts of floating offshore wind turbine (FOWT)," *Ocean Eng.*, vol. 262, no. August, p. 112201, 2022, doi: 10.1016/j.oceaneng.2022.112201.
- [62] T. A. T. Nguyen, S. Y. Chou, and T. H. K. Yu, "Developing an exhaustive optimal maintenance schedule for offshore wind turbines based on riskassessment, technical factors and cost-effective evaluation," *Energy*, vol. 249, p. 123613, 2022, doi: 10.1016/j.energy.2022.123613.
- [63] T. A. T. Nguyen and S.-Y. Chou, "Maintenance strategy selection for improving cost-effectiveness of offshore wind systems," *Energy Convers. Manag.*, vol. 157, pp. 86–95, 2018.
- [64] S. Zhong, A. A. Pantelous, M. Goh, and J. Zhou, "A reliability-and-costbased fuzzy approach to optimize preventive maintenance scheduling for offshore wind farms," *Mech. Syst. Signal Process.*, vol. 124, pp. 643–663, 2019, doi: 10.1016/j.ymssp.2019.02.012.
- [65] T. A. T. Nguyen and S. Y. Chou, "Improved maintenance optimization of offshore wind systems considering effects of government subsidies, lost production and discounted cost model," *Energy*, vol. 187, p. 115909, 2019, doi: 10.1016/j.energy.2019.115909.
- [66] A. Sa'ad, A. C. Nyoungue, and Z. Hajej, "An integrated maintenance and power generation forecast by ANN approach based on availability maximization of a wind farm," *Energy Reports*, vol. 8, no. May, pp. 282– 301, 2022, doi: 10.1016/j.egyr.2022.06.120.
- [67] R. O'Neil, A. Khatab, C. Diallo, and U. Venkatadri, "Optimal joint maintenance and orienteering strategy for complex mission-oriented systems: A case study in offshore wind energy," *Comput. Oper. Res.*, vol.

149, no. December 2021, p. 106020, 2023, doi: 10.1016/j.cor.2022.106020.

- [68] P. T. Baboli, A. Raeiszadeh, D. Babazadeh, and J. Meiners, "Two-Stage Condition-based Maintenance Model of Wind Turbine: From Diagnosis to Prognosis," 2020 IEEE Int. Smart Cities Conf. ISC2 2020, vol. 423, 2020, doi: 10.1109/ISC251055.2020.9239029.
- [69] J. Izquierdo, A. C. Márquez, J. Uribetxebarria, and A. Erguido, "On the importance of assessing the operational context impact on maintenance management for life cycle cost of wind energy projects," *Renew. Energy*, vol. 153, pp. 1100–1110, 2020, doi: 10.1016/j.renene.2020.02.048.
- [70] M. Asgarpour, Assembly, transportation, installation and commissioning of offshore wind farms. Elsevier Ltd, 2016.
- [71] C. Yan, Y. Pan, and C. L. Archer, "A general method to estimate wind farm power using artificial neural networks," *Wind Energy*, vol. 22, no. 11, pp. 1421–1432, 2019, doi: 10.1002/we.2379.
- [72] R. K. Pandit, D. Infield, and A. Kolios, "Comparison of advanced nonparametric models for wind turbine power curves," *IET Renew. Power Gener.*, vol. 13, no. 9, pp. 1503–1510, Jul. 2019, doi: 10.1049/ietrpg.2018.5728.
- [73] R. K. Pandit, D. Infield, and A. Kolios, "Gaussian process power curve models incorporating wind turbine operational variables," *Energy Reports*, vol. 6, pp. 1658–1669, 2020, doi: 10.1016/j.egyr.2020.06.018.
- [74] Y. M. Saint-Drenan *et al.*, "A parametric model for wind turbine power curves incorporating environmental conditions," *Renew. Energy*, vol. 157, pp. 754–768, 2020, doi: 10.1016/j.renene.2020.04.123.
- [75] R. K. Pandit, D. Infield, and J. Carroll, "Incorporating air density into a Gaussian process wind turbine power curve model for improving fitting accuracy," *Wind Energy*, vol. 22, no. 2, pp. 302–315, 2019, doi: 10.1002/we.2285.

- [76] S. Kanev, "Dynamic wake steering and its impact on wind farm power production and yaw actuator duty," *Renew. Energy*, vol. 146, pp. 9–15, 2020, doi: 10.1016/j.renene.2019.06.122.
- [77] M. T. van Dijk, J. W. van Wingerden, T. Ashuri, and Y. Li, "Wind farm multiobjective wake redirection for optimizing power production and loads," *Energy*, vol. 121, pp. 561–569, 2017, doi: 10.1016/j.energy.2017.01.051.
- [78] J. Park and K. H. Law, "A data-driven, cooperative wind farm control to maximize the total power production," *Appl. Energy*, vol. 165, pp. 151–165, 2016, doi: 10.1016/j.apenergy.2015.11.064.
- U. Ciri, M. A. Rotea, and S. Leonardi, "Model-free control of wind farms: A comparative study between individual and coordinated extremum seeking," *Renew. Energy*, vol. 113, pp. 1033–1045, 2017, doi: 10.1016/j.renene.2017.06.065.
- [80] L. Tian, Y. Song, N. Zhao, W. Shen, T. Wang, and C. Zhu, "Numerical investigations into the idealized diurnal cycle of atmospheric boundary layer and its impact on wind turbine's power performance," *Renew. Energy*, vol. 145, pp. 419–427, 2020, doi: 10.1016/j.renene.2019.05.038.
- [81] S. H. Danook, K. J. Jassim, and A. M. Hussein, "The impact of humidity on performance of wind turbine," *Case Stud. Therm. Eng.*, vol. 14, no. April, p. 100456, 2019, doi: 10.1016/j.csite.2019.100456.
- [82] L. M. Bardal and L. R. Sætran, "Influence of turbulence intensity on wind turbine power curves," *Energy Procedia*, vol. 137, pp. 553–558, 2017, doi: 10.1016/j.egypro.2017.10.384.
- [83] H. Arastoopour and A. Cohan, "CFD simulation of the effect of rain on the performance of horizontal wind turbines," *AIChE J.*, vol. 63, no. 12, pp. 5375–5383, 2017, doi: 10.1002/aic.15928.
- [84] O. Yirtici, S. Ozgen, and I. H. Tuncer, "Predictions of ice formations on wind turbine blades and power production losses due to icing," *Wind Energy*, vol. 22, no. 7, pp. 945–958, 2019, doi: 10.1002/we.2333.

- [85] D. B. Stoyanov and J. D. Nixon, "Alternative operational strategies for wind turbines in cold climates," *Renew. Energy*, vol. 145, pp. 2694–2706, 2020, doi: 10.1016/j.renene.2019.08.023.
- [86] X. Dong, D. Gao, J. Li, Z. Jincao, and K. Zheng, "Blades icing identification model of wind turbines based on SCADA data," *Renew. Energy*, vol. 162, pp. 575–586, 2020, doi: 10.1016/j.renene.2020.07.049.
- [87] S. Karimian Aliabadi and S. Rasekh, "Effect of platform disturbance on the performance of offshore wind turbine under pitch control," *Wind Energy*, vol. 23, no. 5, pp. 1210–1230, 2020, doi: 10.1002/we.2482.
- [88] Y. Fang, L. Duan, Z. Han, Y. Zhao, and H. Yang, "Numerical analysis of aerodynamic performance of a floating offshore wind turbine under pitch motion," *Energy*, vol. 192, p. 116621, 2020, doi: 10.1016/j.energy.2019.116621.
- [89] X. Li, C. Zhu, Z. Fan, X. Chen, and J. Tan, "Effects of the yaw error and the wind-wave misalignment on the dynamic characteristics of the floating offshore wind turbine," *Ocean Eng.*, vol. 199, no. February 2019, p. 106960, 2020, doi: 10.1016/j.oceaneng.2020.106960.
- [90] B. Wen, X. Dong, X. Tian, Z. Peng, W. Zhang, and K. Wei, "The power performance of an offshore floating wind turbine in platform pitching motion," *Energy*, vol. 154, pp. 508–521, 2018, doi: 10.1016/j.energy.2018.04.140.
- [91] H. Namik and K. Stol, "Individual blade pitch control of floating offshore wind turbines," *Wind Energy*, vol. 13, no. 1, pp. 74–85, 2010, doi: 10.1002/we.332.
- [92] Y. Chang, J. Chen, C. Qu, and T. Pan, "Intelligent fault diagnosis of Wind Turbines via a Deep Learning Network Using Parallel Convolution Layers with Multi-Scale Kernels," *Renew. Energy*, vol. 153, pp. 205–213, 2020, doi: 10.1016/j.renene.2020.02.004.
- [93] A. K. Wright and D. H. Wood, "The starting and low wind speed behaviour

of a small horizontal axis wind turbine," *J. Wind Eng. Ind. Aerodyn.*, vol. 92, no. 14–15, pp. 1265–1279, 2004, doi: 10.1016/j.jweia.2004.08.003.

- [94] Z. Fan and C. Zhu, "The optimization and the application for the wind turbine power-wind speed curve," *Renew. Energy*, vol. 140, pp. 52–61, 2019, doi: 10.1016/j.renene.2019.03.051.
- [95] J. Dai, X. Yang, W. Hu, L. Wen, and Y. Tan, "Effect investigation of yaw on wind turbine performance based on SCADA data," *Energy*, vol. 149, pp. 684–696, 2018, doi: 10.1016/j.energy.2018.02.059.
- [96] K. A. Kragh and M. H. Hansen, "Potential of power gain with improved yaw alignment," *Wind ENERGY*, vol. 18, pp. 979–989, 2015, [Online]. Available: 10.1002/we.1739.
- [97] J. Dai, T. He, M. Li, and X. Long, "Performance study of multi-source driving yaw system for aiding yaw control of wind turbines," *Renew. Energy*, vol. 163, pp. 154–171, 2021, doi: 10.1016/j.renene.2020.08.065.
- [98] C. Kress, N. Chokani, and R. S. Abhari, "Downwind wind turbine yaw stability and performance," *Renew. Energy*, vol. 83, pp. 1157–1165, 2015, doi: 10.1016/j.renene.2015.05.040.
- [99] M. Yesilbudak, S. Sagiroglu, and I. Colak, "A novel intelligent approach for yaw position forecasting in wind energy systems," *Int. J. Electr. Power Energy Syst.*, vol. 69, pp. 406–413, 2015, doi: 10.1016/j.ijepes.2015.01.030.
- [100] Z. Jiang, M. Karimirad, and T. Moan2, "Dynamic response analysis of wind turbines under blade pitch system fault, grid loss, and shutdown events," *Wind ENERGY*, vol. 17, pp. 1385–1409, 2014, doi: 10.1002/we.1639.
- [101] O. Zamzoum, A. Derouich, S. Motahhir, Y. El Mourabit, and A. El Ghzizal, "Performance analysis of a robust adaptive fuzzy logic controller for wind turbine power limitation," *J. Clean. Prod.*, vol. 265, p. 121659, 2020, doi: 10.1016/j.jclepro.2020.121659.

- [102] E. Mohammadi, R. Fadaeinedjad, and H. R. Naji, "Flicker emission, voltage fluctuations, and mechanical loads for small-scale stall- and yaw-controlled wind turbines," *Energy Convers. Manag.*, vol. 165, no. April, pp. 567–577, 2018, doi: 10.1016/j.enconman.2018.03.094.
- [103] M. Mirzaei, C. Tibaldi, and M. H. Hansen, "PI controller design of a wind turbine: Evaluation of the pole-placement method and tuning using constrained optimization," *J. Phys. Conf. Ser.*, vol. 753, no. 5, 2016, doi: 10.1088/1742-6596/753/5/052026.
- [104] A. Dahbi, N. Nait-Said, and M. S. Nait-Said, "A novel combined MPPT-pitch angle control for wide range variable speed wind turbine based on neural network," *Int. J. Hydrogen Energy*, vol. 41, no. 22, pp. 9427–9442, 2016, doi: 10.1016/j.ijhydene.2016.03.105.
- [105] B. Dou, T. Qu, L. Lei, and P. Zeng, "Optimization of wind turbine yaw angles in a wind farm using a three-dimensional yawed wake model," *Energy*, vol. 209, p. 118415, 2020, doi: 10.1016/j.energy.2020.118415.
- [106] J. Lee, E. Son, B. Hwang, and S. Lee, "Blade pitch angle control for aerodynamic performance optimization of a wind farm," *Renew. Energy*, vol. 54, pp. 124–130, 2013, doi: 10.1016/j.renene.2012.08.048.
- [107] C. L. Archer, H. P. Simão, W. Kempton, W. B. Powell, and M. J. Dvorak, "The challenge of integrating offshore wind power in the U.S. electric grid. Part I: Wind forecast error," *Renew. Energy*, vol. 103, pp. 346–360, 2017, doi: 10.1016/j.renene.2016.11.047.
- [108] M. Yu *et al.*, "Superposition Graph Neural Network for offshore wind power prediction," *Futur. Gener. Comput. Syst.*, vol. 113, pp. 145–157, 2020, doi: 10.1016/j.future.2020.06.024.
- [109] L. T. Paiva, C. Veiga Rodrigues, and J. M. L. M. Palma, "Determining wind turbine power curves based on operating conditions," *Wind Energy*, vol. 17, no. 10, pp. 1563–1575, Oct. 2014, doi: 10.1002/we.1651.
- [110] K. P. B. Sathler, A. Kolios, S. Al-Sanad, and J. Parol, "Application of the

Overall Equipment Effectiveness Concept in Wind Energy Assets," 2020, doi: 10.3850/978-981-14-8593-0.

- [111] Z. Shen and M. Ritter, "Forecasting volatility of wind power production," *Appl. Energy*, vol. 176, pp. 295–308, 2016, doi: 10.1016/j.apenergy.2016.05.071.
- [112] K. P. B. Sathler and A. Kolios, "The Use of Machine Learning and Performance Concept to Monitor and Predict Wind Power Output," Int. Conf. Electr. Comput. Energy Technol. ICECET 2022, no. June, pp. 20– 22, 2022, doi: 10.1109/ICECET55527.2022.9873076.
- [113] H. P. Simão, W. B. Powell, C. L. Archer, and W. Kempton, "The challenge of integrating offshore wind power in the U.S. electric grid. Part II: Simulation of electricity market operations," *Renew. Energy*, vol. 103, pp. 418–431, 2017, doi: 10.1016/j.renene.2016.11.049.
- [114] P. Argyle and S. J. Watson, "Offshore Turbine Wake Power Losses: Is Turbine Separation Significant?," *Energy Procedia*, vol. 137, pp. 134–142, Oct. 2017, doi: 10.1016/j.egypro.2017.10.340.
- [115] J. Dai, W. Yang, J. Cao, D. Liu, and X. Long, "Ageing assessment of a wind turbine over time by interpreting wind farm SCADA data," *Renew. Energy*, vol. 116, pp. 199–208, Feb. 2018, doi: 10.1016/j.renene.2017.03.097.
- [116] Y. Liu and L. Zhang, "Data-driven fault identification of ageing wind turbine*," 2022 13th UKACC Int. Conf. Control. Control 2022, pp. 183–188, 2022, doi: 10.1109/Control55989.2022.9781452.
- [117] X. Chen, "Fracture of wind turbine blades in operation—Part I: A comprehensive forensic investigation," *Wind Energy*, vol. 21, no. 11, pp. 1046–1063, 2018, doi: 10.1002/we.2212.
- [118] A. Sareen, C. A. Sapre, and M. S. Selig, "Effects of leading edge erosion on wind turbine blade performance," *Wind Energy*, vol. 17, no. 10, pp. 1531–1542, Oct. 2014, doi: 10.1002/we.1649.

- [119] S. Scher and J. Molinder, "Machine learning-based prediction of icingrelated wind power production loss," *IEEE Access*, vol. 7, pp. 129421– 129429, 2019, doi: 10.1109/ACCESS.2019.2939657.
- [120] L. Swenson, L. Gao, J. Hong, and L. Shen, "An efficacious model for predicting icing-induced energy loss for wind turbines," *Appl. Energy*, vol. 305, no. April 2021, p. 117809, 2022, doi: 10.1016/j.apenergy.2021.117809.
- [121] A. C. Kheirabadi and R. Nagamune, "A quantitative review of wind farm control with the objective of wind farm power maximization," *J. Wind Eng. Ind. Aerodyn.*, vol. 192, no. May, pp. 45–73, 2019, doi: 10.1016/j.jweia.2019.06.015.
- [122] S. El-Asha, L. Zhan, and G. V. lungo, "Quantification of power losses due to wind turbine wake interactions through SCADA, meteorological and wind LiDAR data," *Wind Energy*, vol. 20, no. 11, pp. 1823–1839, 2017, doi: 10.1002/we.2123.
- [123] S. C. Pryor, R. J. Barthelmie, and T. J. Shepherd, "Wind power production from very large offshore wind farms," *Joule*, vol. 5, no. 10, pp. 2663–2686, 2021, doi: 10.1016/j.joule.2021.09.002.
- [124] C. Choe Wei Chang, T. Jian Ding, T. Jian Ping, M. Ariannejad, K. Chia Chao, and S. B. Samdin, "Fault detection and anti-icing technologies in wind energy conversion systems: A review," *Energy Reports*, vol. 8, pp. 28–33, 2022, doi: 10.1016/j.egyr.2022.10.234.
- [125] J. A. Frederik, B. M. Doekemeijer, S. P. Mulders, and J. W. van Wingerden,
 "The helix approach: Using dynamic individual pitch control to enhance wake mixing in wind farms," *Wind Energy*, vol. 23, no. 8, pp. 1739–1751, 2020, doi: 10.1002/we.2513.
- [126] P. A. Fleming *et al.*, "Evaluating techniques for redirecting turbine wakes using SOWFA," *Renew. Energy*, vol. 70, pp. 211–218, 2014, doi: 10.1016/j.renene.2014.02.015.

- [127] M. F. Howland, S. K. Lele, and J. O. Dabiri, "Wind farm power optimization through wake steering," *Proc. Natl. Acad. Sci. U. S. A.*, vol. 116, no. 29, pp. 14495–14500, 2019, doi: 10.1073/pnas.1903680116.
- [128] T. Shu, D. Song, and Y. Hoon Joo, "Decentralised optimisation for large offshore wind farms using a sparsified wake directed graph," *Appl. Energy*, vol. 306, no. PA, p. 117986, 2022, doi: 10.1016/j.apenergy.2021.117986.
- [129] D. Astolfi, F. Castellani, A. Garinei, and L. Terzi, "Data mining techniques for performance analysis of onshore wind farms," *Appl. Energy*, vol. 148, pp. 220–233, Jun. 2015, doi: 10.1016/j.apenergy.2015.03.075.
- [130] S. Pieralli, M. Ritter, and M. Odening, "Efficiency of wind power production and its determinants," *Energy*, vol. 90, pp. 429–438, 2015, doi: 10.1016/j.energy.2015.07.055.
- [131] N. Karakasis, E. Tsioumas, N. Jabbour, A. M. Bazzi, and C. Mademlis,
 "Optimal Efficiency Control in a Wind System with Doubly Fed Induction Generator," *IEEE Trans. Power Electron.*, vol. 34, no. 1, pp. 356–368, 2018, doi: 10.1109/TPEL.2018.2823481.
- [132] H. Liao *et al.*, "Active power dispatch optimization for offshore wind farms considering fatigue distribution," *Renew. Energy*, vol. 151, pp. 1173–1185, 2020, doi: 10.1016/j.renene.2019.11.132.
- [133] M. Al-Khayat *et al.*, "Performance analysis of a 10-MW wind farm in a hot and dusty desert environment. Part 2: Combined dust and hightemperature effects on the operation of wind turbines," *Sustain. Energy Technol. Assessments*, vol. 47, no. February, 2021, doi: 10.1016/j.seta.2021.101461.
- [134] M. Shahbazi, P. Poure, and S. Saadate, "Real-time power switch fault diagnosis and fault-tolerant operation in a DFIG-based wind energy system," *Renew. Energy*, vol. 116, pp. 209–218, Feb. 2018, doi: 10.1016/j.renene.2017.02.066.
- [135] B. Rona and Ö. Güler, "Power system integration of wind farms and

analysis of grid code requirements," *Renew. Sustain. Energy Rev.*, vol. 49, pp. 100–107, 2015, doi: 10.1016/j.rser.2015.04.085.

- [136] O. N. Nobela, R. C. Bansal, and J. J. Justo, "A review of power quality compatibility of wind energy conversion systems with the South African utility grid," *Renew. Energy Focus*, vol. 31, no. December, pp. 63–72, 2019, doi: 10.1016/j.ref.2019.10.001.
- [137] J. Đaković, M. Krpan, P. Ilak, T. Baškarad, and I. Kuzle, "Impact of wind capacity share, allocation of inertia and grid configuration on transient RoCoF: The case of the Croatian power system," *Int. J. Electr. Power Energy Syst.*, vol. 121, no. March, p. 106075, 2020, doi: 10.1016/j.ijepes.2020.106075.
- [138] E. Sáiz-Marín, E. Lobato, I. Egido, and L. Rouco, "Economic assessment of voltage and reactive power control provision by wind farms," *Wind Energy*, vol. 18, no. 5, pp. 851–864, May 2015, doi: 10.1002/we.1734.
- [139] R. Li, L. Yu, and L. Xu, "Operation of offshore wind farms connected with DRU-HVDC transmission systems with special consideration of faults," *Glob. Energy Interconnect.*, vol. 1, no. 5, pp. 608–617, 2018, doi: 10.14171/j.2096-5117.gei.2018.05.010.
- [140] I. Sowa, J. L. Domínguez-García, and O. Gomis-Bellmunt, "Impedancebased analysis of harmonic resonances in HVDC connected offshore wind power plants," *Electr. Power Syst. Res.*, vol. 166, no. April 2018, pp. 61– 72, 2019, doi: 10.1016/j.epsr.2018.10.003.
- [141] I. D. Margaris, A. D. Hansen, N. A. Cutululis, P. Sørensen, and N. D. Hatziargyriou, "Impact of wind power in autonomous power systems-power fluctuations-modelling and control issues," *Wind Energy*, vol. 14, no. 1, pp. 133–153, Jan. 2011, doi: 10.1002/we.417.
- [142] S. Prasad, S. Purwar, and N. Kishor, "Non-linear sliding mode control for frequency regulation with variable-speed wind turbine systems," Int. J. Electr. Power Energy Syst., vol. 107, no. June 2018, pp. 19–33, 2019, doi:

10.1016/j.ijepes.2018.11.005.

- [143] S. Martín-Martínez, E. Gómez-Lázaro, A. Vigueras-Rodríguez, J. Alvaro Fuentes-Moreno, and A. Molina-García, "Analysis of positive ramp limitation control strategies for reducing wind power fluctuations," *IET Renew. Power Gener.*, vol. 7, no. 6, pp. 593–602, 2013, doi: 10.1049/ietrpg.2012.0188.
- [144] M. Beza and M. Bongiorno, "Identification of resonance interactions in offshore-wind farms connected to the main grid by MMC-based HVDC system," *Int. J. Electr. Power Energy Syst.*, vol. 111, no. April, pp. 101–113, 2019, doi: 10.1016/j.ijepes.2019.04.004.
- [145] K. Luo, W. Shi, and W. Wang, "Extreme scenario extraction of a grid with large scale wind power integration by combined entropy-weighted clustering method," *Glob. Energy Interconnect.*, vol. 3, no. 2, pp. 140–148, 2020, doi: 10.1016/j.gloei.2020.05.006.
- [146] E. V. Mc Garrigle, J. P. Deane, and P. G. Leahy, "How much wind energy will be curtailed on the 2020 Irish power system?," *Renew. Energy*, vol. 55, no. 2013, pp. 544–553, 2013, doi: 10.1016/j.renene.2013.01.013.
- [147] X. Wang, L. Li, A. Palazoglu, N. H. El-Farra, and N. Shah, "Optimization and control of offshore wind systems with energy storage," *Energy Convers. Manag.*, vol. 173, no. April, pp. 426–437, 2018, doi: 10.1016/j.enconman.2018.07.079.
- [148] O. Probst, "A new strategy for short-term ramp rate control in wind farms," *Int. J. Electr. Power Energy Syst.*, vol. 120, no. March, p. 105969, 2020, doi: 10.1016/j.ijepes.2020.105969.
- [149] E. Nycander, L. Söder, J. Olauson, and R. Eriksson, "Curtailment analysis for the Nordic power system considering transmission capacity, inertia limits and generation flexibility," *Renew. Energy*, vol. 152, pp. 942–960, 2020, doi: 10.1016/j.renene.2020.01.059.
- [150] R. Davison-Kernan, X. Liu, S. McLoone, and B. Fox, "Quantification of wind

curtailment on a medium-sized power system and mitigation using municipal water pumping load," *Renew. Sustain. Energy Rev.*, vol. 112, no. June 2019, pp. 499–507, 2019, doi: 10.1016/j.rser.2019.06.004.

- [151] R. Jin, P. Hou, G. Yang, Y. Qi, C. Chen, and Z. Chen, "Cable routing optimization for offshore wind power plants via wind scenarios considering power loss cost model," *Appl. Energy*, vol. 254, no. August, p. 113719, 2019, doi: 10.1016/j.apenergy.2019.113719.
- [152] J. A. Pérez-Rúa, K. Das, and N. A. Cutululis, "Optimum sizing of offshore wind farm export cables," *Int. J. Electr. Power Energy Syst.*, vol. 113, no. May, pp. 982–990, 2019, doi: 10.1016/j.ijepes.2019.06.026.
- [153] M. U. T. Rentschler, F. Adam, and P. Chainho, "Design optimization of dynamic inter-array cable systems for floating offshore wind turbines," *Renew. Sustain. Energy Rev.*, vol. 111, no. April, pp. 622–635, 2019, doi: 10.1016/j.rser.2019.05.024.
- [154] T. Bantras, V. Cuk, J. F. G. Cobben, and W. L. Kling, "Estimation and classification of power losses DUE to reduced Power Quality," *IEEE Power Energy Soc. Gen. Meet.*, no. 1, pp. 1–6, 2012, doi: 10.1109/PESGM.2012.6344944.
- [155] D. Al kez et al., "A critical evaluation of grid stability and codes, energy storage and smart loads in power systems with wind generation," *Energy*, vol. 205, p. 117671, 2020, doi: 10.1016/j.energy.2020.117671.
- [156] O. Benzohra, S. S. Echcharqaouy, F. Fraija, and D. Saifaoui, "Integrating wind energy into the power grid: Impact and solutions," *Mater. Today Proc.*, vol. 30, no. xxxx, pp. 987–992, 2020, doi: 10.1016/j.matpr.2020.04.363.
- [157] O. P. Mahela, B. Khan, H. H. Alhelou, and S. Tanwar, "Assessment of power quality in the utility grid integrated with wind energy generation," *IET Power Electron.*, vol. 13, no. 13, pp. 2917–2925, 2020, doi: 10.1049/ietpel.2019.1351.
- [158] U. Datta, J. Shi, and A. Kalam, "Primary frequency control of a microgrid

with integrated dynamic sectional droop and fuzzy based pitch angle control," *Int. J. Electr. Power Energy Syst.*, vol. 111, no. April, pp. 248–259, 2019, doi: 10.1016/j.ijepes.2019.04.001.

- [159] X. Wang, Y. Wang, and Y. Liu, "Dynamic load frequency control for highpenetration wind power considering wind turbine fatigue load," *Int. J. Electr. Power Energy Syst.*, vol. 117, no. October 2019, p. 105696, 2020, doi: 10.1016/j.ijepes.2019.105696.
- [160] E. Sáiz-Marín, E. Lobato, and I. Egido, "New challenges to wind energy voltage control. Survey of recent practice and literature review," *IET Renew. Power Gener.*, vol. 12, no. 3, pp. 267–278, 2018, doi: 10.1049/ietrpg.2017.0065.
- [161] S. Ge, K. Yu, X. Chen, Y. Liao, X. Huang, and J. Zhao, "Research on power loss reduction method based on continuous regulating features of energyintensive industrial loads," 2016 IEEE Int. Conf. Power Syst. Technol. POWERCON 2016, pp. 1–5, 2016, doi: 10.1109/POWERCON.2016.7753931.
- [162] B. Sridhar and A. Kumar, "Loss Reduction in Distribution System with DSTATCOM and Wind Energy Considering Uncertainty," 2019 IEEE 1st Int. Conf. Energy, Syst. Inf. Process. ICESIP 2019, pp. 1–6, 2019, doi: 10.1109/ICESIP46348.2019.8938222.
- [163] Y. Yoo, J. H. Lee, G. Jang, M. Yoon, and S. Jung, "Study on a modified reactive power allocation strategy of WF management system considering electrical loss reduction and flexibility in practical operation," *IET Renew. Power Gener.*, vol. 13, no. 5, pp. 684–689, 2019, doi: 10.1049/ietrpg.2018.5074.
- [164] J. Liang, K. Zhang, A. Al-Durra, S. M. Muyeen, and D. Zhou, "A state-ofthe-art review on wind power converter fault diagnosis," *Energy Reports*, vol. 8, pp. 5341–5369, 2022, doi: 10.1016/j.egyr.2022.03.178.
- [165] E. B. Mora, J. Spelling, and A. H. Van Der Weijde, "How does risk aversion

shape overplanting in the design of offshore wind farms?," *J. Phys. Conf. Ser.*, vol. 1356, no. 1, 2019, doi: 10.1088/1742-6596/1356/1/012026.

- [166] Z. Zhang, D. Mei, H. Jiang, G. Liu, H. He, and Y. Chen, "Mode for reducing wind curtailment based on battery transportation," *J. Mod. Power Syst. Clean Energy*, vol. 6, no. 6, pp. 1158–1171, 2018, doi: 10.1007/s40565-018-0421-5.
- [167] A. Soroudi, A. Rabiee, and A. Keane, "Distribution networks' energy losses versus hosting capacity of wind power in the presence of demand flexibility," *Renew. Energy*, vol. 102, pp. 316–325, 2017, doi: 10.1016/j.renene.2016.10.051.
- [168] J. Zhang, A. Florita, B. M. Hodge, and J. Freedman, "Ramp forecasting performance from improved short-term wind power forecasting," *Proc. ASME Des. Eng. Tech. Conf.*, vol. 2A, no. May, 2014, doi: 10.1115/DETC2014-34775.
- [169] J. Kiviluoma *et al.*, "Variability in large-scale wind power generation," *Wind Energy*, vol. 19, no. 9, pp. 1649–1665, Sep. 2016, doi: 10.1002/we.1942.
- [170] Y. P. Wimalaratna, H. N. Afrouzi, K. Mehranzamir, M. B. M. Siddique, S. C. Liew, and J. Ahmed, "Analysing wind power penetration in hybrid energy systems based on techno-economic assessments," *Sustain. Energy Technol. Assessments*, vol. 53, no. PB, p. 102538, 2022, doi: 10.1016/j.seta.2022.102538.
- [171] L. Al-Ghussain, A. D. Ahmad, A. M. Abubaker, K. Hovi, M. A. Hassan, and A. Annuk, "Techno-economic feasibility of hybrid PV/wind/battery/thermal storage trigeneration system: Toward 100% energy independency and green hydrogen production," *Energy Reports*, vol. 9, pp. 752–772, 2023, doi: 10.1016/j.egyr.2022.12.034.
- [172] T. Kealy, "The need for energy storage on renewable energy generator outputs to lessen the Geeth effect, i.e. short-term variations mainly associated with wind turbine active power output," *Energy Reports*, vol. 9,

pp. 1018–1028, 2023, doi: 10.1016/j.egyr.2022.12.040.

- [173] Y. Yang, D. Zhu, X. Zou, Y. Chi, and Y. Kang, "Power Compensation Control for DFIG-Based Wind Turbines to Enhance Synchronization Stability During Severe Grid Faults," *IEEE Trans. Power Electron.*, vol. 37, no. 9, pp. 10139–10143, 2022, doi: 10.1109/TPEL.2022.3168883.
- [174] S. D. Ahmed and F. S. M. Al-ismail, "Grid Integration Challenges of Wind Energy: A Review," vol. 8, no. type 1, pp. 10857–10878, 2020.
- [175] B. Basu, A. Staino, and M. Basu, "Role of flexible alternating current transmission systems devices in mitigating grid fault-induced vibration of wind turbines," *Wind Energy*, vol. 17, no. 7, pp. 1017–1033, Jul. 2014, doi: 10.1002/we.1616.
- [176] X. Zhang, Z. Zhu, Y. Fu, and L. Li, "Optimized virtual inertia of wind turbine for rotor angle stability in interconnected power systems," *Electr. Power Syst. Res.*, vol. 180, no. May 2019, p. 106157, 2020, doi: 10.1016/j.epsr.2019.106157.
- [177] A. T. Tharakan and B. K. Panigrahi, "A dynamic programming based energy management algorithm for loss reduction in wind farm systems with storage," *IEEE Int. Conf. Power Electron. Drives Energy Syst. PEDES* 2016, vol. 2016-Janua, pp. 1–6, 2017, doi: 10.1109/PEDES.2016.7914319.
- [178] A. O. Almeida, M. A. Tomim, P. M. Almeida, and P. G. Barbosa, "A control strategy for an offshore wind farm with the generating units connected in series with a VSC-HVDC transmission link," *Electr. Power Syst. Res.*, vol. 180, no. May 2019, p. 106121, 2020, doi: 10.1016/j.epsr.2019.106121.
- [179] N. Wang, J. Li, W. Hu, B. Zhang, Q. Huang, and Z. Chen, "Optimal reactive power dispatch of a full-scale converter based wind farm considering loss minimization," *Renew. Energy*, vol. 139, pp. 292–301, 2019, doi: 10.1016/j.renene.2019.02.037.
- [180] B. Gustavsen and O. Mo, "Variable transmission voltage for loss minimization in long offshore wind farm AC export cables," *IEEE Trans.*
Power Deliv., vol. 32, no. 3, pp. 1422–1431, 2017, doi: 10.1109/TPWRD.2016.2581879.

- [181] M. Cullinane, F. Judge, M. O'Shea, K. Thandayutham, and J. Murphy, "Subsea superconductors: The future of offshore renewable energy transmission?," *Renew. Sustain. Energy Rev.*, vol. 156, no. December 2021, p. 111943, 2022, doi: 10.1016/j.rser.2021.111943.
- [182] Q. Jiang, B. Li, and T. Liu, "Tech-Economic Assessment of Power Transmission Options for Large-Scale Offshore Wind Farms in China," *Processes*, vol. 10, no. 5, 2022, doi: 10.3390/pr10050979.
- [183] S. Makhloufi, S. D. Koussa, and G. G. Pillai, "Cuckoo search algorithm for integration wind power generation to meet load demand growth," *Conf. Proc. - 2017 17th IEEE Int. Conf. Environ. Electr. Eng. 2017 1st IEEE Ind. Commer. Power Syst. Eur. EEEIC / I CPS Eur. 2017*, pp. 1–6, 2017, doi: 10.1109/EEEIC.2017.7977396.
- [184] B. Sultana, M. W. Mustafa, U. Sultana, and A. R. Bhatti, "Review on reliability improvement and power loss reduction in distribution system via network reconfiguration," *Renew. Sustain. Energy Rev.*, vol. 66, pp. 297– 310, 2016, doi: 10.1016/j.rser.2016.08.011.
- [185] W. M. Da Rosa, P. Rossoni, J. C. Teixeira, and E. A. Belati, "Insertion of wind generators in electrical power systems aimed at active losses reduction using sensitivity analysis," *Int. J. Electr. Power Energy Syst.*, vol. 80, pp. 306–311, 2016, doi: 10.1016/j.ijepes.2016.02.002.
- [186] A. Azizi, "Evaluation Improvement of Production Productivity Performance using Statistical Process Control, Overall Equipment Efficiency, and Autonomous Maintenance," *Procedia Manuf.*, vol. 2, no. February, pp. 186–190, 2015, doi: 10.1016/j.promfg.2015.07.032.
- [187] M. Caterino *et al.*, "Simulation techniques for production lines performance control," *Procedia Manuf.*, vol. 42, no. 2019, pp. 91–96, 2020, doi: 10.1016/j.promfg.2020.02.027.

- [188] C. Andersson and M. Bellgran, "On the complexity of using performance measures: Enhancing sustained production improvement capability by combining OEE and productivity," *J. Manuf. Syst.*, vol. 35, pp. 144–154, 2015, doi: 10.1016/j.jmsy.2014.12.003.
- [189] Z. Heng, L. Aiping, X. Liyun, and G. Moroni, "Automatic estimate of OEE considering uncertainty," *Procedia CIRP*, vol. 81, pp. 630–635, 2019, doi: 10.1016/j.procir.2019.03.167.
- [190] M. A. Lutz, P. Görg, S. Faulstich, R. Cernusko, and S. Pfaffel, "Monetarybased availability: A novel approach to assess the performance of wind turbines," *Wind Energy*, vol. 23, no. 1, pp. 77–89, 2020, doi: 10.1002/we.2411.
- [191] J. Olauson, P. Edström, and J. Rydén, *Wind turbine performance decline in Sweden*, vol. 20, no. 12. 2017.
- [192] C. Dao, B. Kazemtabrizi, and C. Crabtree, "Wind turbine reliability data review and impacts on levelised cost of energy," *Wind Energy*, vol. 22, no. 12, pp. 1848–1871, 2019, doi: 10.1002/we.2404.
- [193] N. Sarma, P. M. Tuohy, O. Özgönenel, and S. Djurović, "Early life failure modes and downtime analysis of onshore type-III wind turbines in Turkey," *Electr. Power Syst. Res.*, vol. 216, no. December 2022, 2023, doi: 10.1016/j.epsr.2022.108956.
- [194] Y. Kikuchi and T. Ishihara, "Availability and Icoe analysis considering failure rate and downtime for onshore wind turbines in Japan," *Energies*, vol. 14, no. 12, 2021, doi: 10.3390/en14123528.
- [195] R. Bi, K. Qian, C. Zhou, D. M. Hepburn, and J. Rong, "A survey of failures in wind turbine generator systems with focus on a wind farm in China," *Int. J. Smart Grid Clean Energy*, pp. 366–373, 2014, doi: 10.12720/sgce.3.4.366-373.
- [196] R. R. Hill, J. a Stinebaugh, D. Briand, A. S. Benjamin, and J. Linsday, "Wind Turbine Reliability: A Database and Analysis Approach," Sand2008-0983,

no. February, p. 60, 2008.

- [197] F. Carlsson, E. Eriksson, and M. Dahlberg, "Damage preventing measures for wind turbines," no. August, 2010.
- [198] R. Kumar, T. Stallard, and P. K. Stansby, "Large-scale offshore wind energy installation in northwest India: Assessment of wind resource using Weather Research and Forecasting and levelized cost of energy," *Wind Energy*, vol. 24, no. 2, pp. 174–192, 2021, doi: 10.1002/we.2566.
- [199] R. Ebenhoch, D. Matha, S. Marathe, P. C. Muñoz, and C. Molins, *Comparative levelized cost of energy analysis*, vol. 80. Elsevier B.V., 2015.
- [200] A. Kolios, J. Walgern, S. Koukoura, R. Pandit, and J. Chiachio-Ruano, "openO&M: Robust O&M open access tool for improving operation and maintenance of offshore wind turbines," *Proc. 29th Eur. Saf. Reliab. Conf. ESREL 2019*, pp. 629–635, 2020, doi: 10.3850/978-981-11-2724-3-1134cd.
- [201] M. Asgarpour and R. van de Pieterman, "O&M Cost Reduction of Offshore Wind Farms - A Novel Case Study," no. ECN-E-14-028, p. 43, 2014, [Online]. Available: https://www.ecn.nl/publications/PdfFetch.aspx?nr=ECN-E--14-028.
- [202] E. Lotovskyi, A. P. Teixeira, and C. Guedes Soares, "Availability Analysis of an Offshore Wind Turbine Subjected to Age-Based Preventive Maintenance by Petri Nets," *J. Mar. Sci. Eng.*, vol. 10, p. 1000, 2022, doi: 10.3390/jmse10071000.
- [203] M. R. Soltani, A. H. Birjandi, and M. Seddighi Moorani, "Effect of surface contamination on the performance of a section of a wind turbine blade," *Sci. Iran.*, vol. 18, no. 3 B, pp. 349–357, 2011, doi: 10.1016/j.scient.2011.05.024.
- [204] Reuters, "Optimising Annual Energy Production with apt Handling of Yaw Misalignment," 2013. https://www.reutersevents.com/renewables/windenergy-update/optimising-annual-energy-production-apt-handling-yaw-

misalignment (accessed Jan. 27, 2023).

- [205] D. Song *et al.*, "Coordinated optimization on energy capture and torque fluctuation of wind turbines via variable weight NMPC with fuzzy regulator," *Appl. Energy*, vol. 312, no. December 2021, p. 118821, 2022, doi: 10.1016/j.apenergy.2022.118821.
- [206] H. H. Başaran and İ. Tarhan, "Investigation of offshore wind characteristics for the northwest of Türkiye region by using multi-criteria decision-making method (MOORA)," *Results Eng.*, vol. 16, no. November, p. 100757, 2022, doi: 10.1016/j.rineng.2022.100757.
- [207] E. G. Sakka, D. V. Bilionis, D. Vamvatsikos, and C. J. Gantes, "Onshore wind farm siting prioritization based on investment profitability for Greece," *Renew. Energy*, vol. 146, pp. 2827–2839, 2020, doi: 10.1016/j.renene.2019.08.020.
- [208] C. Moné, M. Hand, M. Bolinger, J. Rand, D. Heimiller, and J. Ho, "2015 Cost of Wind Energy Review," no. May, 2017, doi: 10.2172/1366436.
- [209] M. Baroni, "The Integration of Non-dispatchable Renewables," in *The Palgrave Handbook of International Energy Economics*, M. Hafner and G. Luciani, Eds. Cham: Springer International Publishing, 2022, pp. 269–299.
- [210] I. O. Ozioko, N. S. Ugwuanyi, A. O. Ekwue, and C. I. Odeh, "Wind energy penetration impact on active power flow in developing grids," *Sci. African*, vol. 18, p. e01422, 2022, doi: 10.1016/j.sciaf.2022.e01422.
- [211] J. K. Kaldellis, G. T. Tzanes, C. Papapostolou, K. Kavadias, and D. Zafirakis, "Analyzing the Limitations of Vast Wind Energy Contribution in Remote Island Networks of the Aegean Sea Archipelagos," *Energy Procedia*, vol. 142, pp. 787–792, 2017, doi: 10.1016/j.egypro.2017.12.127.
- [212] B. Kroposki, "Integrating high levels of variable renewable energy into electric power systems," *J. Mod. Power Syst. Clean Energy*, vol. 5, no. 6, pp. 831–837, 2017, doi: 10.1007/s40565-017-0339-3.

- [213] D. A. Katsaprakakis and D. G. Christakis, "2.09 Wind Parks Design, Including Representative Case Studies," in *Comprehensive Renewable Energy (Second Edition)*, Second Edi., T. M. Letcher, Ed. Oxford: Elsevier, 2022, pp. 226–278.
- [214] DM ENERGY, "Wind Turbine Grid Connection and Interaction," 2001. [Online]. Available: https://ec.europa.eu/energy/sites/ener/files/documents/2001_fp5_brochur e_energy_env.pdf.
- [215] C. Sourkounis and P. Tourou, "Grid Code Requirements for Wind Power Integration in Europe," *Conf. Pap. Energy*, vol. 2013, pp. 1–9, 2013, doi: 10.1155/2013/437674.
- [216] M. Q. DUONG, T. V. DINH, N. T. N. TRAN, G. SAVA, and A. KIES, "A COMPARATIVE STUDY OF WIND TURBINE GENERATORS OPERATING PERFORMANCE A CASE STUDY FOR THE VIETNAMESE NINH THUAN-GRID," in Bulletin of the Polytechnic Institute of Jassy: Electrical Engineering, Power Engineering, Electronics, 2017, vol. 63, no. 67, 17–32, [Online]. Available: http://www.bulipipp. eee.tuiasi.ro/archive/2017/fasc.3/p2_f3_2017.pdf.
- [217] D. S. Bankar, S. G. Desai, and V. V. Mehtre, "Performance comparison of SCIG and DFIG based wind farm in ETAP," Int. Conf. Autom. Control Dyn. Optim. Tech. ICACDOT 2016, pp. 403–408, 2017, doi: 10.1109/ICACDOT.2016.7877617.
- [218] IEA, "Renewables 2020 Analysis and forecast to 2025," 2020. [Online].
 Available: https://iea.blob.core.windows.net/assets/1a24f1fe-c971-4c25-964a-57d0f31eb97b/Renewables_2020-PDF.pdf.
- [219] N. B. Negra, J. Todorovic, and T. Ackermann, "Loss evaluation of HVAC and HVDC transmission solutions for large offshore wind farms," *Electr. Power Syst. Res.*, vol. 76, no. 11, pp. 916–927, 2006, doi: 10.1016/j.epsr.2005.11.004.

- [220] J. A. Carta, P. Cabrera, and J. González, "2.20 Wind Power Integration," in *Comprehensive Renewable Energy (Second Edition)*, Second Edi., T. M. Letcher, Ed. Oxford: Elsevier, 2022, pp. 644–720.
- [221] E. Muljadi, C. P. Butterfield, J. Chacon, and H. Romanowitz, "Power quality aspects in a wind power plant," 2006 IEEE Power Eng. Soc. Gen. Meet. PES, 2006, doi: 10.1109/pes.2006.1709244.
- [222] W. Yang and S. W. Tian, "Research on a power quality monitoring technique for individual wind turbines," *Renew. Energy*, vol. 75, pp. 187– 198, 2015, doi: 10.1016/j.renene.2014.09.037.
- [223] S. Heier, Grid integration of wind energy: onshore and offshore conversion systems. John Wiley & Sons, 2014.
- [224] A. Kusiak, H. Zheng, and Z. Song, "On-line monitoring of power curves," *Renew. Energy*, vol. 34, no. 6, pp. 1487–1493, 2009, doi: 10.1016/j.renene.2008.10.022.
- [225] P. P. Reboucas Filho, N. De Medeiros Mendonca E Nascimento, S. S. Araujo Alves, S. Luz Gomes, and C. Marques De Sa Medeiros, "Estimation of the energy production in a wind farm using regression methods and wind speed forecast," *Proc. - 2018 Brazilian Conf. Intell. Syst. BRACIS 2018*, pp. 79–84, 2018, doi: 10.1109/BRACIS.2018.00022.
- [226] M. Batta, "Machine Learning Algorithms A Review," Int. J. Sci. Res., vol. 18, no. 8, pp. 381–386, 2018, doi: 10.21275/ART20203995.
- [227] J. R. Quinlan, "Simplifying Decision Trees," *Int. J. Hum. Comput. Stud.*, vol. 27, pp. 221–234, 1987, doi: 10.1006/ijhc.1987.0321.
- [228] H. Drucker, C. J. C. Surges, L. Kaufman, A. Smola, and V. Vapnik, "Support Vector Regression Machines," *Adv. Neural Inf. Process. Syst.*, vol. 1, pp. 155–161, 1997.
- [229] L. Breiman, "Random forests," Mach. Learn., vol. 45, pp. 5–32, 2001, doi: 10.1201/9780429469275-8.

- [230] R. K. Pandit and D. Infield, "Comparative analysis of Gaussian Process power curve models based on different stationary covariance functions for the purpose of improving model accuracy," *Renew. Energy*, vol. 140, pp. 190–202, 2019, doi: 10.1016/j.renene.2019.03.047.
- [231] S. Saxena, "A Beginner's Guide to Random Forest Hyperparameter Tuning," 2020. https://www.analyticsvidhya.com/blog/2020/03/beginnersguide-random-forest-hyperparameter-tuning/ (accessed May 09, 2021).
- [232] S. Rose and J. Apt, "The Cost of Curtailing Wind Turbines for Frequency Regulation and Ramp-Rate Limitation," *Proc. 29th USAEE/IAEE North Am. Conf.* ..., pp. 1–18, 2010, [Online]. Available: http://www.usaee.org/usaee2010/submissions/OnlineProceedings/Rose and Apt - The Cost of Curtailing Wind Turbines.pdf.
- [233] G. D'Amico, F. Petroni, and S. Vergine, "Ramp Rate Limitation of Wind Power: An Overview," *Energies*, vol. 15, no. 16, pp. 1–15, 2022, doi: 10.3390/en15165850.
- [234] SHAP, "Welcome to the SHAP documentation," 2018. https://shap.readthedocs.io/en/latest/index.html.
- [235] J. Ribrant and L. Bertling, "Survey of failures in wind power systems with focus on Swedish wind power plants during 1997-2005," 2007 IEEE Power Eng. Soc. Gen. Meet. PES, vol. 22, no. 1, pp. 167–173, 2007, doi: 10.1109/PES.2007.386112.
- [236] C. Kaidis, B. Uzunoglu, and F. Amoiralis, "Wind turbine reliability estimation for different assemblies and failure severity categories," *IET Renew. Power Gener.*, vol. 9, no. 8, pp. 892–899, 2015, doi: 10.1049/iet-rpg.2015.0020.
- [237] C. Zhu and Y. Li, "Reliability Analysis of Wind Turbines," in Stability Control and Reliable Performance of Wind Turbines, K. E. Okedu, Ed. Rijeka: IntechOpen, 2018.
- [238] E. Vesely, "Consider Automated Machine Learning for Wind Turbine Asset Maintenance," 2017. https://www.powermag.com/consider-automated-

machine-learning-for-wind-turbine-asset-maintenance/ (accessed Jan. 30, 2023).

- [239] G. Hughes, "Wind Power Costs in the United Kingdom Volume 1," vol. I, 2020.
- [240] S. Pfaffel, S. Faulstich, and K. Rohrig, "Performance and reliability of wind turbines: A review," *Energies*, vol. 10, no. 11, 2017, doi: 10.3390/en10111904.
- [241] L. V. Villamor, V. Avagyan, and H. Chalmers, "Opportunities for reducing curtailment of wind energy in the future electricity systems: Insights from modelling analysis of Great Britain," *Energy*, vol. 195, 2020, doi: 10.1016/j.energy.2019.116777.
- [242] S. EVANS, "Analysis: Record-low price for UK offshore wind cheaper than existing gas plants by 2023," *Carbon Brief*, 2019. https://www.carbonbrief.org/analysis-record-low-uk-offshore-windcheaper-than-existing-gas-plants-by-2023.
- [243] L. Ramirez, D. Fraile, and G. Brindley, "Offshore Wind in Europe Key trends and statistics 2019," 2020. [Online]. Available: https://windeurope.org/wp-content/uploads/files/aboutwind/statistics/WindEurope-Annual-Offshore-Statistics-2019.pdf.
- [244] S. O. Effiom, B. N. Nwankwojike, and F. I. Abam, "Economic cost evaluation on the viability of offshore wind turbine farms in Nigeria," *Energy Reports*, vol. 2, pp. 48–53, 2016, doi: 10.1016/j.egyr.2016.03.001.
- [245] R. Wiser and M. Bolinger, "2010 Wind technologies market report," 2011.[Online]. Available: http://www.osti.gov/bridge.
- [246] IRENA, Renewable Power Generation Costs in 2020. 2020.
- [247] BEIS, "Electricity Generation Cost Report 2020," 2020. [Online]. Available: https://assets.publishing.service.gov.uk/government/uploads/system/uplo ads/attachment_data/file/911817/electricity-generation-cost-report-

2020.pdf.

- [248] Z. Ren, A. S. Verma, Y. Li, J. J. E. Teuwen, and Z. Jiang, "Offshore wind turbine operations and maintenance: A state-of-the-art review," *Renew. Sustain. Energy Rev.*, vol. 144, no. August 2020, 2021, doi: 10.1016/j.rser.2021.110886.
- [249] C. Milne, S. Jalili, and A. Maheri, "Decommissioning cost modelling for offshore wind farms: A bottom-up approach," *Sustain. Energy Technol. Assessments*, vol. 48, no. August, p. 101628, 2021, doi: 10.1016/j.seta.2021.101628.
- [250] Low Carbon Contracts Company Ltd, "CfD Register," 2021. https://www.lowcarboncontracts.uk/cfd-register/.
- [251] N. R. E. L. (NREL) Walter Musial *et al.*, "Offshore Wind Market Report: 2021 Edition," *Dep. Energy*, pp. 0–1, 2020, [Online]. Available: https://www.energy.gov/eere/wind/articles/offshore-wind-market-report-2021-edition-released.
- [252] J. Bosch, I. Staffell, and A. D. Hawkes, "Global levelised cost of electricity from offshore wind," *Energy*, vol. 189, p. 116357, 2019, doi: 10.1016/j.energy.2019.116357.
- [253] Y. Liang, Y. Ma, H. Wang, A. Mesbahi, B. Jeong, and P. Zhou, "Levelised cost of energy analysis for offshore wind farms – A case study of the New York State development," *Ocean Eng.*, vol. 239, no. August, 2021, doi: 10.1016/j.oceaneng.2021.109923.
- [254] T. Wizelius, "2.13 Design and Implementation of a Wind Power Project," in Comprehensive Renewable Energy (Second Edition), Second Edi., T. M. Letcher, Ed. Oxford: Elsevier, 2022, pp. 390–429.
- [255] P. Schwabe, S. Lensink, and M. Hand, "IEA Wind Task 26," Wind Energy, pp. 1–122, 2011, [Online]. Available: http://www.nrel.gov/docs/fy11osti/48155.pdf.

- [256] M. J. Kaiser and B. F. Snyder, Offshore Wind Energy Cost Modelling: Installation and Decommissioning. Springer, 2012.
- [257] D. Cevasco, M. Collu, and Z. Lin, "O&M Cost-Based FMECA: Identification and Ranking of the Most Critical Components for 2-4 MW Geared Offshore Wind Turbines," *J. Phys. Conf. Ser.*, vol. 1102, no. 1, 2018, doi: 10.1088/1742-6596/1102/1/012039.
- [258] T. Poulsen, C. B. Hasager, and C. M. Jensen, "The role of logistics in practical levelized cost of energy reduction implementation and government sponsored cost reduction studies: Day and night in offshorewind operations and maintenance logistics," *Energies*, vol. 10, no. 4, 2017, doi: 10.3390/en10040464.
- [259] S. Alsubal, W. S. Alaloul, M. A. Musarat, E. L. Shawn, M. S. Liew, and P. Palaniappan, "Life cycle cost assessment of offshore wind farm: Kudat malaysia case," *Sustain.*, vol. 13, no. 14, 2021, doi: 10.3390/su13147943.
- [260] R. F. Pereira, R. A. Guedes, and C. S. Santos, "Comparing WAsP and MeteodynWT estimates for the 'regular' user," *Eur. Wind Energy Conf. Exhib. 2010, EWEC 2010*, vol. 5, no. January, p. 3385, 2010.
- [261] Meteodyn Universe Microscale (WT), "User Manual," 2022.
- [262] K. Dykes, S. A. Ning, G. Scott, P. Graf, and R. P. O. W. E. T. O. USDOE Office of Energy Efficiency and Renewable Energy (EERE), "WISDEM® (Wind-Plant Integrated System Design and Engineering Model)." United States, 2021, doi: 10.11578/dc.20211208.2.
- [263] L. Fingersh, M. Hand, A. Laxson, L. Fingersh, M. Hand, and A. Laxson, "Wind Turbine Design Cost and Scaling Model Wind Turbine Design Cost and Scaling Model," no. December, 2006.
- [264] IEA, Projected Costs of Generating Electricity. 2010.
- [265] C. Schwarz and E. Badia, "Decommissioning of Wind Farms: Costs and Opportunities," vol. 49, no. 0, pp. 8–11, 2009, [Online]. Available:

http://www.pre-sustainability.com/manuals.

- [266] D. Middleton, "Cost Comparison: Decommissioning a Wind Turbine vs. Plugging & Abandoning an Oil Well," 2021. https://wattsupwiththat.com/2021/08/29/cost-comparisondecommissioning-a-wind-turbine-vs-plugging-abandoning-an-oil-well/ (accessed Feb. 18, 2023).
- [267] K. Clark, "Improving maintenance by adopting a P-F curve methodology,"
 2019. https://www.isa.org/intech-home/2019/march-april/features/improving-maintenance-by-adopting-a-p-f-curve-meth (accessed Feb. 07, 2023).
- [268] Hanara-Software, "Predict and Prevent: Improving Your Maintenance Strategy." https://www.hanarasoft.com/preventive-maintenance-vspredictive-maintenance/ (accessed Sep. 22, 2022).
- [269] D. Mcmillan and G. Ault, "Condition monitoring benefit for onshore wind turbines: Sensitivity to operational parameters Condition Monitoring Benefit for Onshore Wind Turbines : Sensitivity to Operational Parameters," no. April, 2008, doi: 10.1049/iet-rpg.
- [270] F. Besnard, J. Nilsson, and L. Bertling, "On the economic benefits of using Condition Monitoring Systems for maintenance management of wind power systems," in 2010 IEEE 11th International Conference on Probabilistic Methods Applied to Power Systems, Jun. 2010, pp. 160–165, doi: 10.1109/PMAPS.2010.5528992.
- [271] P. Zhou and P. T. Yin, "An opportunistic condition-based maintenance strategy for o ff shore wind farm based on predictive analytics," *Renew. Sustain. Energy Rev.*, vol. 109, no. March, pp. 1–9, 2019, doi: 10.1016/j.rser.2019.03.049.
- [272] J. M. Pinar Pérez, E. Segura Asensio, and F. P. García Márquez, "Economic Viability Analytics for Wind Energy Maintenance Management," in Advanced Business Analytics, F. P. García Márquez and B. Lev, Eds.

Cham: Springer International Publishing, 2015, pp. 39–54.

- [273] J. Nilsson, S. Member, L. Bertling, and A. Availability, "Maintenance Management of Wind Power Systems Using Condition Monitoring Systems — Life Cycle Cost Analysis for Two Case Studies," vol. 22, no. 1, pp. 223– 229, 2007.
- [274] K. Fischer and D. Coronado, "Condition monitoring of wind turbines: state of the art, user experience and recommendations," *Fraunhofer-IWES, Bremerhaven*, pp. 1–89, 2015, [Online]. Available: https://www.researchgate.net/profile/Katharina-Fischer-8/publication/299560600_Condition_monitoring_of_wind_turbines_State_ of_the_art_user_experience_and_recommendations/links/56fef12c08aee 995dde73bad/Condition-monitoring-of-wind-turbines-State-of-the-art-userexperience-and-recommendations.pdf.
- [275] W. Yang, P. J. Tavner, C. J. Crabtree, Y. Feng, and Y. Qiu, "Wind turbine condition monitoring: technical and commercial challenges," *Wind Energy*, vol. 17, no. 5, pp. 673–693, May 2014, doi: 10.1002/we.1508.
- [276] NewEnergyUpdate, "Wind O & M Data Pack: Turbine Failure Rates & Costs," 2017. https://docs.wind-watch.org/wind-maintenance-2017.pdf (accessed Feb. 07, 2023).
- [277] X. Zhang and M. U. Zubair, "Extending the useful life of elevators through appropriate maintenance strategies," *J. Build. Eng.*, vol. 51, p. 104347, 2022, doi: https://doi.org/10.1016/j.jobe.2022.104347.
- [278] N. J. Prescott, "Equipment life: can we afford to extend it?," in Annual Reliability and Maintainability Symposium 1995 Proceedings, 1995, pp. 529–535, doi: 10.1109/RAMS.1995.513294.
- [279] WindExchange, "No TitleProduction Tax Credit and Investment Tax Credit for Wind Energy." https://windexchange.energy.gov/projects/taxcredits#:~:text=Renewable Energy Production Tax Credit (PTC)&text=Wind energy projects placed into,10 years of electricity

generation. (accessed Feb. 07, 2023).

- [280] US Department of Energy, "Advancing the Growth of the U.S. Wind Industry: Federal Incentives, Funding, and Partnership Opportunities," 2021, [Online]. Available: www.energy.gov/lpo/services/solicitations/.
- [281] Ofgem, "Feed-in Tariff (FIT): Tariff Table 1 April 2022." https://www.ofgem.gov.uk/publications/feed-tariff-fit-tariff-table-1-april-2022 (accessed Feb. 07, 2023).
- [282] W. Wu and B. Lin, "Application value of energy storage in power grid: A special case of China electricity market," *Energy*, vol. 165, pp. 1191–1199, 2018, doi: https://doi.org/10.1016/j.energy.2018.09.202.
- [283] D. Milborrow, "2.15 Wind Energy Economics," in Comprehensive Renewable Energy (Second Edition), Second Edi., T. M. Letcher, Ed. Oxford: Elsevier, 2022, pp. 463–496.
- [284] The Scottish Government, "Scottish Offshore Wind to Green Hydrogen Opportunity Assessment," 2020. [Online]. Available: https://www.gov.scot/binaries/content/documents/govscot/publications/res earch-and-analysis/2020/12/scottish-offshore-wind-green-hydrogenopportunity-assessment2/documents/scottish-offshore-wind-greenhydrogen-opportunity-assessment/scottish-offshore-wind-greenhydrogen-opportunity-assessment/govscot%3Adocument/scottishoffshore-wind-green-hydrogen-opportunity-assessment.pdf.
- [285] C. Wolter, H. Klinge Jacobsen, L. Zeni, G. Rogdakis, and N. A. Cutululis, "Overplanting in offshore wind power plants in different regulatory regimes," *Wiley Interdiscip. Rev. Energy Environ.*, vol. 9, no. 3, pp. 1–16, 2020, doi: 10.1002/wene.371.
- [286] D. P. Zafirakis, "2.06 Energy Yield of Contemporary Wind Turbines," in Comprehensive Renewable Energy (Second Edition), Second Edi., T. M. Letcher, Ed. Oxford: Elsevier, 2022, pp. 124–171.
- [287] B. Johnston, A. Foley, J. Doran, and T. Littler, "Levelised cost of energy, A

challenge for offshore wind," *Renew. Energy*, vol. 160, pp. 876–885, 2020, doi: 10.1016/j.renene.2020.06.030.

- [288] K. P. B. Sathler, K. Salonitis, and A. Kolios, "Overall Equipment Effectiveness as a Metric for Assessing Operational Losses in Wind Farms: A Critical Review of Literature," *Int. J. Sustain. Energy*, 2023, doi: 10.1080/14786451.2023.2189490.
- [289] K. P. B. Sathler, B. Yeter, and A. Kolios, "Impact of operational losses on the levelized costs of energy and in the economic viability of offshore wind power projects [Manuscript submitted for publication]," 2023.
- [290] K. P. B. Sathler, A. Kolios, and B. Yeter, "Effect of Energy Losses on Onshore Wind Turbines Techno-economic Assessment [Manuscript submitted for publication]," 2023.

APPENDICES

Appendix A

	LCOE (£/MWh)	NPV (£/MW)	Critical Year	Investment Breakeven Year
Reference	£88.65	£1,084,481.25	-	13
Scenario C	£92.07	£887,195.71	-	13
Scenario B	£93.22	£823,880.50	-	13
Scenario B + C	£96.71	£641,849.25	20	14
Scenario A	£101.43	£459,592.91	15	14
Scenario A + C	£105.25	£273,207.45	13	15
Scenario A + B	£106.85	£198,992.16	12	16
Scenario A + B + C	£110.74	27,861.00	11	17

Table_Apx A-1 Results of Economic Model - return rate = 4%

Table_Apx A-2 Results of Economic Model - return rate = 8%

	LCOE (£/MWh)	NPV (£/MW)	Critical Year	Investment Breakeven Year
Reference	£110.16	£42,534.77	-	19
Scenario C	£113.69	-£76,267.28	4	20
Scenario B	£114.99	-£118,490.43	3	20
Scenario B + C	£118.55	-£228,721.84	2	20
Scenario A	£121.37	-£333,762.53	1	-
Scenario A + C	£125.15	-£446,541.55	-	-
Scenario A + B	£126.84	-£494,787.73	-	-
Scenario A + B + C	£130.65	-£598,996.11	-	-

Table_Apx A-3 Result MCS – return rate = 4%.

	LCOE (IC 95%)			
	Lower Limit	Average	Upper Limit	NPV > 0 (%)
Reference	£65.75	£88.81	£111.88	0.7771
Scenario C	£68.26	£92.22	£116.18	0.7503
Scenario B	£68.77	£93.27	£117.77	0.7301
Scenario B + C	£71.45	£96.84	£122.23	0.6835
Scenario A	£73.55	£101.51	£129.47	0.6236
Scenario A + C	£76.52	£105.48	£134.44	0.5776
Scenario A + B	£77.45	£106.93	£136.41	0.5534
Scenario A + B + C	£80.53	£111.18	£141.84	0.5082

Table_Apx A-4 Result MCS – return rate = 8%.

	LCOE (IC 95%)			
	Lower Limit	Average	Upper Limit	NPV > 0 (%)
Reference	£82.60	£110.31	£138.02	0.5054
Scenario C	£85.01	£113.81	£142.62	0.4787
Scenario B	£86.21	£115.39	£144.57	0.4536
Scenario B + C	£89.21	£119.01	£148.80	0.4060
Scenario A	£90.22	£121.68	£153.14	0.3719
Scenario A + C	£92.68	£125.35	£158.03	0.3247
Scenario A + B	£94.02	£127.20	£160.38	0.3110
Scenario A + B + C	£96.31	£130.92	£165.52	0.2628

Appendix B



Figure_Apx B-1 MMCS results for returning tax = 4%.



Figure_Apx B-2 MMCS results for returning tax = 8%.